# **Pilkington Library**

Author/Filing Title .	BEANMONT	••••
 Vol. No	Class Mark7	
Please note th	at fines are charg overdue items.	ged on ALL
fûr	din ik.	- MNULT





. ι

# RAPID SCREENING OF PROTEOLYTIC ENZYMES AND THEIR INHIBITORS USING FLUORESCENCE METHODS

By Marie Louise Beaumont

A Doctoral Thesis submitted in partial fulfilment of the requirements for the award of

# DOCTOR OF PHILOSOPHY

of Loughborough University

© by Marie L. Beaumont (2001)

Supervisor: Professor James N. Miller Department of Chemistry Loughborough Leicestershire

11 - 1 - 1	1ĩ	-	ì	-	٤		
------------	----	---	---	---	---	--	--

··· · , ·

> > いたい

- +

Date June 02 Class Acc. Oto258798

------

.

#### ACKNOWLEDGEMENTS

Thank you so very much to Professor Jim Miller for his guidance and support throughout this project. He has been a brilliant supervisor. Many thanks also to everyone at Kalibrant Ltd., especially Derek, Martin, and Adrian, for their continued help and advice, and to Kalibrant Ltd. for its sponsorship.

I would like to say a big thank you to my friends and colleagues Lorna, Louise, Jill, Sonia, Dongfang, Maggi, Rob, Dave, and Hussain. It has been such good fun working with you and I will miss you all.

Last, but definitely not least, I would like to say thanks to Mum, Dad, and Gerry. I love you. I have included two fluorescent penguins for you to find so that you get a little enjoyment out of this thesis!



#### ABSTRACT

New fluorescence methods for the determination of proteolytic enzymes and their inhibitors have been investigated. Fluorescence polarisation and energy transfer methods have both been examined and of these two approaches energy transfer methods have proven superior. Out of the energy transfer methods, the use of an over-labelled substrate, which is virtually non-fluorescent but becomes highly fluorescent upon proteolytic liberation of dye-labelled peptides, has been demonstrated to be most successful. A substrate utilising  $\alpha$ -casein labelled with a long-wavelength BODIPY® dye has been synthesised and shown to be a suitable intramolecularly guenched conjugate for the determination of proteolytic enzymes and their inhibitors in flow-based screening programmes. The assays are rapid, taking at most 2 minutes for the enzyme assays and 5 minutes for the inhibition assays. Aprotinin, a serine protease inhibitor also used as a drug in surgery to prevent blood loss, has been detected at concentrations as low as 0.161 nM, and limits of detection of 2.50  $\mu$ M, 0.14  $\mu$ M, and 7.20 nM were obtained for 3-nitrophenylboronic acid, leupeptin, and pepstatin A respectively. The adaptation of these assays to a multi-determination format has also been considered.

These fluorogenic assays combine the sensitivity of long-wavelength fluorescence with the ease of automation that can be applied to flow injection analysis and could be incorporated into a simple portable fluorescence instrument. They have many clinical applications, since proteases are involved in virtually all biological functions and dysfunctions and therefore are promising as drug targets. The flow-based procedure described here can be used in the high-throughput screening of candidate drug molecules.

II

#### **RESEARCH AIMS**

ŗ

The objective of this investigation was to develop new fluorescence methods for the determination of proteolytic enzymes and their inhibitors. Upon finding a sensitive method, this assay was to be transferred to a flow injection analysis system to create a procedure that would be suitable for the high throughput screening of drug candidate molecules. In order for the assay to be used in a high throughput screening function, the assay had the essential requirement of being rapid. With this in mind the progression of this work was to aim towards the adaptation of the already established assay to a set-up that would allow the multi-determination of proteolytic enzymes and their inhibitors.

# SCOPE OF WORK

The work presented in this thesis reports the investigations that were carried out to achieve the above aims. Chapter 1 provides an overview of the subject areas that are pertinent to the investigation, while chapter 2 describes the materials, instrumentation, and general procedures that have been adopted for use during the practical work.

Chapter 3 deals with the first investigation of a fluorescence method, fluorescence polarisation, for the determination of proteolytic enzymes. This chapter discusses the work carried out and the conclusions derived from that work.

Chapters 4 and 5 investigate the use of energy transfer fluorescence methods The use of fluorescence resonance energy transfer involving substrates that are labelled with two different fluorophores is explored as a fluorescence method in

iii

chapter 4. Chapter 5 describes the synthesis of over-labelled single fluorophore substrates resulting in energy transfer between neighbouring fluorophore molecules. The application of this substrate to a flow injection analysis system and subsequent proteolytic inhibitor determinations is demonstrated.

Chapter 6 discusses set-ups appropriate for multi-determination and investigates these set-ups for use in these types of assays.

Finally, chapter 7 provides a discussion on the practical work that has been carried out and the conclusions that can be derived. Ideas on further extensions of the work carried out and future directions are also discussed.

# LIST OF FIGURES

Figure 1.1	Jablonski diagram showing absorption and emission processes of fluorescence and phosphorescence 3
Figure 1.2	Structure of FITC, isomer I 7
Figure 1.3	Structure of tetramethylrhodamine 8
Figure 1.4	Structure of BODIPY® 630 / 650 –X, succinimidyl ester
Figure 1.5	Structure of Cy <sup>™</sup> 5 bisfunctional dye
Figure 1.6	Structure of Cy <sup>™</sup> 5 5 bisfunctional dye 11
Figure 1.7	The wave model of light (V = velocity, E = electric field, H = magnetic field, vectors)
Figure 1.8	Electric vectors of light randomly distributed (a) and parallel (b and c)
Figure 1.9	Schematic representation of spectral overlap required for FRET
Figure 1.10	Schematic diagram of a typical simple FIA system 23
Figure 1.1 <b>1</b>	Biochemical classes of drug targets of current therapies 27

Figure 1.12	Dependence of initial velocity on substrate concentration	
	for a single-substrate enzyme-catalysed reaction	32
Figure 1.13	Lineweaver-Burk plot	34
Figure 1.14	Classification of immobilisation methods	41
Figure 1.15	Proteolytic action of anglotensin converting enzyme (ACE)	45
Figure 1.16	Trend in number of published articles in the areas of proteases, protease inhibitors and high throughput	
	screening over the last decade	47
Figure 1.17a	Principle of the use of fluorescence polarisation for the	
	detection of a large lightly labelled fluorescent protein	50
Figure 1.17b	Principle of the use of fluorescence polarisation for the	
	detection of protease digested fluorescence peptides	51
Figure 1.18	Schematic representation of (a) a peptide sequence	
	labelled at each end by a donor fluorophore (red) and	
	an acceptor fluorophore (green) and (b) cleavage of	
	this peptide by a protease causing separation	
	of donor and acceptor fluorophores	55
Figure 1.19	Schematic representing the principle of an	
	intramolecularly quenched protein substrate (a)	
	and cleavage of this substrate by a suitable	
	protease to release fluorescent peptides (b)	58

.

Figure 3.1	Effect of incubation time on the final dye protein ratio	
	for different reaction mixture ratio conjugates	82
Figure 3.2	Fluorescence intensity of a reference block against	
	time after switching on the LS 50B spectrophotometer	84
Figure 3.3	Influence of temperature on the digestion of	
	fluorescein albumin by alkaline protease	86
Figure 3.4	Influence of pH and buffer system on the activity	
	of alkaline protease	87
Figure 3.5	Influence of carbonate buffer pH 11.0 concentration	
	on the activity of alkaline protease	88
Figure 3.6	Influence of incubation time on the conversion of	
	substrate to product	89
Figure 3.7	Influence of alkaline protease concentration on the	
	conversion of substrate to product	90
Figure 3.8	Time-dependent comparison of the activity of three	
	protease enzymes – alkaline protease, proteinase K, and	
	lpha-chymotrypsin – on the substrate fluorescein albumin	
	using fluorescence polarisation	91
Figure 4.1	Alternative scenarios following excitation of the conjugate	
	at the donor excitation wavelength	97
Figure 4.2	Spectral overlap of FITC (donor) and TMRITC (acceptor) 10	00

Figure 4.3	Absorbance spectra of fluorescein and	
	tetramethylrhodamine free dyes	101
Figure 4 4	Emission spectra of various	
	fluorescein.tetramethylrhodamine: $\alpha$ -casein conjugates	102
Figure 4.5	Emission spectra of	
	fluorescein: tetramethylrhodamine: $\alpha$ -casein conjugates	
	exhibiting an emission peak at the TMRITC wavelength	103
Figure 4.6	Fluorescence emission spectra demonstrating FRET	
	between fluorescein and tetramethylrhodamine	105
Figure 4.7	Spectral overlap of Cy 5 (donor) and Cy 5.5 (acceptor)	108
Figure 5.1	Absorbance spectra of BODIPY solution (a)	
-	and BODIPY conjugates (b)	116
Figure 5.2	Absorbance spectra showing effect of protease	
	digestion on BODIPY'α-casein	118
Figure 5.3	Absorbance spectra showing effect of denaturation	
	on BODIPY:α-casein	119
Figure 5.4	Excitation spectrum of BODIPY' $\alpha$ -casein	121
Figure 5 5	Effect of altering excitation wavelength on the	
	emission spectrum of BODIPY:α-casein	124
Figure 5 6	Effect of cyclodextrins and CHAPS on the	
	absorbance spectrum of BODIPY:α-casein	126

۱

List of figures

I

Figure 5 7	Effect of cyclodextrins and CHAPS on the	
	fluorescence emission spectrum of BODIPY:α-casein	127
Figure 5.8	Effect of storage conditions on the fluorescence	
	characteristics of BODIPY:α-casein	128
Figure 5 9	Stability of BODIPY: $\alpha$ -casein in aqueous solution	
	represented by the percentage decrease in	
	fluorescence intensity over time	129
Figure 5.10	Lineweaver-Burk plots for the proteolytic digestion	
	of BODIPY: $\alpha$ -casein by various proteases	131
Figure 5.11	Fluorescence intensity of a reference block and	
	output of the Xe lamp light source over time	
	for the F-4500 spectrophotometer	134
Figure 5.12	Fluorescence emission spectra showing enhancement	
	seen upon incubation of BODIPY: $\alpha$ -casein with a	
	protease enzyme (proteinase K)	137
Figure 5 13	Influence of BODIPY $\alpha$ -case n concentration on the	
	production of BODIPY when incubated with a variety of	
	proteases in a static system under optimised	
	conditions	138
Figure 5 14	Fluorescence intensity against enzyme concentration	
	to calculate the limit of detection of alkaline protease	140
Figure 5.15	Fluorescence intensity against enzyme concentration	
	to calculate the limit of detection of proteinase K	140

Figure 5.16	Fluorescence intensity against enzyme concentration	
	to calculate the limit of detection of $\alpha$ -chymotrypsin	141
Figure 5.17	Fluorescence intensity against enzyme concentration	
	to calculate the limit of detection of trypsin	141
Figure 5.18	Fluorescence intensity against enzyme concentration	
	to calculate the limit of detection of pepsin	142
Figure 5.19	Manifold for flow injection enzyme and enzyme	
	Inhibition assays	144
Figure 5.20	Sequence of events for the determination of	–
	proteases and their inhibitors in a flow system	145
Figure 5.21	Actual layout and equipment used for flow analysis	4.40
	protease and protease inhibition assays	140
Figure 5.22	Effect of temperature on the activity of pepsin	149
Figure 5 23	Effect of pH and buffer system on the activity of pepsin	150
Figure 5.24	Effect of buffer concentration on the activity of pepsin	151
Figure 5.25	Effect of incubation time on the activity of pepsin	152
Figure 5.26	Influence of pepsin concentration on the formation of	
	product when incubated with BODIPY. $\alpha$ -casein	153

Figure 5.27	Influence of substrate concentration on the production	
	of BODIPY fluorescence when incubated with alkaline	
	protease in a flowing system	154
Figure 5.28	Influence of substrate concentration on the production	
	of BODIPY fluorescence when incubated with a range	
	of proteolytic enzymes in a flowing system	155
Figure 5.29	Fluorescence intensity against enzyme concentration	
	to calculate the limit of detection of alkaline protease	156
Figure 5 30	Fluorescence intensity against enzyme concentration	
	to calculate the limit of detection of proteinase K	157
Figure 5 31	Fluorescence intensity against enzyme concentration	
•	to calculate the limit of detection of $\alpha$ -chymotrypsin	157
Figure 5.32	Eluorescence intensity against enzyme concentration	
	to calculate the limit of detection of pepsin	158
Figure 5 33	Influence of pre-incubation time on the inhibition	
riguie 0.00	of pepsin by pepstatin A	165
Eiguro 6 24	Influence of inculation time on the inhibition of panein	
Figure 5.54	by pepstatin A	166
<b>F</b> illing <b>5</b> .05		
Figure 5.35	Influence of substrate concentration on the inhibition	
	of pepsin by pepstatin A	167

Figure 5.36	Influence of pepsin concentration on the inhibition	
	of pepsin by pepstatin A	168
Figure 5 37	Influence of methanol concentration on the	
	fluorescence of BODIPY:a-casein	170
Figure 5.38	Influence of percentage concentration of methanol	
	on the emission wavelength of BODIPY: $\alpha$ -casein	171
Figure 5.39	Influence of percentage methanol on the	
	fluorescence of BODIPY:α-casein and pepsin	172
Figure 5.40	Influence of incubation time on the fluorescence	
	of BODIPY: $\alpha$ -casein incubated with pepsin in the	
	presence and absence of methanol	173
Figure 5.41	Influence of percentage methanol and DMSO on	
	the fluorescence of BODIPY $\alpha$ -casein	174
Figure 5.42	Influence of DMSO percentage concentration on the	
	optimum emission wavelength of BODIPY: $\alpha$ -casein	175
Figure 5 43	Inhibition of proteinase K by 3-nitrophenylboronic acid	176
Figure 5.44	Inhibition of $\alpha$ -chymotrypsin by aprotinin	177
Figure 5.45	Inhibition of alkaline protease by leupeptin	177
Figure 5.46	Inhibition of pepsin by pepstatin A	178

List of figures

Figure 5.47	Percentage inhibition against inhibitor concentration in	
	linear portion to calculate limit of detection	179
Figure 5.48	Percentage inhibition against inhibitor concentration in	
	linear portion to calculate limit of detection	180
Figure 5 49	Percentage inhibition against inhibitor concentration in	
	linear portion to calculate limit of detection	180
Figure 5.50	Percentage inhibition against inhibitor concentration in	
	linear portion to calculate limit of detection	181
Figure 6.1	Schematic of flow injection analysis system for	
	multi-determination protease inhibition assays	189
Figure 6.2	Sequence of events for multi-determination protease	
	inhibition assays using immobilised enzyme reactors	190
Figure 6 3	Reaction scheme for the immobilisation of enzyme	
	onto controlled pore glass	191
Figure 6.4	SEM pictures showing (a) 182 Å aminopropyl glass	
	and (b) an individual glass chip	192
Figure 6.5	Appearance of the enzyme reactors: top – packed glass	
	column, bottom – empty glass column	193
Figure 6.6	Fluorescence intensity of BODIPY: $\alpha$ -casein over	
	time when incubated in the presence and absence of	
	proteinase K – CPG	195

~

Figure 6.7	Enhancement of fluorescence seen when	
	BODIPY:α-casein was incubated with	
	proteinase K – CPG compared with CPG	196
Figure 6.8	Fluorescent peaks of BODIPY:α-casein injected	
	into a buffer stream with and without a raw CPG column	197
Figure 6.9	Absorbance of 0.1 % a-casein solution before (1) and	
	after (2) mixing with glutaraldehyde-activated CPG and	
	subsequent readings of freshly added buffer	199
Figure 6.10	Injection of naphthofluorescein with and	
	without 2.5 cm column	202
Figure 6.11	Injection of Cy5 with and without 2.5 cm column	202
Figure 6.12	Injection of $\alpha$ -casein with and without 2.5 cm column	203
Figure 6.13	Injection of BODIPY with and without 2.5 cm column	203
Figure 6 14	Injection of BODIPY: $\alpha$ -casein with and	
	without 2.5 cm column	204
Figure 6.15	Reaction scheme for the immobilisation of enzyme	
	onto UltraLink Biosupport Medium	206
Figure 6.16	Calibration graph of proteinase K concentration	
	using the Bicinchoninic Acid assay	208

List	of	figures
------	----	---------

Figure 6.17	Injection of BODIPY through both an empty column and a column packed with proteinase K – UltraLink	209
Figure 6.18	Injection of BODIPY $\alpha$ -casein through both an empty column and a column packed with proteinase K – UltraLink	209
Figure 6.19	Injection of $\alpha$ -casein through both an empty column and a column packed with proteinase K – UltraLink	210
Figure 6.20	Schematic of flow system used for soluble multi-determination assays	213
Figure 6.22	Sequence of events for multi-determination protease inhibition assays	214
Figure 6.22	Separation of two enzyme peaks using 25 cm and 399 cm length tubing	216
Figure 6.23	Multi-determination using two proteolytic enzymes – pepsin and proteinase K – and the selective inhibition of these enzymes by pepstatin A	217
Figure 7.1	Possible set-ups for multi-determination assays: (a) one inhibitor, three enzymes with three different pH optima, (b) one inhibitor, three enzymes with two different pH optima	229
Figure 7.2	Schematic diagram of the chip designed for enzyme assays in [307]	231
Figure 7.3	Schematic diagram showing the set-up for a confocal fluorescence detection on a chip, taken from [331]	235

1

Figure B	Appearance of the F-4500 in our laboratory	289
Figure C	Internal block diagram of the F-4500	290
Figure E	Appearance of the LS 50B	292
Figure F	Optical layout of the LS 50B fluorescence spectrophotometer	293

- ×

\_

# LIST OF TABLES

,

Table 1.1	Worldwide sales figures for selected angiotensin	
	converting enzyme inhibitors as drugs	46
Table 2 1	Reagents used and their respective supplier	67
Table 3.1	Initial reaction mixture and calculated final dye:protein	
	ratios for fluorescein:albumin conjugates	81
Table 3.2	Assay parameters and optimum conditions for three	
	protease enzymes using the substrate fluorescein:albumin	85
Table 5.1	Initial reaction mixture and calculated final dye protein	
	ratios for BODIPY:α-casein conjugates	115
Table 5 2	Initial reaction mixture and calculated final dye:protein	
	ratios for various BODIPY:α-casein conjugates	122
Table 5.3	$K_{M}$ and $V_{max}$ values obtained experimentally for various	
	proteolytic enzymes on the substrate BODIPY. $\alpha$ -casein	132
Table 5.4	Assay parameters studied and optimum conditions	
	for various proteolytic enzymes using the fluorogenic	400
	substrate BODIPY'a-casein in a static system	136
Table 5 5	Limits of detection determined experimentally for various	
	proteolytic enzymes using the substrate BODIPY: $\alpha$ -casein	
	in a static system	143

Table 5 6	Assay parameters studied and optimum conditions	
Υ.	for various proteolytic enzymes using the fluorogenic	
	substrate BODIPY: $\alpha$ -casein in a flow system	148
Table 5 7	Limits of detection determined experimentally for various	
	proteolytic enzymes using the substrate BODIPY' $\alpha$ -casein	
	in a flowing system	158
Table 5.8	Comparison of assay incubation times for various	
	proteolytic enzymes in static and flow system formats	159
Table 5.9	Comparison of enzyme limits of detection between two	
	different methods using similar protease substrates	161
Table 5.10	Assay parameters studied and optimum conditions	
	for various enzyme – inhibitor pairs using the fluorogenic	
	substrate BODIPY: $\alpha$ -casein in a flow system	164
Table 5.11	Limits of detection determined experimentally for various	
	protease inhibitors using the substrate BODIPY $\alpha$ -casein	
	In a flowing system	181
Table 6.1	Amount of different materials flowing through a packed	
	proteinase-UltraLink column compared with an	
	empty column	210
Table A	Supplier names and addresses	285
Table D	Instrument parameters of the F-4500 and LS 50B	291

# LIST OF ABBREVIATIONS

3-nitrophenylboronic acid
giotensin converting enzyme
Bicinchoninic acid
Controlled-pore glass
Dye:protein
N, N-dimethylformamide
Dimethylsulphoxide
Flow injection analysis
Fluorescein isothiocyanate
Fluorescence polarisation
ce resonance energy transfer
ıman ımmunodeficiency virus
. High throughput screening
Light emitting diode
Limit of detection
Photo-multiplier tube
Quantum yield
thy!rhodamine isothiocyanate
Ultraviolet
. Molar extinction coefficient

# TABLE OF CONTENTS

Acknowledgements	i
Abstract	ii
Research aims and scope of work	İlİ
List of figures	v
List of tables x	viı
List of abbreviations	dx
Table of contents	xx

Chap	oter 1	Introducti	ion 1
1.1	Fluor	escence .	2
	1.1.1	Theory of	fluorescence 2
	1.1.2	Advantage	es of fluorescence 4
	1.1.3	1.1.3 Long wavelength fluorescence	elength fluorescence 4
	1.1.4	Fluoresce	nt labels
		1,1.4 1	Fluorescein isothiocyanate (FITC)
		1.1.4.2	Tetramethylrhodamine isothiocyanate (TMRITC) 8
		1.1.4.3	BODIPY® fluorophores 9
		1.1.4.4	Cyanine dyes (Cy™ 5 and Cy™ 5 5
			bisfunctional dyes) 10
1.2	2 Fluorescence polarisation		olarisation 12
	1.2.1	Theory of	fluorescence polarisation 12
		1.2.1.1	Dipole model of absorption and
			emission 14

### Table of contents

		1.2.1.2	Depolarisation	14
		12.13	Relationship between fluorescence	
			polarisation and molecular weight	16
	1.2.2	Advantages of	of fluorescence polarisation	17
	1.2.3	Disadvantage	es of fluorescence polarisation	17
1.3	Fluor	escence reso	nance energy transfer	18
	1.3.1	Theory of FR	ET	18
	132	Conditions for	or FRET	19
	133	FRET donor	/ acceptor pairs	20
	1.3.4	Advantages	of FRET	21
	1.3.5	Disadvantage	es of FRET	21
1.4	Flow	injection ana	lysis (FIA)	22
	1 4.1	History of flo	w injection analysis	22
	1.4.2	Theory of flo	w injection analysis	23
1.5	High	throughput s	creening (HTS)	25
	1.5.1	Concept of h	igh throughput screening	25
	1.5.2	History of sc	reening	26
	1.5 3	The drug dis	covery process	27
	1.5.4	Miniaturised	high throughput screening	29
1.6	Enzyr	nes		30
	1.6.1	Introduction f	to enzymes	30
	1.6.2	Factors affecting enzyme activity		
	1.6.3	Enzyme kine	tics	32
	1.6.4	Proteolytic ei	nzymes	35
		1.6.4.1	Alkaline protease	37
		164.2	Proteinase K	37
		1.6 4.3	Chymotrypsin	38

		1.6.4.4	Trypsin	38
		1.6 4.5	Pepsin	38
1.7	Enzyn	ne immobilis	ation	39
	1.7.1	Why immobil	lise enzymes?	39
	1.7.2	Immobilisatio	on methods	40
		1.7.2.1	Covalent binding method	41
	1.7.3	Modification	of enzyme properties after immobilisation	42
	1.7 4	Immobilised	enzymes in FIA	43
1.8	Prote	ase assays .		44
	1.8.1	Importance of	of protease assays	44
	1.8.2	General prot	ease assay methods	48
	1.8.3	Fluorescent	protease assay methods	49
		1 8.3.1	Fluorescence polarisation protease assays	49
		1.8.3.2	Fluorogenic protease assays	54
		1.8.3 2.1	Fluorogenic FRET protease assays	55
		1.8.3 2.2	Quenched over-labelled fluorogenic protease	
			assays	57
1.9	Enzyr	ne inhibition		61
	191	Types of inh	bition	61
		1.9.1.1	Reversible inhibition	61
		1.9.1.2	Irreversible inhibition	63
		1.9.1.3	Product inhibition	63
	1.9.2	Protease inh	ibition	64
		1.9.2.1	Aprotinin	64
		1.9 2.2	Leupeptin	65
		1.9.2.3	3-Nitrophenylboronic acid	65
		1.9 2.4	Pepstatin A	65

Chapter 2		Materials, ir	strumentation and general procedures	66
2.1	Mater	ials and inst	rumentation	67
	2 1.1	Reagents		67
	2.1.2	Cells		69
	2.1.3	Chromatogra	aphic columns	69
	2.1 4	Flow system	equipment	69
	2.1.5	Pipettes		70
	2.1.6	Spectromete	rs	70
	2.1.7	Supplier nam	nes and addresses	70
	2.1.8	Temperature	e control	70
2.2	Gene	ral procedure	es	71
	2.2.1	Buffer preparation		
	2.2.2	pH measurement		
	2.2.3	Water purification		
	2.2.4	Mass detern	nination	71
	2.2.5	Pipetting		72
	2.2.6	Fluorescenc	e measurement	72
		2.2 6.1	F-4500 Spectrophotometer	72
		2262	LS 50B Spectrophotometer	72
	2.2.7	Fluorescenc	e polarisation measurement	73
	2.2.8	Absorbance	measurement	73
	2.2.9	Flow system	measurements	74
	2.2.10	Preparation	of dye protein conjugates	74
	2.2.11	Final dye:pro	otein ratio calculations	75
	2.2.12	lmmobilisati	on of enzyme onto CPG	76
	2.2.13	Immobilisati	on of enzyme onto Pierce UltraLink biosupport	

l

#### Table of contents

		medium		77
	2.2.14	BCA ass	ay for determination of protein concentration	77
Chap	ter 3	Experim	ental: fluorescence polarisation protease	
		assays		78
3.1	Introc	luction .		79
3.2	Prepa	aration of	fluorescein conjugates	80
	3.2.1	Procedur	es	80
	3.2.2	Results		80
3.3	Fluor	escence	oolarisation protease assays	83
	3.3.1	Procedu	es	83
	3.3 2	Results		85
		3.3.2.1	Effect of temperature	86
		3.3 2.2	Effect of pH and buffer system	87
		3.3 2.3	Effect of buffer concentration	88
		3324	Effect of incubation time	89
		3.3.2.5	Effect of enzyme concentration	90
		3.3.2.6	Comparison of the activity of three proteolytic	
			enzymes using fluorescence polarisation	91
3.4	Discu	ission		92

~

١

Chap	ter 4	Experimental: double labelled FRET based protease			
		assays	•••••••••••••••••••••••••••••••••••••••	95	
				00	
4.1	Introd	luction		90	
4.2	Invest	tigation into f	the use of fluorescein and		
	tetramethylrhodamine for FRET protease assays				
	4.2.1	Procedures		99	
	4 2.2	Results		100	
		4 2.2.1	Analysis of the overlapping of fluorescein		
			(donor) and tetramethylrhodamine (acceptor)		
			excitation and emission spectra	100	
		4.2.2.2	Determination of optimum donor:acceptor.protein		
			ratios for use in FRET protease assays	102	
		4.2.2.3	Analysis of the 5 75:1		
			FITC:TMRITC:α-casein conjugate	104	
4.3	Inves	tigation into t	the use of Cy 5 and Cy 5.5 for FRET		
	prote	ase assays		107	
	4.3.1	Procedures		107	
	4.3.2	Results		108	
		4 3.2.1	Analysis of the overlapping of Cy 5 (donor) and		
			Cy 5.5 (acceptor) excitation and emission		
			spectra	108	
		4.3.2.2	Analysis of the Cy 5.Cy 5.5:α-casein		
			conjugates	109	

4.4	Discussion		110
-----	------------	--	-----

Chapter 5		Experimental: single labelled substrate protease and		
		protease in	hibition assays	112
<sup>~~</sup> 5.1	Introd	luction		113
5.2	Prepa	aration of BO	DIPY conjugates	114
	5 2.1	Procedures		114
	5.2.2	Results		115
		5.2 2.1	Effect of protease digestion on the absorption	
			characteristics of BODIPY:α-casein	118
		5 2.2.2	Effect of denaturation on the absorption	
			characteristics of BODIPY:α-casein	119
		5.2.2.3	Effect of the number of BODIPY dyes attached	
			to $\alpha$ -casein on the absorbance spectra of the	
			conjugates	120
		5.2 2.4	Comparison of the fluorescence excitation	
			spectrum with the absorbance spectrum of	
			BODIPY:α-casein	120
		5.2.2.5	Re-evaluation of D:P calculations	121
		5.2.2.6	Calculation of percentage quenching in the 2.35	
			BODIPY α-casein conjugate	123
		5 2.2.7	Effect of altering excitation wavelength on the	
			emission spectrum of BODIPY:α-casein	124
		5.2.2.8	Investigation into the effect of additives on the	
			absorbance and emission spectra of	
			BODIPY:α-casein	125
		5.2 2.9	Stability of BODIPY:α-casein	127
_	_			
5.3	Prote	ase kinetics		130
	5.3.1	Procedures		130

	5.3.2	Results		131
51	Proto	aco accave il	a static evetom	133
5.4	5/1	Drocoduros		133
	5.4.2	Poculte		135
	J 4.Z	E A 2 4	Optimization of protocol access conditions	135
		5.4.2.1	Optimisation of protease assay conditions	100
		5.4.2.2	Calibration curves for BODIPY:α-casein	138
		5.4.2.3	Determination of limits of detection	139
5.5	Prote	ase assays iı	n a flowing system	144
	5 5.1	Procedures		144
	552	Results		147
		5.5.2.1	Effect of temperature	149
		5.5.2.2	Effect of pH and buffer system	150
		5.5.2.3	Effect of buffer concentration	151
		5.5.2.4	Effect of incubation time	152
		5.5 2 5	Effect of pepsin concentration on the reaction	
			rate	153
		5.526	Calıbration graphs for BODIPY: $\alpha$ -casein in	
			the flowing system	154
		5.5.2.7	Determination of limits of detection	156
		5.5.2.8	Comparison with previously published work	160
				400
5.6	Prote	ase inhibitio	n assays	162
	5.6.1	Procedures		162
	5.6.2	Results		162
		5.6 2.1	Effect of pre-incubation time	165
		5.6 2.2	Effect of incubation time	166
		5.6.2.3	Effect of substrate concentration	167
		5.6.2.4	Effect of enzyme concentration	168
		5 6.2 5	Problems with inhibitors	169

		5.6.2.6	Generation of inhibition curves	176
		5.6.2.7	Determination of limits of detection	179
5.7	Discu	ssion		183
Chant		Experiment	ale multi determination protocoa inhibition	
Chapt	ero	cxperimenta	a. multi-determination protease mainfulon	186
		assays		100
6.1	Introd	uction		187
6.2	Multi-	determinatio	n protease inhibition assays using	
	contro	olled-pore gla	ass as a support	188
	6.2.1	Procedures		188
	6.2 2	Results		194
		6 2.2.1	Absorbance experiments to investigate sticking	
			of $\alpha$ -casein to glutaraldehyde activated CPG	199
		6.2.2 2	Blocking of free aldehyde groups using	
			ethanolamine	200
		6.2.2 3	Evaluation of CPG with a larger pore size	
			of 972 Å	201
		6 2.2.4	Evaluation of shorter enzyme reactors	201
6.3	Multi-	determinatio	n protease inhibition assays using Pierce	
	Ultral	.ink Medium	as a support	205
	6.3.1	Procedures		205
	6.3.2	Results		207
		6.3.2.1	Determination of amount of proteinase K	
			immobilised onto UltraLink Biosupport Medium	207
		6.3.2.2	Evaluation of UltraLink as a support for	
			multi-determination protease inhibition assays	

			in a flowing system	208
6.4	Multi-	determinatio	on protease inhibition assays using	
	solub	le enzyme .		212
	6.4.1	Procedures		212
	6.4.2	Results		215
		6 4.2.1	Determination of length of flow tubing required	
			for separation of peaks	215
		6 4.2 2	Multi-determination protease inhibition assays	217
6.5	Discu	ission		219
Chap 7.1	oter 7	Conclusion	s and future work	221
	CONC	lusions	•••••••••••••••••••••••••••••••••••••••	222
7.2	Futur	lusions e work		222 226
7.2	Futur 7.2.1	lusions e work Fluorescenc	e polarisation	<b>222</b> <b>226</b> 226
7.2	Futur 7.2.1 7.2.2	lusions e work Fluorescenc Double-labe	e polarisation	<b>222</b> <b>226</b> 227
7.2	Futur 7.2.1 7.2.2 7.2.3	lusions e work Fluorescenc Double-labe Single-label	e polarisation lled substrate	<b>222</b> <b>226</b> 227 227
7.2	Futur 7.2.1 7.2.2 7.2.3 7.2 4	lusions e work Fluorescenc Double-labe Single-label Multi-determ	e polarisation Iled substrate led substrate	<b>222</b> <b>226</b> 227 227 227
7.2	Futur 7.2.1 7.2.2 7.2.3 7.2 4 7.2.5	lusions e work Fluorescenc Double-labe Single-label Multi-determ Miniaturised	e polarisation lled substrate led substrate nination protease and protease inhibition assays	<ul> <li>222</li> <li>226</li> <li>227</li> <li>227</li> <li>227</li> <li>230</li> </ul>

Appendices ...... 284

# **CHAPTER 1**

# Introduction

## 1.1 FLUORESCENCE

Fluorescence spectroscopy is an extremely powerful analytical tool that has seen a remarkable growth in its use over the past 15 years in a wide range of areas from clinical chemistry to environmental monitoring. There are numerous sources of information on the theory of fluorescence and its applications [1, 2, 3, 4] with the consequence that the theory will not be discussed in great detail here. An excellent paper covers the basic principles of fluorescence spectroscopy [5] and a recent book by Lakowicz provides a wealth of knowledge on the basic aspects on fluorescence spectroscopy, as well as covering the many experimental and instrumental advances that have been made during the last 16 years [6]. Reviews covering new developments in the theory and practice of molecular luminescence for chemical analysis are published every two years [7, 8] and yearly reviews on advances in atomic emission, absorption, and fluorescence spectrometry, and related techniques are also published [9, 10].

### 1.1.1 Theory of fluorescence

The sequence of events that occur in fluorescence are shown in figure 1.1 and are as follows:

- 1. A quantum of light is absorbed by a dipole (a)
- 2. Rapid radiationless relaxation (internal conversion) takes the dipole to the thermally relaxed S<sub>1</sub> level (b)
- 3. The dipole retains this energy for a certain period of time, generally designated as the lifetime  $(\tau)$

2

~ 5

4. The excess energy is dissipated through radiationless de-excitation (c, d, and e) caused predominantly through vibrational and rotational modes and collisions with solvent molecules, or by photon emission (f, g). When this photon emission is from a singlet electronic state, it is referred to as fluorescence (f). Conversely, when the emission is from a triplet state, phosphorescence is the result (g).

Figure 1.1 – Jablonski diagram showing absorption and emission processes of fluorescence and phosphorescence.  $S_0$  = ground singlet state,  $S_1$  and  $S_2$  = first and second singlet levels respectively,  $T_1$  and  $T_2$  = first and second triplet levels respectively


#### 1.1.2 Advantages of fluorescence

When compared with conventional absorbance-based techniques, fluorescence is several orders of magnitude more sensitive and more selective. Increase in sensitivity arises because the emitted radiation is measured directly and can be increased simply by increasing the incident power. Moreover, as fluorescence is measured at right angles to the incident beam, it is considered a zero background technique. Absorbance however, is a measure of the difference between incident and transmitted intensities i.e., signals are measured as a small change in a large background.

Selectivity in fluorescence-based techniques is also much higher than in absorbance-based approaches. All molecules absorb photons. However, only a few fluoresce and fewer still phosphoresce at ambient temperatures. Thus the rarity of luminescent materials makes fluorescence an attractive technique for the resolution of trace fluorescent components in complex mixtures. In addition, the ability to employ both excitation and emission wavelengths as selectivity parameters gives fluorescence increased selectivity.

The ease of handling fluorophores compared with the expense and difficulties of handling and disposing radioactive material makes fluorescence an attractive alternative to radioactivity.

#### 1.1.3 Long wavelength fluorescence

Fluorescence spectrometry has been accepted for many years as a major technique for trace analysis, but in real samples the sensitivity of fluorescence is not as high as that seen in pure solutions that contain the same fluorophore.

molecules being effectively non-fluorescent. Thus, fluorophores with moderate molar absorptivities and quantum yields can give excellent limits of detection because of the low backgrounds due to so few fluorophores that absorb and emit at these wavelengths.

The development of semiconductor-based detectors, photodiodes, and inexpensive long wavelength laser diodes and LEDs has allowed the reduction of detector noise and an enhancement of the fluorescence signal. There is therefore the opportunity for simple, robust but sensitive instruments based on solid-state light sources and photodiode detectors and cheap and effective fiber optics. Both the simplicity and sensitivity of long wavelength fluorescence thus make it an obvious choice as a detection method.

Further discussions of the advantages of long wavelength fluorescence and its applications can be found [15, 16].

## 1.1.4 Fluorescent labels

A variety of fluorescent labels have been used in this study, and the structure, advantages, and disadvantages of each fluorophore are discussed below. The energy capture efficiency of a fluorescent dye is expressed as the extinction coefficient,  $\varepsilon$ , and usually ranges from 10 000 – 250 000 cm<sup>-1</sup> M<sup>-1</sup>. The emission efficiency is expressed as the quantum yield, QY, and usually ranges from 0 05 – 1.0. Quantum yield is strongly influenced by the local environment and so is not usually reported as a 'constant' like the extinction coefficient. The best fluorophores will have both a high extinction coefficient and high quantum yield.

# 1.1.4.1 Fluorescein isothiocyanate (FITC)

A more detailed description of FITC can be found elsewhere [17].





# Advantages:

- high extinction coefficient (73 000 cm<sup>-1</sup> M<sup>-1</sup>)
- excellent fluorescence quantum yield (~ 0.9)
- good water solubility

# **Disadvantages:**

- relatively high rate of photobleaching
- pH-sensitive fluorescence that is significantly reduced below pH 7.0 (pK<sub>a</sub> ~ 6.4)

# 1.1.4.2 Tetramethylrhodamine isothiocyanate (TMRITC)

Further reading on this fluorophore can be found [18].



Figure 1.3 – Structure of tetramethylrhodamine, mixed isomers

# Advantages:

- good photostability
- insensitive to solvent polarity

# **Disadvantages:**

• prone to aggregation

This is because of background-scattered light and fluorescence signals from endogenous sample components. This problem is particularly prevalent in biological samples where high levels of proteins and other natural polymers cause intense light scattering, and which commonly contain fluorophores whose spectra overlap those of the target fluorophore. Substances found in blood plasma, for example proteins and amino acids, exhibit fluorescence emission at 350 nm, and protein-bound bilirubin shows emission at 520 nm.

A major source of errors in all fluorescence measurements is interferences due to scattered light. Scatter is a property of the solvent or particulate matter in the sample. Some photons are uniformly scattered without change in frequency due to collisions with the solvent molecules. The emitted light has the same wavelength as the exciting light since the absorbed and emitted photons are of the same energy. This scattering is termed Rayleigh scattering and occurs at all wavelengths. Its intensity, however, varies in proportion to  $\lambda^4$ , so its effect can be minimised by working at longer wavelengths [11]. Raman scattering is a source of interference that occurs in aqueous solutions, which are typically used for biological samples. Raman scatter always occurs at a constant wavenumber difference from the incident light. With longer excitation wavelengths, the Raman band decreases in intensity, becomes further displaced from the Rayleigh scatter band, and has a wider bandwidth [12]. At longer wavelengths therefore, both Rayleigh and Raman scatter are much reduced.

Background fluorescence originating from endogenous fluorescent species and from Raman and Rayleigh scattering can greatly limit the potential sensitivity of the analysis. However, long wavelength fluorescence detection in the region 600 - 1000 nm has proven to reduce interference [13].

Photochemical theory predicts a sharp diminution in the quantum yields of fluorophores as their emission wavelengths increase [14]. In practice the long wavelength region boasts a few fairly intense fluorophores, with almost all other

#### 1.1.4.3 BODIPY® fluorophores

The BODIPY dyes are dibenzopyrrometheneboron difluoride compounds and were synthesised for the first time by Treibs and Kreuzer [19], in 1968. Later an ionic derivative was synthesised by Worries *et al.* [20], and several reactive forms have been prepared recently by Kang and Haugland [21]. All the BODIPY dyes have a difluoro-bora-diaza-indacene structure indicated by the red-circled moiety shown in figure 1.4. Different substituents are then added onto this basic structure to produce different BODIPY dyes with different excitation and emission wavelengths. The particular BODIPY dye that has been used in this study is shown in figure 1.4. Numerous sources of information can be found on these fluorophores [22, 23, 24].





#### Advantages:

 high extinction coefficients (100 300 cm<sup>-1</sup> M<sup>-1</sup> for BODIPY<sup>®</sup> 630/650-X, SE))

- high fluorescence quantum yields (often approaching 1.0, even in water)
- spectra that are insensitive to solvent polarity and pH due to lack of ionisable groups (hence no ionic charge)
- narrow emission bandwidths
- greater photostability than fluorescein they remain intensely fluorescent even when constantly illuminated

#### **Disadvantages:**

- expensive (~£153 per 5 mg BODIPY® 630/650-X, SE)
- small Stokes shift (15 nm)

#### **1.1.4.4** Cyanine dyes (Cy<sup>™</sup> 5 and Cy<sup>™</sup> 5.5 bisfunctional dyes)

Papers are available for more comprehensive reading on the cyanine dyes Cy 5 and Cy 5.5 [25, 26, 27].







Figure 1.6 – Structure of Cy™ 5.5 bisfunctional dye

## Advantages:

- high extinction coefficients (250,000 cm<sup>-1</sup> M<sup>-1</sup>)
- quantum yields of >0 28
- highly water soluble

# **Disadvantages:**

- small Stokes shift (21 nm for Cy 5 and 19 nm for Cy 5.5)
- expensive (~£25.00 per mg)

#### 1.2 FLUORESCENCE POLARISATION

The practical application of fluorescence polarisation to the determination of proteolytic enzymes has been carried out in order to evaluate the potential of this detection method for use in protease high throughput screening assays. There are numerous references on the theory of fluorescence polarisation and its applications [28, 29, 30, 31, 32]. The theory of fluorescence polarisation is crucial to the understanding of the idea behind these assays and will therefore be described.

#### 1.2.1 Theory of fluorescence polarisation

The concept of molecular movement and rotation is the basis of fluorescence polarisation. To fully understand the theory of fluorescence polarisation, consider the wave model of light as shown in figure 1.7.





This single ray may be characterised by the orientation of its electric vector within a co-ordinate system. Thus, if one looked directly into an oncoming ray, the electric vector would be at some angle  $\theta$ . Likewise, a beam composed of a number of rays could be characterised by the angular distribution of its electric vectors. Two extreme situations are possible: the vectors can be evenly (randomly) distributed (figure 1.8a), or they can be parallel (1.8b and 1.8c).





These two situations can be differentiated experimentally by viewing light through a polariser, which allows maximum transmission of light when it is aligned parallel to the electric vector and zero transmission when it is perpendicular. Therefore, as the polariser is turned through 180° in 1.8a, the intensity remains constant. However, in 1.8b the intensity would be a maximum with the polariser vertical and zero with it horizontal, the converse being true for 1.8c.

It is this experimental observation that allows one to define the polarisation of light as:  $p = \frac{(V-H)}{(V+H)}$  (equation 1), where p = polarisation, V = vertical component of emitted light, H = horizontal component of emitted light, when excited by vertical plane polarised light.

As can be seen by equation 1, P ("polarisation unit") is a dimensionless entity and is not dependent on the intensity of the emitted light or on the concentration of the fluorophore. This is a fundamental power of fluorescence polarisation. For convenience the term "mP" (pronounced "millipee") is used, where 1 mP equals one thousandth of a P. Applying equation 1 to the situation in figure 1.8a the two components (V and H) are always equal and p = 0. In 1.8b, H = 0 and p = 1, while in 1.8c V = 0 and p = -1. When p = 0, the light is called *unpolarised* or *natural light*. With  $p = \pm 1$ , it is *completely polarised*, whereas with  $1 \ge p \ge -1$ , but not equal to 0, it is *partially polarised*.

## **1.2.1.1** Dipole model of absorption and emission

Light is absorbed by and emitted from molecules via electric dipoles, which have certain orientations relative to the structure of the molecule. Maximum absorption of light occurs when the absorption dipole is parallel to the electric vector of the light. Specifically, the probability of absorption is proportional to  $\cos^2 \theta$  where  $\theta$  is the angle between the dipole and the electric vector. Therefore, absorption is a maximum when  $\theta = 0^\circ$  and is zero when  $\theta = 90^\circ$ .

Emission from a dipole can occur with the electric vector at any angle to the dipole, but the probability is proportional to  $\sin^2 E$  where E is the angle between the dipole and the electric vector.

#### 1.2.1.2 Depolarisation

The sequences of events that occur in fluorescence have already been discussed in section 1.1.1. In each of these processes, changes can occur such

that the polarisation of the fluorescent light is different from that of the exciting light.

The absorption process itself is the first source of depolarisation. As discussed in section 1.2.1.1 when undergoing excitation with vertically polarised light the vertical dipoles will be preferentially excited. In reality emission from these dipoles could give only partial polarisation, as non-vertical dipoles do absorb some of the vertically polarised excited light.

If the emission dipole is at an angle to the absorption dipole, a second source of depolarisation is evident which is dependent upon the angle between the two dipoles, the larger the angle the greater the depolarisation.

A third important depolarising factor is the Brownian rotation of the molecule. Rotation of the molecule will result in rotation of the emitting dipole to a new orientation. Therefore, if the excited state lifetime is comparable to the time required for Brownian rotation to occur over an appreciable angle, significant depolarisation results.

Depolarisation can also result from non-radiative energy transfer. Transfer takes place preferentially between parallel dipoles, but nonparallel dipoles can also undergo transfer with resulting depolarisation.

A final mechanism of depolarisation is trivial reabsorption the light emitted from one molecule is absorbed by another molecule, which then reemits it at a different orientation.

The term  $p_0$  is defined as the intrinsic polarisation of a molecule in the absence of any depolarisation due to Brownian rotation or intermolecular processes.

# 1.2.1.3 Relationship between fluorescence polarisation and molecular weight

Rate of rotation of a molecule is described by the Stokes equation:  $\rho = \frac{3 \eta V}{RT}$  (equation 2), where  $\rho$  = rotational relaxation time,  $\eta$  = viscosity of the medium, V = molecular volume, R = gas constant, and T = temperature in degrees Kelvin.

The molecular volume is defined by: V = vM (equation 3), where V = molecular volume, v = partial specific volume, and M = molecular weight in Daltons.

From equations 2 and 3 it can be seen that the higher the molecular weight of a molecule the higher the molecular volume and therefore the higher the rotational relaxation time will be.

The Perrin equation (equation 4), which was first described in 1926 [33], describes the relationship between the observed fluorescence polarisation, the limiting polarisation, the fluorescence lifetime of the fluorophore, and its rotational relaxation time:  $\left(\frac{1}{P} - \frac{1}{3}\right) = \left(\frac{1}{P_0} - \frac{1}{3}\right) \left(1 + \frac{3\tau}{\rho}\right)$  (equation 4), where P = polarisation,

 $P_0$  = limiting polarisation,  $\tau$  = fluorescence lifetime, and  $\rho$  = rotational relaxation time. From equation 4 it can be seen that the smaller the fluorescence lifetime, the higher the fluorescence polarisation will be. The smaller the rotational relaxation time, the smaller the fluorescence polarisation will be.

Combining the Stokes equation (equation 2) and the Perrin equation (equation 4), the relationship between the molecular weight of a molecule and its fluorescence polarisation is achieved – the fluorescence polarisation of a fluorophore is proportional to its rotational relaxation time, which in turn depends

د

upon its molecular volume and hence its molecular weight (1/P is proportional to 1/M). The smaller the molecule, the faster it rotates, and so the lower the fluorescence polarisation will be This relationship stands provided that the temperature and solution viscosity remain constant.

## 1.2.2 Advantages of fluorescence polarisation

- fluorescence polarisation is a homogenous technology requiring no immobilisation, washing or separation
- measurements are very rapid
- reagents are stable
- large batches may be prepared which results in high reproducibility
- fluorescence polarisation has proven to be highly automatable
- in theory fluorescence polarisation is independent of intensity, so is relatively immune to the inner filter effect and so works in coloured solutions and cloudy suspensions
- fluorescence polarisation is relatively insensitive to instrument changes such as drift, gain settings, lamp changes, etc
- fluorescence intensity is obtained in addition to polarisation, if required

## **1.2.3 Disadvantages of fluorescence polarisation**

- fluorescence polarisation is instrument dependent, the quality of the results is directly linked to the quality of the instrument
- the difference between the size of the starting material and the end material should be significant enough to give a reasonable polarisation change

# **1.3 FLUORESCENCE RESONANCE ENERGY TRANSFER (FRET)**

## 1.3.1 Theory of FRET

Much of the work presented in this thesis relies upon the phenomenon of FRET. As numerous sources of information are available on the theory of FRET [34, 35, 36, 37], it will not be discussed in great detail here. However, as this technique is crucial to understanding the theory behind much of the practical work, a brief description will be given.

FRET is a distance-dependent interaction between the electronic excited states of two dye molecules in which energy from an electronically excited molecule (the donor fluorophore) is shifted to a neighbouring molecule (the acceptor fluorophore), returning the donor molecule to its ground state without the emission of a photon. Once excited, the acceptor can undergo de-excitation by the same emissive and non-emissive processes that occur upon excitation of a fluorophore

Although the phenomenon of resonance energy transfer was observed by Perrin at the beginning of the 1900's, it was Forster who proposed a theory describing long-range molecular interactions by resonance energy transfer in the late 1940's [38]. Forster derived a transfer rate equation that related the interchromophore distance and the spectroscopic properties of the chromophores. The rate constant for energy transfer is proportional to the inverse sixth power of the intermolecular separation, making it useful over distances comparable with the dimensions of biological macromolecules [39]. Thus, FRET is an important technique for investigating a variety of biological phenomena where markers that track physical proximity are necessary or useful. As the efficiency of FRET is proportional to  $r^{-6}$  where r = distance of intermolecular separation, not only does energy transfer only occur over short ranges, but it definitely does not occur over long ranges Energy transfer is thus an excellent method for use in protease and protease inhibition assays where proteolytic action completely abolishes the energy transfer between fluorophores that occurs in the intact protein. Digestion of the protein substrate separates the fluorophores, increasing the distance of intermolecular separation and so completely preventing the phenomenon of energy transfer from being able to occur.

### 1.3.2 Conditions for FRET

\_

A number of conditions are crucial for the mechanism of FRET to occur. The primary requirement for FRET is that the energy lost by de-excitation of the donor molecule be matched by the energy required for excitation of the acceptor. In other words, the absorption spectrum of the acceptor molecule must overlap the emission spectrum of the donor molecule, as shown in figure 1.9.



The spectral overlap seen in figure 1.9 can involve two chemically different molecules as donor and acceptor (as is the case in the protease substrate

described in chapter 4), or two identical groups due to the common tendency for the excitation and emission spectra of fluorophores to overlap (as is the case in the protease substrate described in chapter 5).

Donor and acceptor molecules must also be in close proximity – typically 10 - 100 Å. FRET is a distance-dependent energy transfer, and the distance at which energy transfer is 50 % efficient (i e. 50 % of excited donors are deactivated by FRET) is defined by the Forster radius (R<sub>0</sub>). The magnitude of R<sub>0</sub> is dependent upon the spectral properties of the donor and acceptor dyes (the magnitude of FRET is dependent upon the quantum yield of the donor), and for good donor-acceptor pairs, R<sub>0</sub> values of 30 to 60 Å are common.

One other fundamental requirement for FRET is that donor and acceptor transition dipole orientations must be approximately parallel.

# 1.3.3 FRET donor / acceptor pairs

In recent years, numerous donor / acceptor pairs have been proposed; these include 5-(2-aminoethyl)aminonaphthalene-1-sulphonyl (EDANS) / 4-(4-dimethylaminophenylazo)benzoyl (DABCYL) [40], fluorescein / rhodamine [41], pairs based upon the BODIPY® dyes [42], coumarin / fluorescein [43], o-aminobenzoyl (Abz) / 3-nitrotyrosine [44], as well as near-infrared squaraine dyes Sq635 / Sq660 [45]. Different donor / acceptor dye pair combinations have previously been discussed with regard to dye pairs with superior spectroscopic properties [46].

Non-fluorescent acceptors exist which accept energy from a donor without any resulting fluorescence emission. These acceptors as a group are known as "dark quenchers", for example DABCYL. These dyes have the particular advantage of

eliminating the potential problem of background fluorescence resulting from direct acceptor excitation.

# 1.3.4 Advantages of FRET

- simultaneous observation of quenching and enhancement effects allows specific effects to be distinguished from non-specific environmental effects on fluorescence intensity
- distances or changes in distances can be measured in a complex of molecules
- structural information can be gained

# 1.3.5 Disadvantages of FRET

- direct excitation of acceptor causes a high background
- the availability of adequate donor / acceptor pairs is somewhat limited, especially at near infrared wavelengths
- when measuring a change in distance between two probes, the result is scalar and gives no indication of which probe (donor and / or acceptor) moves
- the presence of free unbound labels in solution could mask a change in energy transfer

# 1.4 FLOW INJECTION ANALYSIS (FIA)

# **1.4.1** History of flow injection analysis

The term flow injection analysis was coined in the early spring of 1974 at the Technical University of Denmark by Ruzicka and Hansen [47]. FIA was used to describe a type of continuous flow analysis that utilises a flowing stream, unsegmented by air bubbles, into which highly reproducible volumes of sample are injected. Application of this principle to automatic analysis yields a fast, precise, accurate, and extremely versatile system that is simple to operate.

The concept of FIA received earliest attention from the electrochemists. In 1970 Nagy, Feher, and Pungor published their first in a series of papers describing injection of a sample into a flowing stream of electrolyte [48]. Their system involved passage of the sample carrier stream through a magnetically stirred mixing chamber followed by flow past a silicone-rubber-based graphite electrode.

This technique was subsequently modified simultaneously by Stewart, Beecher, and Hare in the United States [49] and by Ruzicka and Hansen in Denmark [47]. Their primary innovation was in the use of flow-induced sample dispersion as the sample carrier stream was pumped through narrow bore tubing. This allowed controlled mixing of the sample as opposed to the gross mixing generated in the mechanically stirred chamber, thus avoiding excessive sample dilution. The Danish group developed the method using primarily instrumentation normally associated with segmented flow analysers (SFA). In contrast, the American group based their initial work on high-performance liquid chromatography (HPLC) components.

## 1.4.2 Theory of flow injection analysis

Theory, applications, and developments of FIA are covered by many papers [50 - 57] and books [58, 59]. The technique of FIA depends on four primary factors: unsegmented flow, sample injection, reproducible timing, and controllable sample dispersion. A typical FIA system is shown in figure 1.10 and usually consists of the following components: (a) a pump (commonly a peristaltic pump), which is used to propel the carrier stream through a thin tube at a steady pulse-less rate; (b) an injection port, which allows a well-defined volume of sample solution to be injected into a carrier stream in a reproducible manner without stopping the flow; (c) a reaction coil in which the sample zone disperses and reacts with the components of the carrier stream; and (d) a flow-through detector, for example a fluorescence spectrophotometer, which transduces some property of the analyte into a continuous signal which is recorded. A typical detector output has the form of a peak, the height and area of which is related to the concentration of the analyte.



Figure 1.10 – Schematic diagram of a typical simple FIA system

When the sample is first injected, it forms a well-defined sample plug in the stream. As the sample is swept downstream through the narrow bore tubing, the plug disperses into, and thus mixes with, the carrier stream under laminar flow conditions to form a gradient. The magnitude of this dispersion is dependent upon the operating parameters applied to the system, including sample volume, tubing bore size, tubing length, flow rate, the molecular diffusion coefficient of the species concerned, and, possibly, coil diameter. By changing these parameters, the dispersion can be easily manipulated to suit the requirement of a particular analytical procedure so that an optimum response is obtained at minimum time and reagent expense.

For the purpose of FIA, the dispersion (D) is defined as the ratio of concentrations before and after the dispersion process has taken place in the element of fluid that yields the analytical readout. The dispersion or dilution of a sample in an FIA system is given by the following equation:  $D = \frac{C_0}{C_{max}}$  (equation 5), where D is the dispersion coefficient at the peak maximum produced by the ratio between C<sub>0</sub>, the concentration of a pure dye, and C<sub>max</sub>, the concentration of that same injected dye as it passes through the detector.

A number of alternative modes of FIA have been developed, illustrating the versatility of this technique. These include merging zone [60], stopped flow [61], sequential injection [62 - 64], bead injection [65], all injection analysis [66], and multi-component [67 - 72] techniques.

Flow injection analysis has also been used on a miniature scale where microfluidic devices commonly have channels 200  $\mu$ m wide and 100  $\mu$ m deep with typical channel lengths of 2 – 3 cm long. This is a vast difference to those dimensions used in typical FIA where tubing of around 0.8mm internal diameter with lengths of the order of tens of cm are typically used.

# 1.5 HIGH THROUGHPUT SCREENING (HTS)

# 1.5.1 Concept of high throughput screening

Drug discovery is a numbers game. The well-known attrition rate of candidates screened to drugs on the market is dismal (commonly cited to be 5,000 – 10,000 to 1). With increasing pricing pressure on marketed drugs, escalating research costs and competition, and expiring patents on legacy cash cows, the pharmaceutical industry needs to improve its efficiency [73]. One simple, but hopefully effective, strategy is to screen more compounds more efficiently: test more, discover more.

High throughput screening (HTS) is the process by which large numbers of compounds are tested, in an automated fashion, for activity as inhibitors or activators of a particular biological target, such as a cell surface receptor or a metabolic enzyme.

It can be seen from figure 1.16 that since 1995 the number of published articles on high throughput screening has increased from year to year, illustrating the importance that research in this area is gaining.

Nearly all drug discovery research projects in the pharmaceutical industry employ HTS assays as initial steps to discover the chemical leads. The primary role of the technology is to detect lead compounds and supply directions for their optimisation using other techniques. Many properties critical to the development of a successful drug cannot be assessed by HTS, such as specificity, toxicity and pharmacokinetics. The dropout rate during development is still considerable and is associated with high financial losses in the pharmaceutical industry. Only one out of approximately ten potential drug candidates entering phase I of

development will finally reach the market. Insufficient pharmacological and pharmaceutical properties represent the majority of reasons for failure.

## 1.5.2 History of screening

In the 1970s, lead identification typically involved the manual synthesis of about 2 g of a compound with screening being carried out *in vivo* by injecting about 80 mice. The major pharmaceutical companies screened approximately two compounds per week.

In the 1980s, *in vivo* tests were largely replaced by *in vitro* biological assays which were faster and required less compound to be synthesised – screening increased to tens of compounds per week.

HTS has been developed in the last few years to test several thousands of compounds for biological activity per day. Companies are now looking to screen hundreds of thousands of compounds per day per machine.

Although any assays performed on the bench top can, in theory, be applied to HTS, conversion to an automated format imposes certain constraints that affect the design of the assay in practice. Procedures that are routine at the bench are often extremely difficult to automate. Also, the more steps required for an assay, the more difficult to automate the HTS. The ideal assay is one that can be performed in a single well with no other manipulation other than addition of the sample to be tested.

A number of assay formats have been developed or modified over the past few years to conform to the constraints imposed by HTS. Several of the more recently developed or exploited assay protocols for HTS have been the focus of several papers [74, 75].

#### 1.5.3 The drug discovery process

Drug discovery is a complex process, which generally starts with the identification and characterisation of a target like an enzyme or other protein implicated in a disease.

A comprehensive analysis of the drug targets underlying current drug therapy undertaken in 1996 showed that present-day therapy addresses only about 500 molecular targets and as shown in figure 1.11, cell membrane receptors, largely heterotrimeric GTP-binding protein (G protein)-coupled receptors, constitute the largest subgroup with 45 % of all targets, and enzymes account for 28 % of all current drug targets. This analysis was discussed in an excellent review that also covers the evolution of drug discovery and discusses the historical aspects of drug discovery from its infancy to the current day [76].





HTS assays are developed to screen for a desired activity against the target. The introduction of robotic HTS, with its ability to devour up to 100,000 samples per day per machine in mass screening, has given rise to an insatiable demand for new compounds and the flood of targets that will ultimately come from the Human Genome Project will further compound this demand. Continuing advances in molecular biology, human genetics, genomics [77], and proteomics [78, 79] have also accelerated identification of the mechanisms underlying a growing number of human diseases. This progress has increased the number of novel protein targets available for potential therapeutic intervention by drug treatment. Historically, a medicinal chemist is able to synthesise 25 - 100 compounds per year one-by-one, depending upon complexity. Clearly this classical approach is a totally unsatisfactory way to feed the HTS process.

Combinatorial Chemistry (Combi Chem) is an attempt to increase the productivity of medicinal chemists by synthesising large numbers of compounds with the help of robots using standardised chemistry (dozens to millions of compounds in one run, depending upon technology).

The bottleneck in drug development has shifted from the generation of lead 'structures to their transformation into orally active drugs with the desired physiological properties and performance results in clinical trials. It will therefore no longer be enough to synthesise large and diverse libraries; rather, it will become more and more important to find ways to integrate demands such as bioavailability at an early stage of the drug search. These challenges can only be met by creating efficient interfaces between combinatorial syntheses and bioassays. Recent developments in this area have been the focus of a recent article [80].

A selection of compounds from combinatorial chemistry is screened using the HTS assays to identify active compounds. Hits identified in this first round must be retested to confirm their activity. By synthesising a series of compounds that are chemically related to the lead compound, lead optimisation then attempts to improve the activity, availability, stability, and specificity while reducing toxicity and potential side effects. Their improved biological activity is reconfirmed and, those few that seem to be promising are then moved forward into pre-clinical animal studies to evaluate biological activity *in vivo*.

## 1.5.4 Miniaturised high throughput screening

The transition from slow, low throughput screening to industrialised robotic ultra high throughput screening in the last few years has made it possible to screen hundreds of thousands of compounds against a biological target in a short timeframe. The need to minimise the cost of screening has been addressed primarily by reducing the volume of sample to be screened. This, in turn, has resulted in the miniaturisation of HTS technology as a whole.

Miniaturisation of HTS results in three significant benefits that address the pharmaceutical industry's needs: lower costs, faster turnaround, and reduced space requirements. Approximately three-quarters of the costs of screening derive from the costs of both chemical and biological reagents. Miniaturisation reduces these costs in proportion to the reduction in volume (typical flow dimensions were given in section 1.4.2). Miniaturisation also allows for the compression of more samples onto one system, reducing sample-to-sample distance. As sample transport is a significant time factor in screening, the total analysis time per sample is reduced significantly by miniaturising. Due to the picoliter volumes of highly miniaturised systems, and the high-density formats that can be designed, assay throughput can be increased without an associated research and development cost. The concept, scope, and impact of miniaturised methods as applied to HTS have been explored [81, 82]. Trends in the field of ultra high throughput screening and the technological developments that are necessary to enable the routine application of miniaturised systems within an industrial environment have been summarised [83].

In a recent paper, the parallel development, adaptation, and integration of different microelectronic sensors into miniaturised biochips was demonstrated [84]. This was carried out for a multiparametric, functional on-line analysis of living cells in physiological environments, which would be applied to HTS

## 1.6 ENZYMES

## 1.6.1 Introduction to enzymes

Enzymes are the reaction catalysts of biological systems. They have high molecular weights ranging from around 5,000 to over 1 million. They increase the rate of chemical reactions taking place within living cells without themselves suffering any overall change. The reactants of enzyme-catalysed reactions are termed *substrates* and each enzyme is quite specific in character, acting on a particular substrate or substrates to produce a particular product or products via an enzyme-substrate intermediate.

With the exception of a small group of catalytic RNA molecules, all enzymes are proteins. Their catalytic activity depends upon the integrity of their native protein conformation. If an enzyme is denatured or dissociated into subunits, catalytic activity is usually lost. If an enzyme is broken down into its component amino acids, its catalytic activity is always destroyed. Thus the primary, secondary, tertiary, and quaternary structures of enzymes are essential to their catalytic activity is called the tertiary structure that is responsible for the catalytic activity is called the 'active site' of the enzyme, and often makes up only 10 - 20% of the total volume of the enzyme. The active site contains the residues that directly participate in the making and breaking of bonds. These residues are called the catalytic groups.

Water is normally excluded from the active site unless it is a reactant. The nonpolar character of much of the active site enhances the binding of substrate. However, the active site may also contain polar residues, creating a microenvironment in which certain of these residues acquire special properties essential for catalysis. In addition to hydrophobic interactions, electrostatic

bonds, hydrogen bonds, and van der Waals forces mediate reversible interactions between enzyme and substrates. These highly specific non-covalent enzyme-substrate binding interactions confer upon the enzyme high substrate specificity.

Many sources of literature covering the structure, function and basic concepts of enzymes are available for a more detailed discussion [85 - 87].

# 1.6.2 Factors affecting enzyme activity

Several factors affect the rate at which enzymatic reactions proceed – temperature, pH, and the presence of any inhibitors or activators.

Like most chemical reactions, the rate of an enzyme-catalysed reaction increases as the temperature is raised to a maximum level, known as the optimum temperature. Activity then abruptly declines with further increase of temperature, normally due to denaturation of the enzyme.

Enzymes are affected by changes in pH. The most favourable pH – the point where the enzyme is most active – is known as the optimum pH. Extremes of pH generally result in complete loss of activity for most enzymes where activity may not be restored even after re-adjustment of the pH.

Enzyme inhibitors are substances that alter the catalytic action of the enzyme and consequently slow down, or in some cases, stop catalysis. Enzyme inhibition is discussed in more detail in section 1 9.

Activators are substances that increase the catalytic activity of an enzyme, are used in very small amounts and which, unlike coenzymes, do not participate in the enzyme reaction.

### 1.6.3 Enzyme kinetics

Experimental evidence shows that for many single-substrate enzyme-catalysed reactions a hyperbolic relationship exists between initial velocity  $v_0$  and initial substrate concentration [S<sub>0</sub>] as shown in figure 1.12.



## Figure 1.12 - Dependence of initial velocity on substrate concentration for a single-substrate enzyme-catalysed reaction

The graph  $V_0$  versus [S<sub>0</sub>] (figure 1.12) shows three distinct regions:

- At low substrate concentration the velocity of the reaction is directly related to the substrate concentration; this is the region of first order kinetics.
- At intermediate substrate concentration there is a big curve, where the velocity of the reaction increases more slowly with increase in the substrate concentration.

3. At high substrate concentration  $V_{max}$  is attained and the observed velocity is independent of further addition of substrate; this is the region of zero order kinetics.

Kinetic models to explain these findings were proposed by Henri (1903) and Michaelis and Menten (1913). These were essentially similar, but Michaelis and Menton did a great deal of experimental work to give their work a sounder basis. The critical feature in their treatment is that a specific enzyme – substrate (ES) complex is a necessary intermediate in catalysis. The model proposed, which is the simplest one that accounts for the kinetic properties of many enzymes, is:



where E = enzyme, S = substrate, ES = enzyme – substrate complex, and P = product. E combines with S to form ES with a rate constant of  $k_1$ . ES then has two fates. It can dissociate to E and S with a rate constant of  $k_2$  or proceed to form P with a rate constant of  $k_2$ .

 $V_0 = \frac{V_{max}[S]}{K_M + [S]}$  (equation 6) is the *Michaelis-Menten equation*, the rate equation

for a single-substrate, enzyme-catalysed reaction. It is a statement of the quantitative relationship between the initial velocity  $V_0$ , the maximum initial velocity  $V_{max}$ , and the initial substrate concentration [S], all related through the Michaelis-Menten constant  $K_M$ .

The meaning of  $K_M$  is evident from equation 6. When  $[S] = K_M$ , then  $V = V_{max} / 2$ . Thus  $K_M$  is equal to a substrate concentration at which the reaction rate is half its maximal value. The Michaelis-Menten equation gives a hyperbolic curve as shown in figure 1.12. This is not entirely satisfactory for the determination of  $V_{max}$  and  $K_{M}$ . Unless, after a series of experiments, there are at least three consistent points on the plateau of the curve at different [S] values, then an accurate value of  $V_{max}$ , and hence of  $K_{M}$  cannot be obtained.

This problem was overcome by Lineweaver and Burk (1934) without making any fresh assumptions. They simply inverted the Michaelis-Menten equation thereby conveniently transforming it into a straight-line plot:  $\frac{1}{V} = \frac{K_M}{V_{max}} \cdot \frac{1}{[S]} + \frac{1}{V_{max}}$  (equation 7). This is of the form y = mx + c, which is the equation of a straight line graph; a plot of y against x has a slope of m and intercept of c on the y-axis. As shown in figure 1.13,  $K_M$  and  $V_{max}$  can be read directly from the Lineweaver-Burk plot.



#### Figure 1.13 - Lineweaver-Burk plot

Numerous sources of literature are available that discuss enzyme kinetics in greater detail [88 - 90].

# 1.6.4 Proteolytic enzymes

Proteases are enzymes that catalyse the hydrolysis of peptide bonds. They are widely distributed in cells, and are either soluble, or associated with plasma membranes and subcellular organelles, or secreted. This extensive distribution of proteases in both organs and subcellular compartments suggests that they play a key biological role in both physiological and pathophysiological situations. In 1960 the biochemist Hartley [91] proposed that proteases should be classified according to their catalytic mechanism: serine proteases, cysteine proteases, aspartate proteases, and metallo proteases.

The serine proteases are characterised by the presence of a catalytic triad composed of aspartic acid (Asp), serine (ser) and histidine (His). Serine acts, in this case, as a very reactive residue, which forms a covalent bond with the substrate. Hydrolysis of the substrate occurs in two successive stages: acylation and deacylation of the enzyme. The carbonyl group of the peptide bond to be cleaved undergoes nucleophilic attack by the unusually reactive OH group of the active site serine, resulting in the formation of a covalent tetrahedral intermediate. Decomposition of the tetrahedral adduct results in the release of the amino portion of the substrate and the formation of an acyl serine derivative The departing amino group receives a proton from the imidazole ring of histidine. Subsequent hydrolysis of the acyl enzyme to active enzyme and the carboxylic acid product occurs through a reaction sequence analogous to that involved in the formation of the acyl enzyme. This catalytic process is facilitated by the proximity at the enzyme's active site of the histidine and aspartate residues, which thus actively participate in the catalysis.

The cysteine proteases are characterised by the catalytic triad cysteine (Cys), histidine (His), and aspartic acid (Asp). Cysteine is a very reactive residue, which is involved in the formation of the intermediate complex with substrates.

The aspartate proteases are also called acid proteases due to their low pH optima. Their catalytic site consists of two aspartic acid residues. The enzymesubstrate intermediate is of the general acid-base type rather than the covalent type. The two aspartic acid molecules are linked through a hydrogen bond. Aspartate proteases facilitate nucleophilic attack by two simultaneous proton transfers: the first from a water molecule to the carboxylate ion of one aspartic acid to the oxygen atom of the carbonyl group of the substrate. These transfers lead to the formation of a neutral tetrahedral intermediate. The same dual transfer mechanism destroys this intermediate, which donates a proton to the carboxylate ion of the second aspartic acid, during which the proton of the carboxyl group of the second aspartic acid is transferred to the nitrogen atom, thus hydrolysing the C-N bond of the substrate.

The metallo proteases have an active site that contains a metal ion, usually zinc  $(Zn^{2+})$ , which is an integral part of the protein structure. They do not form covalent enzyme-substrate intermediates. In addition to a metal ion, the catalytic site always contains a glutamic acid (Glu) residue, sometimes associated with tryptophan (Trp) or histidine (His).

Serine and cysteine proteases therefore form a covalent intermediate complex with their substrates because the amino acids at the catalytic site are very strong nucleophiles. Aspartate and metallo proteases, on the other hand, usually act by an acid-base catalytic mechanism, using a water molecule.

A variety of sources can be found in the general area of proteolytic enzymes [92 - 95].

The proteases play a key role in the digestion of food proteins, zymogen activation, activation of the complement system, and in blood coagulation and

fibrinolysis. They are also involved in the regulation of blood pressure, tissue regeneration, and in reproduction both during ovulation and the fusion of gametes. Proteases are also involved in immune response, processing of peptides and proteins, and growth and development. When the balance of proteases in these biological functions becomes upset, pathological conditions occur. As discussed in section 1.8.1, as the proteases are involved in virtually all biological functions and dysfunctions they represent an important drug target.

Five different proteolytic enzymes have been investigated in this work and thus a brief description of each is given below. An extremely comprehensive book covers databanks, name and history, activity and specificity, structural chemistry, preparation, biological aspects, and distinguishing features of proteolytic enzymes known to date [96].

#### **1.6.4.1** Alkaline protease

Alkaline proteases are covered in detail elsewhere [97]. Enzyme commission number: unknown, type: serine, MW: 50,000, activity: ~ 26 Units / mg solid, specificity: broad, pH: maximal activity at pH 11 0, additional information: type XXI protease.

#### 1.6.4.2 Proteinase K

Specific proteinase K information is available [98, 99]. Enzyme commission number: 3 4.21 64, type: serine, MW: 28,930, activity: 10 - 20 Units / mg protein, specificity. peptide bond adjacent to the carboxyl group of aliphatic and aromatic amino acids with blocked alpha-amino groups, pH: active over the range 7.5 to 12.0, but most often used in pH range 7.5 – 9.0, additional information: remains active in the presence of sodium dodecyl sulfate and urea.

## 1.6.4.3 Chymotrypsin

Enzyme commission number: 3.4.21.1, type: serine, MW: 25,000, activity: 40 – 60 Units / mg protein, specificity: hydrolyses peptide bonds with aromatic or large hydrophobic side chains (Tyr, Trp, Phe, Met) on the carboxyl end of the bond, pH: maximal activity around pH 80, additional information:  $\alpha$ - and  $\gamma$ - chymotrypsins have identical primary structures, but different enzymatic properties, stabilities and tertiary structures - during experimental work  $\alpha$ - chymotrypsin has been used.

## 1.6.4.4 Trypsin

Enzyme commission number: 3.4.21.4, type: serine, MW. 23,800, activity: 1,000 – 2,000 Units / mg protein, specificity: peptide bonds with arginine or lysine on the carboxyl end of the bond with a preference for arginine over lysine that is 2-to 10-fold, pH: maximal activity around pH 8.0, additional information: first described and named in 1876 by Kuhne as the proteolytic activity in pancreatic secretions.

## 1.6.4.5 Pepsin

Enzyme commission number: 3.4.23.1, type: aspartic acid, MW: 35,000, activity: 3200 - 4500 Units / mg solid, specificity: preferentially cleaves C-terminal to phenylalanine, leucine, and glutamate, does not cleave at valine, alanine, or glycine, pH<sup>-</sup> pH optimum from 2 - 4 (irreversibly inactivated at pH > 6), additional information: active in 4 M urea and 3 M guanidine HCI.

#### 1.7 ENZYME IMMOBILISATION

The term 'immobilised enzyme' describes an enzyme that has been chemically or physically attached to a water-insoluble matrix, polymerised into a water-insoluble gel, or entrapped within a water-insoluble gel matrix or water-insoluble microcapsule resulting in reduced or complete loss of enzyme mobility. These immobilised enzymes retain their catalytic activity and can be used repeatedly and continuously. The usage of binding enzymes onto solid materials goes back to 1916 when Nelson and Griffin [100] first immobilised invertase on charcoal and observed that the enzyme retained most of its activity over a long period of time. However, this work was largely ignored until the 1950s. Since then immobilisation of proteins in general, and enzymes in particular, has been widely studied and the number of applications of immobilised enzymes is increasing Many sources of information can be found on the preparation, characteristics, and application of immobilised enzymes for more detailed reading [101 - 105].

#### 1.7.1 Why immobilise enzymes?

There are several reasons for the preparation and use of immobilised enzymes. In addition to a more convenient handling of enzyme preparations, the two main benefits are (1) easy separation of the enzyme from the product, and (2) reuse of the enzyme.

Easy separation of the enzyme from the product simplifies enzyme applications and supports a reliable and efficient reaction technology. Reuse of enzymes provides cost advantages, which are often an essential prerequisite for establishing an enzyme-catalysed process in the first place.
Immobilised enzymes offer many more advantages over their soluble analogues than the two main benefits mentioned above [106]. These include:

- increased stability against denaturing conditions such as high temperature and extreme pH during storage and use
- ease of use as no enzyme solution preparation is required
- greater applicability to multi-determination due to the ease of combining immobilised enzymes in a variety of designs in flow systems
- they do not contaminate the sample and insoluble enzymes can be easily recovered from batch experiments, while in the case of continuous flow systems the sample is passed through enzyme reactors

# 1.7.2 Immobilisation methods

Immobilised enzymes can be classified on the basis of the methods adopted for their immobilisation, and the classification of immobilisation methods according to different chemical and physical properties is shown in figure 1.14. Chemical methods of immobilisation include any that involve the formation of at least one covalent bond between residues of an enzyme and a functionalised waterinsoluble support or between two or more enzyme molecules. Physical methods include any that involve localising an enzyme in any manner, which is not dependent upon covalent bond formation.

The immobilisation procedure should be conducted in a way that allows the enzyme to maintain its active conformation and necessary catalytic flexibility. Furthermore, catalytically essential residues of the enzyme should be undisturbed and conserved.



# Figure 1.14 – Classification of immobilisation methods

# 1.7.2.1 Covalent binding method

The method of enzyme immobilisation that has been adopted for use in the practical work described in this thesis has been the use of the bi-functional reagent glutaraldehyde to covalently immobilise proteases onto the support controlled-pore glass (CPG). This method has been described elsewhere [107] and the reaction scheme of the coupling of enzyme to CPG is shown in figure 6 3.

Glutaraldeyde is one of the most extensively used linking reagents and has been discussed in detail elsewhere [108]. For this study, aminopropyl-CPG has been used. Glutaraldehyde has been used to covalently link proteolytic enzymes to "the CPG as it has a high reactivity with an amino group. The reaction of an amino-terminal support with glutaraldehyde leads to the introduction of aldehyde groups onto the support that can bind enzymes. This reaction has been termed activation.

Controlled pore glass has gained much acceptance as a support for immobilisation and has been widely used over the last couple of decades [109 - 114].

CPG has many advantages as a support:

- high surface area giving high coupling of ligands and high yield
- high mechanical strength
- thermostable and autoclavable
- high flow rate giving higher throughput
- rigid glass structure which is immune to biological degradation and inert to solvents or changing conditions

# 1.7.3 Modification of enzyme properties after immobilisation

Immobilisation of an enzyme results in a change in many of the physical and kinetic properties of the corresponding soluble counterpart. One of the main concerns is the reduction in the biological activity due to immobilisation. However, immobilisation can also induce novel characteristics to make enzymes tolerant to even harsh environments. Some attempts to intentionally modify the catalytic behaviour have been reviewed [115]. These changes in enzyme

properties that occur upon immobilisation may be due to changes in the protein conformation after immobilisation, structural modification of the protein during immobilisation, or changes in the protein microenvironment resulting from the interaction between the support and the protein. Goldstein *et al.* made one of the earliest demonstrations that immobilisation can markedly affect the properties of enzymes [116] It was shown that the pH-activity curves of immobilised trypsin, when compared with those of soluble trypsin, were displaced 2 - 3 pH units towards the alkaline region. This effect was attributed to a pH difference between the microenvironment of the polymer matrix and the bulk fluid.

Numerous enzymes have been reported with an increase in long-term storage and thermal stability upon immobilisation [117, 118]. This increase in stability may be due to the stabilisation of the native structure of the enzyme.

#### 1.7.4 Immobilised enzymes in FIA

A variety of flow injection systems using immobilised enzymes have been described [119 - 129].

The application of immobilised enzymes to flow injection analysis systems allows the combination of immobilised enzymes in different arrangements to permit multi-determination. An excellent review discusses the determination of two or more species in the same sample with a single flow injection analysis system [70] and the simultaneous determination of several components by flow injection analysis using one or more detectors has also been discussed [71]. Analysis of more than one analyte is made possible by using immobilised enzyme reactors in parallel [130 - 132] or in series [133]. Striking examples of the advances that have been made in this area are the determination of a single analyte by using a single immobilised column and progressing towards the determination of five or six analytes [69, 134 - 135]

# 1.8 PROTEASE ASSAYS

#### 1.8.1 Importance of protease assays

Since the discovery of pepsin in the late eighteenth century there has been continuing investigation into the chemistries and activities of the proteolytic enzymes. Recent years have seen a remarkable acceleration of the pace of this research, fuelled by numerous applications in biotechnology, and the realisation that the proteases are major therapeutic targets. As a result of worldwide media attention, the term "protease inhibitor" has been closely associated with HIV - so much so, that the publicity has eclipsed the fact that proteases are involved in virtually all biological functions and dysfunctions Protease inhibitors for HIV work by blocking a key enzyme involved in viral replication, the HIV protease. First introduced in late 1995, protease inhibitors are widely used for the treatment of HIV infection, in combination with other antiretroviral drugs. Approximately 215,000 of the estimated 350,000 patients receiving treatment for HIV infection in the United States take at least one protease inhibitor. The worldwide market for protease inhibitors was nearly \$2 billion in 1999. However, the proteases are involved in many other conditions such as lung disease [136 - 138], osteoporosis [139], microbial infections [140, 141], viral infections [142 - 144], malaria [145], and the common cold [146]. As such, protease inhibitors have broad potential as a class of drugs [147].

There are around 400 known human proteases and of these about 14% are currently under investigation as drug targets [148]. In addition to the involvement of these enzymes in so many disease states their popularity as drug targets also stems from the accumulated data on their enzymology, high throughput assays, biochemistry, physiology, pathology, 3D structures, small-molecule inhibitors, and endogenous inhibitory proteins. The Human Genome Project, by revealing all

potential proteins, will also have a major effect on both academic and pharmaceutical protease research.

Proteases as drug targets can be broken down into groups – the serine proteases, for example thrombin as a target for cardiovascular disease [149 - 152], cysteine proteases, e.g. the cathepsins as targets for inflammatory disease, the aspartic proteases, for example renin which has had tremendous effort expended on it for the treatment of cardiovascular disease [153], and also HIV inhibitors for the treatment of HIV [154], and the metalloproteases, for example angiotensin converting enzyme inhibitors for the treatment of cardiovascular disease and more recently the matrix metallo protease inhibitors which are being looked at for inflammation and cancer [155 - 157].

One of the most successful groups of protease inhibitors from the pharmaceutical industry point of view over the last 20 years has been the discovery of the angiotensin converting enzyme (ACE) inhibitors. These have seen tremendous sales in the treatment of high blood pressure (hypertension) and heart failure. As shown in figure 1.15, the proteolytic angiotensin converting enzyme converts angiotensin I into an active form called angiotensin II. Angiotensin II is a vasopressor and so stimulates the contraction of the smooth muscle of blood vessels causing a rise in blood pressure. If the formation of the vasopressor angiotensin II could be prevented there would be a drop in blood pressure. A protease inhibitor that would stop angiotensin converting enzyme from working, hence preventing the formation of angiotensin II, would therefore be a potential drug for high blood pressure.





GENERIC NAME OF DRUG	TRADE NAME OF DRUG	SALES FOR FINANCIAL YEAR ENDING 2000
Lisinopril	Zestril + others	\$ 1, 800, 000, 000
Enalapril	Renitec + others	\$ 1, 800, 000, 000
Captopril	Capoten + others	\$ 400, 000, 000

#### Table 1.1 – Worldwide sales figures for selected angiotensin converting enzyme inhibitors as drugs

Table 1.1 shows just three compounds that are angiotensin converting enzyme inhibitors. Their "generic" names are given and one example each of a Trade Name (drugs are marketed under different Trade Names usually depending on the country of sale or whether the compound is licensed to additional companies for marketing). The worldwide sales for these three inhibitors alone were more than 4 billion US dollars. This is a very significant amount in drug sales terms and illustrates just how profitable protease inhibitors can be to pharmaceutical companies.

During screening for promising drug candidates, hundreds of thousands of compounds need to be assayed. Figure 1.16 shows the number of 'hits' i.e. the number of published articles, obtained on Web of Science when the terms protease, protease inhibitor, and high throughput screening are put as keywords. A trend can be seen over the last ten years of an increase in published papers in these three fields illustrating the importance of the application of high throughput screening assays for protease inhibitors. The development of a simple, sensitive, and efficient method for assaying protease activity, which would facilitate the testing of a large number of samples using HTS, was therefore investigated in this work.



Detection of protease activity is also important for biochemical applications such as quantifying protease activity in cell or tissue extracts, during enzyme purification, and for quality control testing. Contaminating protease activity directly impacts product or reagent stability and biological efficacy in the synthesis, production, and purification of naturally occurring or genetically engineered recombinant proteins. Some protease assays may require detection of specific protease activity; for example certain pathogens use proteolytic activity as a virulence mechanism and differ from their non-pathogenic counterparts in this characteristic. Conversely, assay for contamination of biological preparations by unknown proteases requires a substrate that can detect a variety of proteases.

# Figure 1.16 – Trend in number of published articles in the areas of proteases, protease inhibitors and high throughput screening over the last decade

#### 1.8.2 General protease assay methods

The concept of assaying proteolytic activity is not a new one and this type of assay has been around for many years. Gel electrophoresis [158 - 160], radioactivity [161], capillary electrophoresis [162], a dot-blotting method [163], immuno-quantification [164], and detection through an indicator enzyme [165] have all been used in the past to detect proteolytic enzymes. However, such methods are either time consuming or utilise relatively large quantities of substrates.

Several protease assays have been described that use HPLC to analyse the products of peptidase hydrolysis of peptide or protein substrates [166 - 169]. There are, however, several drawbacks to HPLC procedures, aside from the high cost of the equipment. Only one sample can be analysed at a time, turn-around times for injections may be long and will vary with the type of column and solvents used, contamination of the column and loss of resolution can occur, and analytical columns need to be replaced at regular intervals.

Luminogenic substrates have also been applied to the assay of proteolytic enzymes [170 - 174], as well as chromogenic substrates [175 – 178]. However, due to fewer luminescent species than fluorescent ones and the increase in sensitivity of fluorescence over absorbance-based methods, fluorescence spectroscopy seems more versatile as a method.

Different methods for assaying proteases have been reviewed elsewhere [179], and so discussions here will concentrate solely on fluorimetric protease assays as these are relevant to this particular work. Comprehensive reading on fluorimetric assays of proteolytic enzymes can also be found [180 - 183].

#### 1.8.3 Fluorescent protease assay methods

Protease assays using a variety of fluorescently labelled substrates have been in use for many years [184 - 187]. These assays have frequently employed proteins such as haemoglobin [187], and fluorescent dye-labelled proteins such as fluorescein thiocarbamoyl casein (FTC-casein) [188 - 190]. These methods include precipitation assays where the undigested substrate is insoluble, such as fibrin [186], or is made insoluble by treatment with trichloroacetic acid or hydrochloric acid [187, 191]. However, such assays require sampling at time intervals, careful control of sample volume, and quantitative separation of the labelled peptides from the unhydrolysed protein.

#### 1.8.3.1 Fluorescence polarisation protease assays

Fluorescence polarisation was first described in 1926 by Perrin [33]. This was expanded to cover biological applications by Weber [192 - 194]. The first instrumentation for the measurement of fluorescence polarisation was also developed by Weber [195]. Several groups have used this technique for the study of antigen-antibody and hormone-receptor interactions [196 – 199], as well as its use commercially for the monitoring of therapeutic drug levels [200]. Recently this technique has been introduced for high throughput screening in the drug discovery process [201 - 205].

However, despite the fact that there is a solid theoretical foundation with regards to fluorescence polarisation, it has proven to be a very under-utilised tool and only recently has it become recognised as a valuable tool in the area of high throughput screening. The main reason for this is lack of high-performance and low-cost instrumentation.

#### Chapter 1

The principle of how fluorescence polarisation is used to detect protease activity using a fluorescent substrate is shown in figure 1.17 a and b.

A relatively large fluorescent molecule such as a fluorophore-labelled protein has a significant fluorescence polarisation value. If a protease is introduced, the labelled fragments produced have a lower fluorescence polarisation value which drops with time, the decrease in fluorescence polarisation being a measure of proteolytic activity.

#### Figure 1.17a – Principle of the use of fluorescence polarisation for the detection of a large lightly labelled fluorescent protein





#### Figure 1.17b – Principle of the use of fluorescence polarisation for the detection of protease digested fluorescence peptides

The application of fluorescence polarisation to protease assays has long been established [189,190, 203, 207, 208].

An early paper by Maeda [189] described a detailed study of proteolytic enzyme assay development in which various fluorescein isothiocyanate conjugated proteins were used as substrates. These reactions were carried out in a static

#### Chapter 1

Introduction

cuvette and enzyme reactions that required a pH that was different to that of the pH of the measurement (pH 7 to 9) were adjusted back prior to detection. The results showed a near linear decrease in polarisation with time due to the apparent decrease in molecular weight of the substrates with hydrolysis. This method was shown to be simple and sensitive with trypsin activity being detected as low as 0.05  $\mu$ g / ml. This was comparable at the time to the most sensitive methods using synthetic chromogenic substrates where trypsin was used at 0.01 to 0.1  $\mu$ g / ml [206] or to those utilising radiolabeled substrates, which allowed the determination of trypsin at 0.065  $\mu$ g / ml [161]. Maeda also demonstrated the proteolytic inhibition of papain or trypsin using various protease inhibitors (leupeptin, soybean trypsin inhibitor or antipain).

More recently Bolger and Checovich [190] compared the sensitivity of a polarisation-based fluorescence protease assay that used fluorescein thiocarbamoyl (FTC)-casein as a substrate with a non-fluorescence polarisation acid precipitation technique, as described by Twining [188]. Twining had previously estimated that the FTC-casein assay, which requires separation of the intact casein from the cleaved peptides by differential precipitation with trichloroacetic acid (TCA) and subsequent quantification of cleaved peptide by fluorimetry, was several orders of magnitude more sensitive than the measurement of cleaved peptides at A280. In general it was found that the fluorescence polarisation assays resulted in an increase in sensitivity of 2- to 20fold compared with the non-fluorescence polarisation assay of Twining, depending on the protease tested. In addition, the fluorescence polarisation assay also eliminates the centrifugation and two transfer steps required for the TCA assay and therefore is much simpler in design. These fluorescence polarisation assays took 1 hour to be carried out.

Due to the pH sensitivity of FITC, these methods cannot be utilised at low pH values without the necessity of adjusting the solution pH prior to detection. In

more recent years a new pH-independent fluorescent protein substrate has been developed based upon BODIPY-FL:α-casein [202, 207].

Since many proteases of interest in drug discovery do have optimum pH values below pH 6, where fluorescein is essentially non fluorescent, and because of a general desire to move into the red portion of the spectrum to minimise potential interferences from fluorescent library members, the potential of BODIPY fluorophores was explored as labels in fluorescence polarization protease assays [202]. The BODIPY fluorophores used were BODIPY FL (ex = 502 nm, em = 510 nm) and BODIPY TRX (ex = 588, em = 616). The assays were highly sensitive and could detect less than 1 ng / ml pepsin and less than 10 ng / ml trypsin, with a 5 minute incubation time. Soybean trypsin inhibitor (STI) could be readily detected at less than 50 ng / ml with a 5 minute incubation time.

BODIPY FL (ex = 504 nm, em = 511 nm) conjugated to  $\alpha$ -casein was also shown to be successful as a substrate for proteases over the pH range 2 to 11 [207]. The molar ratio of BODIPY to protein was 0.14 and it was suggested that a molar ratio of 0.1 to 0.2 is optimum for fluorescence polarisation work. Addition of trypsin at a final concentration of 0.5 µg / ml gave a decrease in fluorescence polarisation of about 150 mP in 3 minutes at 37°C. This large decrease in the fluorescence polarisation of BODIPY: $\alpha$ -casein showed that this conjugate was excellent for use as a substrate for protease assays. *S. griseus* alkaline protease activity on the BODIPY: $\alpha$ -casein substrate produced a decrease in polarisation of 137 mP units in the first minute and proteinase K produced a 128 mP units decrease in the first minute.

Molecular Probes Inc. currently market a polarisation kit for proteases using BODIPY FL and BODIPY TR-X conjugates of casein [208].

A fluorescence polarisation assay was designed to measure proteolytic cleavage of a specific peptide substrate for human cytomegalovirus protease in a 96 well

Chapter 1

format [209]. The peptide substrate was derivatised by biotinylation of the aminoterminus and labelled with a fluorophore at the carboxy-terminal end. Incubation of the substrate with the protease and addition of avidin produced a polarisation signal that was proportional to the relative amounts of cleaved and uncleaved substrate. Due to the specificity of the substrate for the protease of interest, this protease assay is more useful than those previously described assays and can be further utilised by designing specific substrates. The effects of absorptive interferants were evaluated by including increasing concentrations of dye interferants absorbing in the wavelength region of either the fluorophore absorption or emission. The assay was found to be insensitive to a wide range of absorptive interferants that may be present in complex mixtures such as blood, when screening for protease inhibitors. An absorptive dye interferant resulted in greater than 90% loss in total fluorescence, while the fluorescence polarisation varied less than 10%. The results showed that while absorptive interferants dramatically affected total fluorescence, they had little effect on fluorescence polarisation. The ability of this assay to study the inhibitory activity of 3, 4 dichloroisocoumarin (a cytomegalovirus protease inhibitor) suggested the versatility of this technique for the study of protease inhibitors.

BODIPY:α-casein has also been used to measure the proteases in subgingival plaque [210], implicated in the pathogenesis of adult periodontitis. This assay was rapid, sensitive and quantitative.

#### 1.8.3.2 Fluorogenic protease assays

The use of fluorogenic substrates for proteases is widespread [211 - 214]. These substrates are non- or weakly fluorescent but are converted by an appropriate enzyme to a highly fluorescent product. Such methods combine the amplification effects of the catalytic enzyme action with the high sensitivity of fluorescence spectrometry alongside low background fluorescence.

#### 1.8.3.2.1 Fluorogenic FRET protease assays

Specific proteases have commonly been assayed using fluorogenic substrates involving Fluorescence Resonance Energy Transfer (FRET). Substrates in which a peptide sequence is constructed to match the specificity of a particular enzyme are labelled at one end by a donor fluorophore and at the other by an acceptor fluorophore. FRET from donor to acceptor quenches the donor emission, an effect reversed upon proteolysis due to separation of donor and acceptor moieties. The increase in fluorescence intensity is proportional to enzyme activity. This method of detecting proteases is represented schematically in figure 1.18.

Figure 1.18 – Schematic representation of (a) a peptide sequence labelled at each end by a donor fluorophore (red) and an acceptor fluorophore (green) and (b) cleavage of this peptide by a protease causing separation of donor and acceptor fluorophores



FRET is one of the most sensitive methods for the detection of proteolytic activity as first shown by Latt et al. [215] for carboxypeptidase A. This method has been applied to the determination of a variety of specific proteases including Alzheimer's disease-associated proteases [216, 217], hepatitis C virus [218, 219], human and feline immunodeficiency virus proteases [40, 217, 220, 221], Escherichia coli leader peptidase [222], renin [217, 223, 224], tail-specific protease (which represents a new class of proteases that have no sequence homology to any well-characterised proteases) [225], and angiotensin-converting enzyme [226] amongst others [227 - 246]. One paper describes a general method for the preparation of these FRET protease substrates using solid-phase synthesis. The paper uses the donor and acceptor pair of 5-[(2'-aminoethyl)amino]naphthelenesulfonic acid (EDANS) and 4-[[4-(dimethylamino)phenyl]azo]benzoic acid (DABCYL) and synthesises substrates for rennin and HIV proteinase as examples [247]. These assays are sensitive, easy to perform, and ideal for the detection of specific proteases with known sequences of proteolytic cleavage. However, these assays are not useful as general proteolysis detection tools and require synthetic expertise. Several papers describe the synthesis of some of these FRET substrates [248, 249]. Another potential problem with this type of substrate is that peptides in aqueous solutions can take conformations such that the donor and acceptor moieties are in close proximity, causing quenching of the emissions of the fluorophores. In general, if the fluorophores have hydrophobic characteristics then they will form dye-to-dye close contact in an aqueous environment.

A FRET protease assay designed for interstitial collegenase and gelatinase [250] has been used as a test system in an investigation into the feasibility of assay miniaturisation [251]. The investigation described precise and rapid measurement of the progress of an enzymatic reaction and its inhibition, which was carried out by mixing at the 200  $\mu$ l level and detecting in 2  $\mu$ l volumes with less than 6 minutes of total assay time. For comparison to a standard format, the

same assay was performed in a 96-well microtiter plate in 200  $\mu$ l using 30 minutes of incubation. Median inhibitory concentrations (IC<sub>50</sub>) for actinonin, an analog inhibitor of collegenase, of 73 ± 16 and 100 ± 14 nM were obtained in the 2- and 200- $\mu$ l assays, respectively.

The use of these FRET protease substrates was further expanded by using homogeneous time-resolved fluorescence [252]. This assay relied upon the cleavage of a peptide substrate, which was labelled at the N terminus with biotin and had a phosphotyrosine residue downstream of the cleavage site. In the presence of streptavidin conjugated XL665, a cross-linked conjugate of allophycocyanin, and a europium-labelled antiphosphotyrosine antibody FRET occurs and a signal was generated. Upon cleavage, FRET is abolished and the signal decreased. This assay was developed for human immunodeficiency viral (HIV) protease and combined the sensitivity of lanthanide-based techniques with a time-resolved measurement.

These substrates have also been used in a solid-phase assay for the complete subsite mapping of the active site of endoproteases [253]. Using a variation of this approach, a solid-phase combinatorial assay has been demonstrates which allows the rapid discovery of potent fluorogenic FRET substrates for use in the assay of novel proteolytic enzymes [254]. This technology has the potential to greatly facilitate the drug discovery process by speeding up the development of high throughput screening assays. This exploits more effectively the ever-increasing information gained from genomics and combinatorial chemistry.

#### 1.8.3.2.2 Quenched over-labelled fluorogenic protease assays

Assays based on fluorogenic protein substrates, which are over-labelled by a suitable fluorophore to such an extent that the dye is efficiently quenched by internal energy transfer in the protein conjugate, have also been employed [255 -

259]. As shown in figure 1.19, upon protease digestion this internal energy transfer is abolished and the virtually non-fluorescent substrates release highly fluorescent dye-labelled peptides, the fluorescence intensity increases being proportional to enzyme activity. This type of protease assay removes the necessity for the separation of substrate from degradation products.

#### Figure 1.19 – Schematic representing the principle of an intramolecularly quenched protein substrate (a) and cleavage of this substrate by a suitable protease to release fluorescent peptides (b)



Fluorescein-labelled bovine serum albumin (BSA) or casein have commonly been used in this type of assay [256, 257]. *Porphyromonas gingivalis* was used in a study using FITC-labelled proteins as it has been implicated as one of the causative organisms of periodontal disease in humans [256].

FITC labelled bovine serum albumin (BSA) has been shown to serve as a general protease fluorescent enhancement substrate [257]. However digestion with trypsin yielded a lower fluorescence increase than that seen with pronase,  $\alpha$ -chymotrypsin and proteinase K. This reduced fluorescence enhancement was consistent with the specificity of FITC conjugation for the  $\epsilon$ -amino groups of lysine

residues, which along with arginine residues are required for trypsin activity. This effect is serious, as it is hard to achieve an adequate degree of fluorescence quenching in fluorescein conjugates unless the degree of labelling is very high. These heavily labelled conjugates may no longer be suitable as substrates for the assay of enzymes that depend upon the availability of the lysine residues to which the dyes are typically conjugated. The substitution of a large number of positive lysine residues with the negative charges of fluorescein may also cause conformational changes such that the interaction of the protein with the active site of the enzyme becomes problematic. Moreover fluorescein is only weakly fluorescent below pH 6, so fluorescein conjugates are not suitable for studies of acid proteases.

For these reasons the BODIPY® fluorophores have been investigated [255]. The BODIPY dyes, discussed in section 1.1.4.3, have no charge and it was found that conjugation of equivalent amounts of fluorescein and BODIPY dyes to both casein and BSA gave substrates that were 89 - 93% and 98 - 99% quenched, respectively. Conjugation of 4.5 to 6 BODIPY FL or even fewer BODIPY TR-X dyes to each casein molecule was optimal As less BODIPY dye is needed for quenching this should minimise the effects of dye conjugation on the protein substrate conformation and enzyme-active site interactions. Α dramatic increase in detection sensitivity of various enzymes with the BODIPYcasein substrates was seen (2- to 30-fold depending on which enzyme), compared to the FTC-casein fluorescence polarisation assay as reported by Bolger & Checovich, 1994 [190]. Hydrolysis of the BODIPY FL casein substrate by chymotrypsin, protease XIV, and proteinase K, generated over a twofold increase in fluorescence intensity when compared with hydrolysis of BODIPY FL BSA, indicating that BODIPY FL casein may have advantages for use as a general protease substrate relative to BODIPY FL BSA. The BODIPY dye labelled substrates are insensitive to pH, making it possible to measure the activity of various enzymes such as pepsin and cathepsin D, which are most active at low pH. The inhibition of trypsin proteolytic activity by pre-incubation

with aprotinin was demonstrated. Trypsin was pre-incubated with aprotinin for 20 minutes and then incubated with a casein conjugate of the fluorescein analogue BODIPY-FL for 1 hour at room temperature, followed by fluorescent measurement. Full inhibition was observed at 4  $\mu$ g ml<sup>-1</sup> aprotinin and 44 % inhibition was observed with 2  $\mu$ g ml<sup>-1</sup> aprotinin.

The BODIPY FL dye has also been used in these over-labelled substrates for the \* 'determination of the protease calpain in a microtiter plate format [258] and as a probe for monitoring of protein degradation in intact cells [259].

Molecular Probes market two protease assay kits using intramolecularly quenched substrates of BODIPY FL:casein and BODIPY TR-X:casein which emit at 513 nm and 617 nm, respectively. These assays are carried out in 1 hour [260].

A new assay for O-Sialoglycoprotein endopeptidase was demonstrated which was based on the increased fluorescence which resulted from proteolytic cleavage of a fluorescence-quenched micellar substrate, BODIPY-FL:glycophorin A [261]. Micellar association of glycophorin A molecules resulted in 97% fluorescence quenching despite a low molar ratio of BODIPY-FL:glycophorin A. This assay was different to those that use hyperconjugated dye protein substrates in that this substrate has only five lysines available for conjugation. However, it readily aggregates, and the resultant intermolecular interactions between fluorophores contribute to fluorescence quenching within the micelle. Glycoprotease was incubated with substrate for 10 - 15 minutes and as little as 1 ng of glycoprotease could be detected.

# 1.9 ENZYME INHIBITION

# 1.9.1 Types of inhibition

Inhibitors are substances that decrease the rate of an enzyme-catalysed reaction. There are two broad classes of enzyme inhibitors: reversible and irreversible. These types of inhibition will be described briefly, but more detailed accounts can be found for further reading [262, 263]. Information covering the use of enzyme inhibitors as drugs can be found elsewhere [264, 265].

# 1.9.1.1 Reversible inhibition

In this type of inhibition the inhibitor binds to the enzyme in a reversible fashion and can be removed by dialysis (or simply dilution) to restore full enzymatic activity. Three distinct mechanisms of reversible inhibition are known, which are recognised by the effect of substrate concentration on the inhibition.

# (I) Competitive inhibition

Competitive inhibitors often closely resemble the substrates whose reactions they inhibit, and because of this structural similarity they compete with the substrate (S) for the active site of the enzyme (E). Inhibitor (I) binds to the enzyme to form an enzyme-inhibitor (EI) complex. While the inhibitor occupies the active site it prevents binding from the substrate.



It can be seen from equation 8 that EI and ES complexes are formed and no ESI complex appears in this type of inhibition. As the inhibitor binds reversibly to the enzyme, the competition can be biased to favour the substrate simply by adding more substrate and hence making the reaction proceed towards ES and then product (P) formation. Inhibitor can only bind with free enzyme and not with the ES complex By increasing the substrate concentration, percentage inhibition decreases

#### (ii) Uncompetitive inhibition

An uncompetitive inhibitor binds to the enzyme at a site distinct from the substrate. As shown in equation 9, in uncompetitive inhibition the inhibitor can bind only with the ES complex and not with the free enzyme. In this case therefore, a high concentration of substrate is unable to reverse the inhibition, rather an increased substrate concentration actually increases the percentage inhibition



#### (iii) Noncompetitive inhibition

A non-competitive inhibitor is one that binds to a site distinct from that that binds the substrate. Inhibitor binding does not block substrate binding (or vice versa). This means that inhibitor and substrate can bind at the same time to an enzyme to form ES, EI, and ESI complexes (equation 10). The enzyme is inactivated when inhibitor is bound, whether or not substrate is also present. As such, noncompetitive inhibition completely depends upon the inhibitor concentration and is unaffected by the substrate concentration.



#### 1.9.1.2 Irreversible inhibition

Irreversible inhibitors are those that combine with or destroy a functional group on the enzyme that is essential for its activity. Formation of a covalent link between an irreversible inhibitor and an enzyme is common. In this type of inhibition the enzymatic activity cannot be restored by dilution or dialysis.

# 1.9.1.3 **Product inhibition**

The final step in an enzyme-catalysed reaction, the release of product, is often neglected, but is of course an essential component of the process. The product is bound to the active site by the same bonds that bind the substrate. As a result, product molecules are capable of binding to free enzyme to form enzyme-product complexes. In a single-substrate enzyme this is really a specialised form of competitive inhibition as substrate and product are mutually exclusive and compete for the enzyme active site as both cannot bind together.

# 1.9.2 Protease inhibition

In this work a variety of inhibitors have been investigated and thus a brief description of each is given. Comprehensive accounts of different protease inhibitors can be found in an excellent book [266].

# 1.9.2.1 Aprotinin

Aprotinin (also know as basic pancreatic trypsin inhibitor (BPTI), Kunitz protease inhibitor, and trasylol) is a competitive reversible serine protease inhibitor with an approximate molecular weight of 6,500. Most aprotinin-protease complexes dissociate at pH > 10 or < 3. It is water-soluble and due to its compact (pear-shaped) tertiary structure is relatively stable against denaturation due to high temperature, acid, alkalies, organic solvents, or proteolytic degradation. Numerous papers cover the biochemistry and kinetics of aprotinin [267 - 272].

Aprotinin has been used in numerous clinical settings, including open heart surgery and coronary artery bypass surgery [273], acute pancreatitis, adult respiratory distress syndrome, trauma, and septic shock. In the 1960s, aprotinin was used in adult cardiac surgery in an attempt to reduce excessive blood loss thought to be due to increased fibrinolysis. By 1984, the benefits of aprotinin in paediatric cardiopulmonary bypass surgery were being studied, and decreases in postoperative blood loss and fibrin split products were demonstrated [274]. The ability of aprotinin to blunt the inflammatory response is, in fact, equally as effective as corticosteroid treatment. Aprotinin has also been investigated as an anti-coagulant [275].

#### 1.9.2.2 Leupeptin

Leupeptin is a serine and cysteine competitive reversible inhibitor. It has a molecular weight of 475 6 and is a transition-state analogue, acting by mimicking the tetrahedral intermediate formed during peptide bond hydrolysis. Leupeptin is a tripeptide containing an aldehyde molety in place of the usual carboxylate. It is water-soluble and solutions are stable for just a few hours. More information on leupeptin can be found elsewhere [276, 277].

#### 1.9.2.3 3-Nitrophenylboronic acid

3-nitrophenylboronic acid is a reversible competitive serine protease inhibitor with a molecular weight of 166.93 and acts as a transition state analogue. Saccharides have been shown to act as a 'co-inhibitor' in the boronic acid inhibition system [278]. Investigation into the mechanism of binding of peptide boronic acids to proteases have been carried out [279, 280].

#### 1.9.2.4 Pepstatin A

Pepstatin A is a potent uncompetitive reversible inhibitor of pepsin and other aspartate proteases. It has a molecular weight of 685.9, is a pentapeptide-like compound, and is soluble in methanol and DMSO but insoluble in water. Pepstatin A was discovered in 1970 by Umezawa and co-workers [281] in the course of searching for inhibitors of therapeutically important enzymes. The crystal structure of human pepsin and its complex with pepstatin A has been determined [282].

# **CHAPTER 2**

# Materials, instrumentation and general procedures

# 2.1 MATERIALS AND INSTRUMENTATION

# 2.1.1 Reagents

REAGENT	SUPPLIER
Buffer reagents	
Cıtric acid, dihydrate	Sigma
Glycine	Sigma
Sodium carbonate, anhydrous	Sigma
Sodium dihydrogen orthophosphate	Fisher Scientific UK
Sodium hydrogen carbonate	Fisher Scientific UK
Sodium phosphate, dibasic anhydrous	Sigma
Enzymes	
$\alpha$ -chymotrypsin Type II: from bovine	
pancreas	Sigma
Pepsin: from Porcine stomach mucosa	Sigma
Protease Type XXI: from Streptomyces	
griseus	Sigma
Proteinase K: from Tntırachium album	Sigma
Trypsin Type II-S: from porcine pancreas	Sigma
Fluorescent labels	
BODIPY® 630 / 650-X, SE	Molecular Probes
Cy™ 5 bisfunctional dye	Amersham Pharmacia Biotech UK
Cy™ 5.5 bisfunctional dye	Amersham Pharmacia Biotech UK
Fluorescein isothiocyanate isomer I	Sigma

Naphthofluorescein	In-house [283]
Tetramethylrhodamine isothiocyanate	
mixed isomers	Sigma
Immobilisation supports	
Controlled pore glass, aminopropyl, 972	
Å mean pore diameter, 200 / 400 mesh	Cambio
Controlled pore glass, aminopropyl, 182	
Å mean pore diameter, 200 / 400 mesh	Cambio
UltraLink™ biosupport medium	Pierce
Inhibitors	
Aprotinin: from bovine lung	Sigma
4-hydroxycoumarin	Sigma
Leupeptin hemisulfate salt	Sigma
3-nitrophenylboronic acid	Sigma
Pepstatin A	Molecular Probes
Warfarin (3-(α-acetonylbenzyl)-4-	
hydroxycoumarin)	Sigma
Other reagents	
Bicinchoninic acid	Sigma
CHAPS (3-[3-cholamidopropyl]-	
dimethylamino)-1-propane sulfonate	Sigma
Copper (II) sulfate pentahydrate 98 %	Sigma
$\alpha$ -cyclodextrin	Sigma
β-cyclodextrin	Sigma
N, N-dimethylformamide (DMF)	Sigma
Dimethyl sulphoxide (DMSO)	Sigma
Ethanolamine	Fisher Scientific UK
Glutaraldehyde grade II	Sigma

2-hydroxypropyl-β-cyclodextrin	Sigma
Methanol spectrophotometric grade	Sigma
Sodium cvanoborohydride	Sigma
Lirea	Fisher Scientific LIK
0.04	
Proteins	
$\alpha$ -casein from bovine mılk	Sigma
Albumin, bovine	Sigma
Albumin, bovine – fluorescein	
isothiocyanate	Sigma

# Table 2.1 - Reagents used and their respective supplier

# 2.1.2 Cells

Quartz fluorescence cell (3500 $\mu l$ volume, 1 cm light path) $\hdots \ldots$	Hellma
Quartz UV cell (3500 $\mu l$ volume, 1 cm light path) $\ .$	Hellma
Flow cell (UV and fluorescence, 25 $\mu$ l volume,	
1.5 mm X 1.5 mm light path)	Helima

# 2.1.3 Chromatographic columns

PD-10 column ...... Amersham Pharmacia Biotech UK

# 2.1.4 Flow system equipment

Enzyme reactors (3.0 mm microbore chromatography column, 25 mm length with 0.17 ml bed volume, and 50 mm length with 0.35 ml bed volume), frits (stainless steel  $2\mu$ m, and PTFE 25  $\mu$ m), injection values

# 2.1.5 Pipettes

000 and 5000 Anachem	Automatic adjustable pipettes: P20,
-10 μl Eppendorf	Volu
10 μl Elkay	Volu

# 2.1.6 Spectrometers

Model F-4500 Fluorescence Spectrophotometer	Hıtachi
LS 50B Fluorescence Spectrophotometer Perkir	Elmer
Unicam 8700 Series	
UV / Visible spectrometer Unicam Analytical S	ystems

# 2.1.7 Supplier names and addresses

A list of supplier names and addresses is given in appendix A.

# 2.1.8 Temperature control

Water bath (0 – 99 °C) ..... Grant

# 2.2 GENERAL PROCEDURES

# 2.2.1 Buffer preparation

Buffers were prepared in triply deionised water throughout. Carbonate, citrate, glycine, and phosphate buffers have been used during experimental work.

# 2.2.2 pH measurement

pH measurements were carried out using a Whatman PHA 230 bench pH meter, with a pH range of 0 00 to 14 00 and an accuracy of  $\pm$  0 01 pH. The instrument was calibrated with high-resolution buffer solutions at pH 4.000, 7.000, and 10.000,  $\pm$  0.005, before use.

# 2.2.3 Water purification

Solutions were prepared using triply deionised water purified by a Maxima ultra pure water system (USF Elga Ltd.).

# 2.2.4 Mass determination

Weighing measurements were carried out on two balances depending on the weighing range needed:

- Precisa 300 series electronic balance, weighing range from 41 g to 12 400 g
- Precisa 40SM-200A four decimal place electronic balance

# 2.2.5 Pipetting

Pipetting was carried out using Gilson (supplied by Anachem), Eppendorf, and Elkay variable volume automatic pipettes.

#### 2.2.6 Fluorescence measurement

Two fluorimeters were used in this study – the Hitachi F-4500 Fluorescence Spectrophotometer, and the Perkin-Elmer LS 50B.

# 2.2.6.1 F-4500 Spectrophotometer

The appearance of the F-4500 spectrophotometer is shown in appendix B. The block diagram (taken from [284]) of the signal processing and control systems of the F-4500 can be seen in appendix C. The spectrophotometer mainframe has a xenon lamp light source, the beam of which is incident on the excitation monochromator. The beam after wavelength selection with the excitation monochromator ( $\lambda_{ex}$ ) is irradiated onto the sample cell. Light emitted from the sample in the form of fluorescence ( $\lambda_{em}$ ) then passes through the emission monochromator (which is at right angles to the incident light) to remove any light that is not at the emission wavelength. The intensity of fluorescence is detected via a red-sensitive photomultiplier tube. Instrument parameters are shown in appendix D.

#### 2.2.6.2 LS 50B Spectrophotometer

The appearance of the LS 50B Spectrophotometer is shown in appendix E. The optical layout of the LS 50B (taken from [285]) can be seen in appendix F. As

with the F-4500, the LS 50B is composed of a pulsed xenon excitation source, a scanning excitation monochromator, a scanning emission monochromator, and a red-sensitive R928 photomultiplier tube. The LS 50B is fitted with FL WinLab version 2.01 software and parameters for the LS 50B are shown in appendix D. The LS 50B is also fitted with a Fast Filter Accessory for the measurement of fluorescence polarisation (amongst other things), described in 2.2.7.

# 2.2.7 Fluorescence polarisation measurement

Fluorescence polarisation measurements were carried out on a Perkin Elmer LS 50B spectrophotometer using the Fast Filter Accessory. With this accessory, filters appropriate for the application are rotated into the excitation or emission beam. The rotation of the wheel is synchronised to the mains frequency so that when a filter is momentarily in the correct position in the beam, the source is triggered and energy passes through that filter. The wheel then rotates 90° before the next flash occurs. When the Fast Filter Application is not being used, the fast filter wheels are parked in the clear beam position with none of the filters in either beam. Each measurement takes into account a G-factor measurement to correct for instrumental bias.

# 2.2.8 Absorbance measurement

Absorbance measurements were carried out on an Unicam 8700 Series UV/Visible Spectrometer using distilled water or buffer solution to obtain a baseline.

# 2.2.9 Flow system measurements

Flow system measurements were taken using the basic set-up shown in figure 5.19, and variations thereof. Buffer solution was pumped using a peristaltic pump and injections of substrate, enzyme, and inhibitor into the flowing stream were made using manual sample injection valves fitted with 25  $\mu$ I sample loops. A manual 4-port rotary valve was used to direct the main flow stream into an incubation loop where assay components were captured. The incubation loop was situated in a thermostatted water bath to enable the reactions to occur at a defined temperature. The rotary valve was then again switched to direct the assay components back into the main stream. This flowed through a fluorescence flow cell situated in the Hitachi F-4500 fluorescence detector to measure the fluorescence intensity and then through to waste.

# 2.2.10 Preparation of dye:protein conjugates

Dye:protein conjugates were prepared according to the following method. FITC:albumin was prepared for fluorescence polarisation experiments, BODIPY: $\alpha$ -casein for single-labelled experiments, and FITC:TMRITC: $\alpha$ -casein and Cy 5:Cy 5.5: $\alpha$ -casein for double-labelled experiments. The preparation will be illustrated by the conjugation of FITC to albumin (in reaction mixture ratios of 50:1 and 100:1), but the general method is applicable to all conjugates prepared.

- 2.5 mg / ml albumin was prepared by dissolving 2.5 mg albumin in 1 ml
  0.1 M carbonate buffer pH 9.0.
- 2. Amount of FITC required for 50:1 and 100:1 reaction mixture ratios of FITC:albumin were calculated by multiplying the number of moles of

albumin by the molecular weight of FITC followed by multiplying by the ratio.

- 3. FITC solutions for each reaction mixture ratio were made up by dissolving the required amount of FITC in DMF.
- 4. For each reaction mixture ratio conjugate, the standard FITC solution in DMF was added to a 2.5 mg / ml albumin solution. The total volume was made up to 2.5 ml with 0.1 M carbonate buffer pH 9.0.
- 5. The solutions were incubated for 2 hours in the dark at room temperature.
- 6. Two PD-10 Sephadex columns were equilibrated each with 25 ml 0.1 M phosphate buffer pH 7.2.
- 7. Each conjugate was passed through a PD-10 column in order to remove unconjugated dye from the conjugate.
- 3.5 ml 0 1 M phosphate buffer pH 7.2 was added to each PD-10 column in order to elute the conjugate. Aliquots of conjugate were stored in black microcentrifuge tubes in the freezer.

# 2.2.11 Final dye:protein ratio calculations

Final dye protein (D:P) ratios were calculated using the Beer Lambert Law, which states that absorbance is related to the molar extinction coefficient  $\varepsilon$  (in cm<sup>-1</sup> M<sup>-1</sup>), concentration c (in M), and path length I (in cm) by A =  $\varepsilon$ cl. Absorbance values were taken for the conjugate at protein and dye wavelengths, which were corrected for dye absorbance at the protein wavelength and *vice versa*. Using the Beer Lambert Law, the concentration of both dye and protein was calculated
using the actual protein and dye absorbance values and their respective molar extinction coefficients. The dye concentration divided by the protein concentration gave a final dye protein ratio.

### 2.2.12 Immobilisation of enzyme onto controlled pore glass (CPG)

Proteolytic enzymes were immobilised onto CPG using a method previously described by Masoom and Townshend [120]. 30 ml 25 % aqueous glutaraldehyde (grade II) was added to 05 g aminopropyl CPG in a well-stoppered glass vial in the fume cupboard and left for 1.5 hours at room temperature on a roller bed. Nitrogen was passed through the solution to remove oxygen every 10 minutes for the first 40 minutes. The activated glass was then washed with distilled water.

0.5 g activated CPG was added to a cold (4 °C) solution of enzyme in 2.5 ml 0.1 M phosphate buffer pH 6.0 and placed on a roller bed. Nitrogen was bubbled through the solution every 10 minutes for 2.5 hours. The immobilised enzyme was washed first with cold 0 1 M phosphate buffer pH 6.0, then with cold distilled water to remove free enzyme and then stored moist at 4 °C.

To block unreacted groups 10 ml 3 0 M ethanolamine pH 9.0 was added to the beads. 10 mM sodium cyanoborohydride was also added to the beads to make the linking permanent. Initial mixing by shaking was carried out and then the beads were left standing overnight. The beads were then washed and stored in 0.1 M phosphate buffer pH 6.0 at 4 °C.

The reaction scheme for immobilisation of enzyme onto CPG is given in figure 6.3.

### 2.2.13 Immobilisation of enzyme onto Pierce UltraLink biosupport medium

Proteolytic enzymes were immobilised onto Pierce UltraLink biosupport medium according to the manufacturer's instructions. 0.125 g of UltraLink were weighed out. A solution of enzyme in 2 ml cold (4 °C) 0.1 M carbonate buffer pH 8 5 was added to the dry beads and the solution vortexed, followed by gentle rotation on a roller bed for 2 hours at room temperature The solution was then filtered to remove unbound enzyme from the beads and this filtrate was kept in order to determine the amount of enzyme not coupled to the beads using the BCA assay. A quench solution of 2.5 ml 3.0 M ethanolamine pH 9.0 was then added to the beads to block unreacted azlactone sites, vortexed and gently rotated on a roller bed for 2.5 hours at room temperature. The beads were then separated from the quench solution by filtration and resuspended in 2.5 ml 0.1 M phosphate pH 8.0 followed by vortexing and gentle rotation for 15 minutes. Buffer was removed by filtration and the beads were resuspended in 2.5 ml 1.0 M NaCl to remove nonspecifically attached protein. The solution was vortexed and rotated for 15 minutes after which the NaCI was removed by filtration and the beads resuspended in 0.1 M phosphate pH 8 0, the solution vortexed and gently rotated for 15 minutes followed by filtration to remove the buffer. This last step was repeated and the beads stored in 0.1 M phosphate buffer pH 8.0 at 4 °C. Figure 6.15 depicts the reaction scheme for immobilisation of enzymes onto UltraLink.

### 2.2.14 Bicinchoninic Acid (BCA) assay for determination of protein concentration

The BCA assay [286] was used to determine the amount of enzyme immobilised onto UltraLink biosupport medium. This method was adopted as the use of Triton® X-100 surfactant in the bead production may have interfered with an assay using absorbance at 280 nm for the measurement of uncoupled enzyme.

77

## **CHAPTER 3**

# Experimental: fluorescence polarisation protease assays

### 3.1 INTRODUCTION

The principle of how fluorescence polarisation is used to detect proteolytic activity using a fluorescent substrate was shown in figure 1.17 (a) and (b), and previous work that has been carried out in this area was discussed in section 1 8 3.1.

Lightly labelled substrates for use in fluorescence polarisation studies could offer an advantage over highly quenched fluorogenic substrates. It seems likely that highly derivatised molecules might hinder the proteolytic action of some proteases. The substrate may possess fluorescent ligands at a strategic position(s) that keeps the protein from being cleaved by some proteases due to steric hindrance.

Fluorescence polarisation is a very elegant technique with numerous advantages (discussed in section 1.2.2), the most prominent advantage over other fluorescence techniques being its relative immunity to the inner filter effect.

These obvious benefits gained from using fluorescence polarisation therefore lead to an interest in whether this technique could be suitably applied to the study of proteases

The aim of these experiments was to determine whether fluorescence polarisation is an appropriate method for the sensitive and rapid detection of protease enzymes.

### 3.2 PREPARATION OF FLUORESCEIN CONJUGATES

### 3.2.1 Procedures

Fluorescein: albumin conjugates were prepared according to the method described in section 2.2.10. A concentration of 2.5 mg / ml albumin was used and reaction mixture ratios of 7:1, 25:1, 50:1, 75:1, and 100:1 dye:protein were prepared.

All absorbance measurements were taken using an Unicam 8700 UV / visible spectrometer with a path length of 1 cm. A standard 3 ml quartz absorbance cuvette was used and baselines were taken with buffer prior to measurements.

### 3.2.2 Results

FITC solution was used to obtain an UV spectrum to calculate absorbance at 280 nm as a percentage of that at 494 nm. This was found to be 40 %, (appendix G), and this absorbance of the dye at the protein wavelength was taken into account during final dye:protein (D:P) ratio calculations (calculated as described in section 2.2.11).

Initial reaction mixture ratios and calculated final dye:protein molar ratios are given for each conjugate in table 3.1.

REACTION MIXTURE RATIO	CALCULATED DYE:PROTEIN RATIO			
7:1	1.70:1			
25:1	2.14:1			
50.1	1.80:1			
75:1	1 78:1			
100:1	1.61:1			

 Table 3.1 – Initial reaction mixture and calculated final dye:protein

 ratios for fluorescein:albumin conjugates

From table 3.1 it can be seen that the calculated final D.P ratios did not appear to truly reflect the number of dye molecules per protein molecule. It would be expected that as the amount of dye in the reaction mixture was increased, a concomitant increase would be seen in the final D.P ratio. However, the calculated final D:P ratios increased with the first increase in reaction mixture ratio and then decreased with increasing amounts of FITC in the reaction mixture, the opposite of what might be expected. For an unknown reason, the rules for the final D:P ratio calculations appeared to have broken down. One possibility was that the absorbance measurements were inaccurate. As the absorbance at 280 nm was much smaller than that at 494 nm, the error associated with the protein concentration was much larger. All protein absorbance values used for the calculation of protein concentration were below 0.4 absorbance units, and the most accurate readings are between 0.5 and 1.0 absorbance units. In addition, above reaction mixture ratios of 50:1, the peak at 494 nm was above 1.5 absorbance units, again outside the accurate range. Unfortunately due to the difference between the absorbance values of the 280 nm and 494 nm peaks, it was impossible to strike a balance and obtain accurate absorbance values. These inaccuracies could very well explain the apparent error in final D:P ratios. All ratios henceforth will therefore be given in terms of reaction mixture ratios.

The protocol for conjugate preparation (section 2.2.10) required an incubation time of 2 hours for the conjugation of dye to the protein before separation of free dye from the conjugate. This incubation time was increased for 4 different reaction mixture ratio conjugates to see if this had an effect upon the final dye:protein ratio. Incubation times of 2 hours, 5 hours, and 24 hours were investigated.



Figure 3.1 – Effect of incubation time on the final dye:protein ratio for different reaction mixture ratio conjugates

It can be seen from figure 3.1 that incubation time did not greatly affect the final dye:protein ratio. An incubation time of 2 hours was therefore kept for all preparations of conjugates.

### 3.3 FLUORESCENCE POLARISATION PROTEASE ASSAYS

### 3.3.1 Procedures

As mentioned previously, reaction mixture ratios of 7:1, 25:1, 50:1, 75:1, and 100:1 fluorescein: albumin were prepared. When subjected to polarisation measurements, only the 7:1 reaction mixture ratio conjugate resulted in a decrease in fluorescence polarisation upon incubation with protease. This conjugate was therefore used for further polarisation experiments. This was in line with findings by Jolley [202], who found that at high labelling ratios the initial fluorescence polarisation and assay span were reduced. At these ratios the \*fluorescence became highly quenched and the fluorescence polarisation dropped to very low values, probably due to a mechanism of depolarisation resulting from non-radiative energy transfer. A low initial substrate fluorescence polarisation value will make a poor substrate for these types of experiments, which monitor decreases in polarisation.

When substrate concentration was varied from  $0.5 - 2.3 \,\mu\text{g}$  / ml, no polarisation change was seen upon incubation with proteinase K except for at a concentration of 1.19  $\,\mu\text{g}$  / ml. This substrate concentration was therefore used in all experiments.

Fluorescence polarisation measurements were carried out on a Perkin Elmer LS 50B spectrophotometer using the Fast Filter Accessory as described in section 2.2.7. A G-factor (grating factor) of 0 916 was used to counteract the instrument polarisation response. An excitation wavelength of 488 nm and emission wavelength of 520 nm was used with excitation and emission bandwidths of 10 nm. A substrate concentration of 1.19  $\mu$ g / ml was added to buffer in a standard 3 ml quartz fluorescence cuvette and incubated with enzyme

in a total volume of 3 ml in a thermostatted water bath The cuvette was then transferred to the LS 50B and fluorescence polarisation readings taken.

To determine the precision of the fluorescence polarisation measurements taken on the LS 50B, 20 replicate readings were taken of a single fluorescein:albumin sample in buffer. Between each reading, the cuvette was removed from the holder and inverted three times. Measurements of 15, 0, 10, 0, 10, 5, 15, 0, 5, 10, 10, 5, 10, 5, 10, 15, 5, 20, 15, and 15 mP units were obtained corresponding to an average of 9 mP units with a standard deviation of 5 8 mP units.

The stability of the Perkin Elmer LS 50B spectrophotometer xenon lamp was tested using a fluorescence reference block of anthracene and naphthalene and an excitation wavelength of 296 nm. As can be seen from figure 3.2, the output of the xenon lamp was relatively stable over a period of 5 hours, which was suitable for carrying out experimental readings.





Three proteolytic enzymes – alkaline protease, proteinase K, and  $\alpha$ -chymotrypsin – were investigated in these experiments (descriptions in section 1.6.4).

### 3.3.2 Results

The effect of the following parameters on each enzyme were studied and optimised:

- [enzyme]
- incubation time
- temperature
- pH
- buffer system
- [buffer]

Results showing range of values studied and corresponding optimised values for each parameter are shown in table 3.2 for all three enzymes.

	[ENZYME] (µg / ml)	TIME (minutes)	TEMPERATURE (° C)	рН	BUFFER	[BUFFER] (mM)
ALKALINE PROTEASE						
RANGE STUDIED OPTIMUM	0 – 14 2.67	0 – 20 15	20 – 50 40	7 – 11 11.0	PH / CA CA	6 - 100 10
PROTEINASE K						
RANGE STUDIED OPTIMUM	0 – 14 6.67	0 – 20 10	20 – 60 40	7 – 11 9.5	PH / CA CA	5 - 100 10
α-CHYMOTRYPSIN						
RANGE STUDIED OPTIMUM	0 – 14 13.33	0 – 20 10	20 – 55 40	7 – 11 7.5	PH / CA PH	5 - 100 10

KEY: CA = Carbonate, PH = Phosphate

Table 3.2 – Assay parameters and optimum conditions for three protease enzymes using the substrate fluorescein:albumin As there were multiple enzymes and each behaved similarly with changes in assay parameters, the effect of temperature, pH, buffer system, and buffer concentration will be discussed only for alkaline protease.

### 3.3.2.1 Effect of temperature

The effect of temperature on the activity of alkaline protease was examined over the range 20 - 50 °C using a thermostatted water bath. The polarisation results were corrected for the changes in polarisation of the blank conjugate with temperature thereby giving specific results for enzyme activity. From figure 3.3 it can be seen that as temperature is increased from 20 - 40 °C there is an accompanying increase in enzyme activity. However, after 40 °C there is a decrease in enzymatic activity. This is probably due to the disruption of the quaternary structure of the enzyme at high temperature and hence a disruption of the catalytic site resulting in loss of activity.





Maximal enzyme activity, signified by the greatest decrease in fluorescence polarisation, occurred at 40 °C and this optimal temperature was therefore used in further experiments. However, as polarisation is temperature-dependent, the effects seen are a combination of temperature effects on the enzyme and polarisation of the fluorophore. As values were corrected for substrate polarisation, these effects are not dependent on the effect of temperature on the substrate.

### 3.3.2.2 Effect of pH and buffer system

The influence of pH and buffer system on the activity of alkaline protease was also investigated using a pH range of pH 7.0 - 11.0, and two buffer systems – carbonate and phosphate. The buffer concentrations were kept constant at 0.1 M.





As shown by figure 3.4, a relatively steady polarisation value was seen from pH 7.0 – 10.0 irrespective of buffer system. From pH 10.0 – 11.0 there was a sharp decrease in polarisation value with pH 11.0 showing a maximum decrease. This was expected, as the description of alkaline protease in section 1.6.4.1 states that this enzyme shows maximal activity at pH 11.0.

As optimal enzyme activity occurred at pH 11.0 (carbonate buffer), this pH and buffer system was used in subsequent experiments.

#### 3.3.2.3 Effect of buffer concentration

The influence of carbonate buffer pH 11.0 concentration between the values of 6 - 100 mM on the activity of alkaline protease was studied. As shown in figure 3.5, a maximum peak in activity was seen with a buffer concentration of 10 mM, with a reduction in activity either side of this value.





An optimised carbonate buffer pH 11.0 concentration of 10 mM was selected for further experiments as this concentration yields a maximum enzyme activity whilst maintaining sufficient buffering capacity.

### 3.3.2.4 Effect of incubation time

The influence of incubation time on the enzymatic reaction was evaluated between 2 to 20 minutes. It can be seen from figure 3.6 that as incubation time was increased the amount of proteolytic activity on the substrate also increased, represented by a drop in polarisation indicating the digestion of the protein substrate into smaller peptides. As incubation time was increased from 2 minutes to 15 minutes there was more time for the enzyme to act on the substrate and hence more product was formed. At 15 minutes a maximum was seen and further increases in incubation time produced no further increases in the enzymatic reaction. At this point another factor has become limiting, for example there is no substrate left due to the complete conversion to product.





Since a maximum in the enzymatic reaction occurred at 15 minutes of incubation, this was used in all subsequent experiments.

#### 3.3.2.5 Effect of enzyme concentration

Using the optimum conditions found for pH, buffer system, buffer concentration, temperature and time, it was possible to follow the effect of changing the concentration of alkaline protease on the hydrolysis of the substrate. Figure 3.7 shows that at a constant concentration of substrate, the reaction rate increased with increasing enzyme concentration until a maximal velocity was reached. This saturation effect was due to more substrate being converted to product by adding more enzyme to carry out the reaction. However, at an enzyme concentration of 2.67 µg / mI the enzyme was in excess and the substrate became limiting. Further increase in reaction rate was impossible without increasing the concentration of substrate because at an enzyme concentration of 2.67 µg / ml all the substrate was converted to product efficiently by this amount of enzyme.





A maximum reaction rate was seen at an alkaline protease concentration of 2.67  $\mu$ g / ml.

### 3.3.2.6 Comparison of the activity of three proteolytic enzymes using fluorescence polarisation

Using the optimised conditions shown in table 3.2 for each enzyme, the time dependent proteolysis of fluorescein: albumin by all three enzymes was compared. These results are shown in figure 3.8. From this comparison it can be seen that  $\alpha$ -chymotrypsin exhibited poor proteolytic activity on fluorescein: albumin when compared with proteinase K and alkaline protease. Alkaline protease had the greatest proteolytic activity, and proteinase K also has good proteolytic activity although lower than alkaline protease. This method has therefore proved to be applicable to the determination of proteolytic enzymes.





### 3.4 DISCUSSION

The use of covalently linked dye:protein conjugates for fluorescence polarisation studies assumes that the dye:protein conjugate is a rigid particle because then the rotational properties of the dye are identical to those of the macromolecule. The linkage between the fluorophore and the protein must be as short and as rigid as possible to avoid the "propeller effect" i.e. the fluorophore moving independently of the protein. The fluorophore must have the appropriate lifetime and as high a quantum yield and extinction coefficient as possible. Fluorescein fulfils these criteria.

It was suggested in section 3.1 that lightly labelled substrates for use in fluorescence polarisation studies might offer an advantage over highly quenched fluorogenic substrates. The use of fluorescein: albumin as a fluorescent substrate for the determination of protease enzymes using fluorescence polarisation has been investigated. It has been demonstrated that fluorescence polarisation has indeed successfully been used to analyse different proteases with broad specificity. This has enabled the comparison of the catalytic activities of alkaline protease, proteinase K, and  $\alpha$ -chymotrypsin with time.

However, polarisation changes are small, and in such cases, the accuracy of the assay suffers. Polarisation changes could be detected down to 5 mP, at which point noise prevented accurate readings from being taken.

The use of fluorescein as a label rather limited the choice of enzymes that could be looked at due to its pH sensitive fluorescence that is significantly reduced below pH 7 ( $pK_a \sim 6.4$ ). Longer wavelength fluorophores would be preferred due to the reasons discussed in section 1.1.3 of increased sensitivity. The BODIPY fluorophores would overcome both of these limitations found with fluorescein.

Chapter 3

These dyes are pH insensitive and can be obtained with a variety of excitation and emission wavelengths including the long-wavelength region of the spectrum. Experiments with BODIPY 630/650-X, SE as a label fluorophore were carried out without success using the substrate 5:1 BODIPY: $\alpha$ -casein. It has been observed that fewer BODIPY dye molecules are required to produce quenching effects than with fluorescein [255] and indeed 5:1 BODIPY: $\alpha$ -casein worked extremely well as an intramolecularly quenched protein substrate as discussed in chapter 5. It is apparent that only lightly labelled substrates are needed for this type of assay in comparison to the highly derivatised substrates in fluorescence enhancement experiments. This can be attributed to a high rate of depolarisation through non-radiative energy transfer in over-labelled substrates, as discussed in section 1.2.1.2. This was in line with findings by Jolley [202] who found that at high labelling ratios the initial fluorescence polarisation and span were reduced.

To further the studies of the application of fluorescence polarisation to the assay of proteolytic enzymes an investigation into the use of long-wavelength dyes needs to be carried out. This would constitute future work. Preferably a pH insensitive long-wavelength dye should be used due to advantages of a pHinsensitive dye and of long-wavelength fluorescence. However, such a dye would require a suitable fluorescence lifetime for appreciable Brownian rotation to occur. There are worries about the extent of polarisation effects with shortlifetime long-wavelength fluorophore molecules (section 1.2.1.3). This may be one of the few areas where long-wavelength fluorophores are not very useful. However, if a suitable long-wavelength fluorophore could be found this could prove valuable.

The usefulness of this technique is also limited by the requirement of a cuvettebased approach, with the associated low throughput and high reagent usage A flow-based system would improve this throughput significantly. To do this an evaluation of the application of fluorescence polarisation to this type of system would be needed in order to determine the related sensitivity.

Chapter 3

It seems reasonable to presume that polarisation methods would be less sensitive than straight fluorescence methods. Inherent problems in polarisation methods include (a) polarisation filters reducing exciting and emitted light intensities; (b) polarised light pre-selecting only a fraction of the potentially excitable molecules; and (c) polarisation changes being small and temperature sensitive. One disadvantage of fluorescence polarisation is that measurements are instrument dependent and the quality of the results is directly related to the quality of the instrument. Unfortunately the LS 50B spectrophotometer is not particularly sensitive and the availability of cheap sensitive fluorescence polarisation instruments is somewhat limited. Whilst fluorescence polarisation does possess attractive features in that it is simple, homogeneous, and relatively immune to the inner filter effect that is found in other homogeneous fluorescence techniques and so works in coloured solutions and cloudy suspensions (an obvious advantage in HTS), energy transfer techniques seem more promising. The decision was therefore taken to concentrate on the use of highly quenched fluorogenic substrates in protease assays.

## **CHAPTER 4**

# Experimental: double labelled FRET based protease assays

### 4.1 INTRODUCTION

The aim of this experimental section was to investigate the use of donor – \*acceptor fluorophore pairs for use in Fluorescence Resonance Energy Transfer (FRET) based protease assays.

The principle behind this work was similar to that described in section 1.8 3.2.1 and depicted in figure 1.18. However, in this work the principle described in section 1.8.3.2.1 has been modified such that instead of using a specific peptide sequence, the protein  $\alpha$ -casein has been used as the substrate. This protein was labelled with many donor and acceptor fluorophore molecules (unlike a single donor and acceptor molecule used with specific peptide sequences) with the idea that upon donor excitation, FRET from donor to acceptor would quench donor emission whilst enhancing acceptor emission. Upon proteolytic cleavage this effect is reversed and the increase in donor fluorescence and decrease in acceptor fluorescence is proportional to enzyme activity. The ability to measure both donor enhancement and acceptor quenching offers an advantage over other fluorescence techniques as the simultaneous observation of quenching and enhancement effects allows proteolytic cleavage effects to be distinguished from non-specific environmental effects on fluorescence intensity. As intensity ratio measurements are in effect being taken, instrument fluctuations will largely cancel out, which is a significant advantage over other energy transfer methods.

However, one disadvantage of this method is that, as shown in figure 4.1, several competing events can occur.

### Figure 4.1 (a, b, and c) – Alternative scenarios following excitation of the conjugate at the donor excitation wavelength



Figure 4.1 (a) shows excitation of the donor fluorophore followed by energy transfer to the acceptor in the protein conjugate, and subsequent disruption of this energy transfer caused by proteolytic cleavage of the substrate accompanied by donor enhancement and acceptor quenching. This sequence of events represents those that are needed for the protease assays. However, (b), (c), and (d) represent competing events that could also occur. Figure 4.1 (b) shows excitation of the donor and internal energy transfer between donor fluorophores, with the consequence that the substrate is virtually non-fluorescent. Figure 4.1 (c) depicts direct excitation of acceptor fluorophore by the donor wavelength and internal energy transfer between acceptor molecules causing quenching. Finally, figure 4.1 (d) shows direct excitation of either or both donor and acceptor molecules followed by emission with no energy transfer.

In reality, any of the situations shown in figure 4.1 may exist, as well as any combination of those situations. As a number of fates of the exciting light can occur, this makes interpretation of the results more complex and also requires much optimisation of the concentrations of donor and acceptor fluorophores needed to favour FRET from donor to acceptor, a requirement of this assay.

The donor-acceptor pairs of fluorescein isothiocyanate (FITC) / tetramethylrhodamine isothiocyanate (TMRITC), and Cyanine<sup>TM</sup> 5 (Cy 5) / Cyanine<sup>TM</sup> 5 5 (Cy 5.5) and the conditions required to synthesise a suitable substrate using these pairs for use in protease assays were investigated.

This work was carried out in conjunction with James E. Allard as a research project submitted in partial fulfilment of the requirements for the award of Master of Science.

## 4.2 INVESTIGATION INTO THE USE OF FLUORESCEIN AND TETRAMETHYLRHODAMINE FOR FRET PROTEASE ASSAYS

### 4.2.1<sup>±</sup> Procedures

Fluorescence measurements were carried out using a Perkin Elmer LS 50B fluorescence spectrophotometer with an excitation wavelength of 480 nm for FRET conjugate experiments. Excitation and emission wavelengths of 480 nm and 520 nm and 551 nm and 575 nm were used for fluorescein and tetramethylrhodamine free dyes respectively. Excitation and emission bandwidths of 10 nm were used throughout. Measurements were made statically in a standard 3 ml quartz fluorescence cuvette

All absorbance measurements were taken using an Unicam 8700 UV / Visible Spectrometer. A standard 3 ml quartz absorbance cuvette was used and baselines were taken with buffer prior to measurements

Fluorescein:tetramethylrhodamine: $\alpha$ -casein conjugates were prepared as described in section 2.2.10 using 2.5 mg / ml  $\alpha$ -casein with reaction mixture ratios of 10:40:1, 25:25:1, 1:50.1, 40:10:1, and 5:75:1. Final molar dye:protein ratios were not calculated due to the complex absorbance spectra obtained with these conjugates.

Fluorescein and tetramethylrhodamine were chosen for initial FRET studies as these dyes have promising spectral characteristics and are cheap.

### 4.2.2 Results

# 4.2.2.1 Analysis of the overlapping of fluorescein (donor) and tetramethylrhodamine (acceptor) excitation and emission spectra

Solutions of FITC and TMRITC were used to obtain the excitation and emission spectra for these dyes. As shown in figure 4.2, the emission spectrum of the donor fluorophore (fluorescein) overlapped the excitation spectrum of the acceptor (tetramethylrhodamine), a primary condition for FRET. This suggested that these dyes were suitable, in theory, for use in FRET protease assays.





Chapter 4

From figure 4.2 it can be seen that the excitation wavelength of fluorescein was 494 nm. However, using this excitation wavelength the influence of the scatter peak on the emission peak was too great and the two were in danger of being inadequately resolved. In addition to this it is evident from figure 4.3 that the absorbance spectra of fluorescein and tetramethylrhodamine strongly overlap, with the consequence that a fluorescein excitation wavelength would also directly excite the tetramethylrhodamine fluorophore. Direct acceptor excitation needed to be kept to a minimum, as excitation by FRET from the donor is required for the protease assays. For these reasons a sub-optimal excitation wavelength of 480 nm was used in all subsequent experiments.



Figure 4.3 – Absorbance spectra of fluorescein and tetramethylrhodamine free dyes

### 4.2.2.2 Determination of optimum donor:acceptor:protein ratios for use in FRET protease assays

Conjugates were prepared with reaction mixture ratios of 1:50:1, 5:75:1, 10:40:1, 25:25:1, and 40:10:1 fluorescein:rhodamine: $\alpha$ -casein to represent conjugates containing different ratios of dyes with tetramethylrhodamine in excess, fluorescein in excess, and equal quantities of donor and acceptor. As shown in figure 4.4, the 10:40:1, 25:25:1, and 40:10:1 conjugates all showed an overwhelming fluorescein emission peak. The tetramethylrhodamine emission peak at 575 nm was not evident because the fluorescein dye was so fluorescent that its peak completely hid any tetramethylrhodamine fluorescence.

Figure 4.4 – Emission spectra of various fluorescein:tetramethylrhodamine:α-casein conjugates





Figure 4.5 – Emission spectra of fluorescein:tetramethylrhodamine:α-casein conjugates exhibiting an emission peak at the TMRITC wavelength

It was found that tetramethylrhodamine emission was only observed in conjugates containing an excess of tetramethylrhodamine, as demonstrated by figure 4.5 which shows the fluorescence of the 5:75:1 and 1:50:1 conjugates.

These results suggest that optimum conditions require lightly labelled donor and heavily labelled acceptor conjugates for use in FRET protease assays. This was likely to be due to the fact that high donor ratios are more susceptible to quenching by the energy transfer mechanism. In these heavily donor labelled conjugates, fluorescein-to-fluorescein energy transfer competes with fluorescein-to-tetramethylrhodamine energy transfer. As shown by figure 4.2, the excitation and emission spectra of fluorescein overlap to a considerable degree. The fluorescein-to-fluorescein energy transfer is thus likely to be quite efficient. To favour fluorescein-to-tetramethylrhodamine energy transfer is thus likely to be quite efficient.

of acceptor is required. As the amount of acceptor increases, the probability of one or more TMRITC groups being attached close to the donor also increases: energy transfer is thus more efficient because of the lower mean distance between the fluorescein and tetramethylrhodamine groups.

A more defined tetramethylrhodamine peak was observed in the 5:75:1 conjugate when compared with the 1:50:1 conjugate. This was probably due to more transferred light from a higher concentration of fluorescein in the 5:75:1 conjugate being able to excite the tetramethylrhodamine fluorophore. As this conjugate looked to be more promising as a FRET substrate further experiments were carried out using the 5:75:1 conjugate.

### 4.2.2.3 Analysis of the 5:75:1 FITC:TMRITC:α-casein conjugate

5:1 FITC:α-casein and 75:1 TMRITC:α-casein conjugates were also prepared with the assumption that in theory these conjugates should contain roughly the same quantity of dye as the doubly labelled conjugate. This would obviously only be a crude estimate because in reality a great excess of the label to protein is normally used in labelling experiments, and possibly even an excess of label to linking (lysine) residues. It is thus not easy to argue that the degree of labelling in the single conjugate is going to be the same as in the double conjugate. If the linking reactions and their rates are assumed to be the same, there is going to be competition between fluorescein and rhodamine for a limited number of lysines, hence in this particular case rhodamine would be more likely to win and less fluorescein would be bound. Nevertheless, this experiment was carried out with a view to considering the rough rather than exact extent of direct excitation of FITC and TMRITC and any donor-to-donor or acceptor-to-acceptor energy transfer.

Tetramethylrhodamine would emit fluorescence only if excited. There were just two ways that this could occur: by direct excitation and by FRET from fluorescein emission.

The 75:1 TMRITC: $\alpha$ -casein conjugate was excited at 480 nm and as shown in figure 4.6 upon excitation at this wavelength an emission peak appeared. Unfortunately direct excitation of TMRITC by the fluorescein wavelength was expected due to the considerable overlap in their absorbance spectra (figure 4.3). However, it was assumed that this peak, and therefore its height, was solely due to the direct excitation of TMRITC. The 5:75:1 conjugate was excited at 480 nm and it could be concluded that the peak that resulted at the tetramethylrhodamine wavelength was an additive combination of direct excitation of TMRITC. It therefore followed that the difference in peak heights, in theory, could be attributed to the phenomenon of FRET between donor and acceptor.



Figure 4.6 – Fluorescence emission spectra demonstrating FRET between fluorescein and tetramethylrhodamine Chapter 4

To confirm these results, a 5:1 FITC: $\alpha$ -casein conjugate was prepared with the idea that the fluorescence emission spectrum of this conjugate should have a peak height above the 5:75:1 conjugate. The difference between these two heights would in theory be equivalent to that previously found with 75:1 conjugate as shown in figure 4.6. This was under the assumption that energy gained by TMRITC equals energy lost by FITC.

However, as demonstrated in figure 4.6, this was not observed experimentally. The 5:1 FITC: $\alpha$ -casein peak was much lower than the peak at the fluorescein wavelength found with the 5:75:1 FITC:TMRITC: $\alpha$ -casein conjugate. The most plausible reason for this observation was due to competition for binding sites on the protein molecules during conjugation. In the doubly labelled conjugate fluorescein and tetramethylrhodamine compete for the same sites to bind to whereas in the fluorescein: $\alpha$ -casein conjugate only fluorescein can bind. It would seem likely that more fluorescein is able to bind to the protein in the absence of competition i.e. in the fluorescein: $\alpha$ -casein conjugate may have produced a quenched substrate due to internal energy transfer between fluorescein molecules, accounting for the low fluorescence intensity found in this conjugate.

It was evident from figure 4.6 that the 75:1 TMRITC: $\alpha$ -casein conjugate displayed a peak in the fluorescein region, which was unexpected. An emission spectrum was taken of the TMRITC free dye obtained from Sigma with extreme care to prevent contamination. It was seen that the Sigma sample also contained a peak at the fluorescein wavelength and it was therefore concluded that the dye supplied by Sigma contained a contaminant. This emission at the fluorescein wavelength may have also contributed to the peak height found in the 5:75:1 conjugate at the fluorescein wavelength which would have further complicated the situation described above where 5:1 FITC: $\alpha$ -casein was synthesised to determine energy transfer.

### 4.3 INVESTIGATION INTO THE USE OF CY 5 AND CY 5.5 FOR FRET PROTEASE ASSAYS

### 4.3.1 Procedures

Fluorescence measurements were carried out using a Perkin Elmer LS 50B fluorescence spectrophotometer with an excitation wavelength of 649 nm for FRET conjugate experiments. Excitation and emission wavelengths of 649 nm and 670 nm and 675 nm and 694 nm were used for Cy 5 and Cy 5.5 free dyes respectively. Excitation and emission bandwidths of 10 nm were used and measurements were made statically in a standard 3 ml quartz fluorescence cuvette.

Cy 5:Cy 5.5: $\alpha$ -casein conjugates were prepared as described in section 2.2.10 using 1.0 mg / ml  $\alpha$ -casein and reaction mixture ratios of 2:30:1 and 5:20:1. As Cy 5 and Cy 5.5 were more expensive than either FITC or TMRITC, the amount of protein was reduced in order that the reaction mixture ratios could be scaled down to use lower concentrations of Cy dyes. It was decided that 2:30:1 and 5:20:1 Cy 5:Cy 5.5: $\alpha$ -casein conjugates should be synthesised, as it had previously been shown (section 4.2.2.2) that optimum conditions for use in FRET protease assays required lightly labelled donor and heavily labelled acceptor conjugates.

This FRET pair was chosen because these Cy dyes have suitable spectral characteristics, are pH-insensitive, and emit in the long-wavelength region (ca. 600 – 800 nm). Cy 5 and 5.5 have also previously been shown to act as an energy transfer pair [287, 288].

### 4.3.2 Results

### 4.3.2.1 Analysis of the overlapping of Cy 5 (donor) and Cy 5.5 (acceptor) excitation and emission spectra

Excitation and emission spectra were taken of the Cy 5 and Cy 5.5 free dyes. From figure 4.7 it can be seen that these dyes showed significant spectral overlap between the Cy 5 (donor) emission and Cy 5.5 (acceptor) excitation. This suggested that these dyes were ideal as a donor-acceptor pair for use in FRET protease assays.



#### Figure 4.7 – Spectral overlap of Cy 5 (donor) and Cy 5.5 (acceptor)

### **4.3.2.2** Analysis of the Cy 5:Cy 5.5:α-casein conjugates

Fluorescence emission spectra were taken of the 2:30:1 and 5:20:1 conjugates using an excitation wavelength of 649 nm. However, a broad emission peak was obtained with both conjugates over the spectral region of around 665 nm to 720 nm with no distinct Cy 5 or Cy 5.5 peaks. As the emission wavelengths of Cy 5 and Cy 5.5 were so close (670 nm and 694 nm respectively), it was likely that the broad peak was due to the merging of both the Cy 5 and Cy 5.5 peaks.

It was therefore concluded that the Cy 5 / Cy 5.5 dye pair was not ideal as an energy transfer pair. Unfortunately due to lack of time, the use of Cy 5 and Cy 5.5 was not pursued for use in protease assays. However, from these initial results it would appear that a great deal of work would need to be carried out in order to establish the optimum ratio of donor and acceptor fluorophores for energy transfer to occur and to be able to monitor both the donor enhancement and acceptor quenching effects that would be seen upon proteolytic cleavage of the substrate. In addition, the use of multivariate statistical analysis using a software package such as Unscrambler to resolve overlapping peaks may be needed.

### 4.4 DISCUSSION

Experiments were conducted to establish whether fluorescein / tetramethylrhodamine and Cy 5 / Cy 5.5 could be used as fluorophore pairs for use in protease assays based on the phenomenon of FRET.

Results suggested that optimum conditions required lightly labelled donor and heavily labelled acceptor conjugates for use in FRET protease assays. This was in agreement with findings by Lim *et al.*, who similarly found that these conditions were necessary for an energy transfer immunoassay for human serum albumin [41]. It was shown that the phenomenon of energy transfer may have occurred in the 5:75:1 FITC:TMRITC: $\alpha$ -casein conjugate. However, these results were extremely difficult to interpret due to the competing events shown in figure 4.1 and therefore not concrete evidence of energy transfer. As the synthesis of these substrates was not successful, the application of these substrates to protease assays was not carried out.

The initial results obtained with Cy 5 and Cy 5.5 were not promising and suggested that this would be a very complicated dye pair to work with.

One area that was not investigated, which would improve these assays and perhaps simplify the analysis of results, would be the effect of altering the fluorescence spectrophotometer bandwidth. Direct excitation of the acceptor fluorophore was a significant obstacle, as it occurred in addition to and in competition with the donor excitation. This happened because of the overlapping of the donor excitation and acceptor excitation spectra. Thus, at larger bandwidths the likelihood of direct acceptor excitation is much greater. The use of smaller bandwidths may have therefore improved the process of FRET between donor and acceptor and would be well worth investigating in any further

studies. However, although narrower bandwidths may show up more energy transfer, it would presumably be at the expense of sensitivity. Laser excitation may therefore be more helpful, which may allow less direct excitation of the acceptor at the same time as increasing the sensitivity.

It was obvious from experiments with both FITC / TMRITC and Cy 5 / Cy 5 5 that optimum conditions required for FRET to occur would need to be investigated to a much greater depth, which time did not permit. The synthesis of these doubly labelled conjugates was evidently not as simple or quick as that for the single labelled conjugates used in chapter 5 and so single labelled conjugates were pursued in this study for the determination of proteolytic enzymes and their inhibitors. However, although making doubly labelled conjugates of this kind is obviously more complex both to investigate and then to achieve in routine practice, if it can be done this kind of method may give advantages over other energy transfer methods.
## **CHAPTER 5**

# Experimental: single labelled substrate protease and protease inhibition assays

#### 5.1 INTRODUCTION

During screening for promising drug candidates, hundreds of thousands of compounds need to be assayed. The aim of these experiments was therefore to develop a simple, sensitive, and efficient method for assaying protease activity that would facilitate the testing of a large number of samples using HTS.

The application of a long-wavelength pH-insensitive intramolecularly quenched substrate to the determination of proteolytic enzymes and their inhibitors is demonstrated using the principle that was shown in figure 1.19. A substrate utilising  $\alpha$ -casein labelled with a long-wavelength BODIPY® dye has been synthesised and these assays have been incorporated into a flow injection analysis system that allows the rapid determination of protease inhibitors.

Fluorescein: albumin has previously been used in this type of assay [256, 257], but the substitution of a large number of positive lysine residues with the negative charges of fluorescein may cause conformational changes such that the interaction of the protein with the active site of the enzyme becomes problematic. However, BODIPY dyes have no charge and less dye is required for quenching [255]. This should minimise the effects of dye conjugation on the protein substrate conformation and enzyme-active site interactions.

In addition, the pH dependence of fluorescein has prompted the use of the pHinsensitive BODIPY dyes in place of fluorescein in this type of assay.

#### 5.2 PREPARATION OF BODIPY CONJUGATES

#### 5.2.1 Procedures

BODIPY<sup>®</sup> 630 / 650 – X, SE: $\alpha$ -casein conjugates (from this point referred to as BODIPY for ease) were prepared using the method described in section 2.2.10. A concentration of 2.5 mg / ml  $\alpha$ -casein was used and reaction mixture labelling ratios of 5:1, 15:1, and 45:1 BODIPY: $\alpha$ -casein were prepared To determine the final dye protein (D:P) conjugate ratios, the Beer-Lambert Law was used to calculate dye and protein concentrations, as shown in section 2.2.11.

All fluorescence spectra were obtained using an Hitachi F-4500 fluorescence spectrophotometer with a standard 3 ml quartz fluorescence cuvette.

All absorbance measurements were taken using an Unicam 8700 UV / Visible Spectrometer. A standard 3 ml quartz absorbance cuvette was used and baselines were taken with buffer prior to measurements.

An absorbance spectrum of a solution of BODIPY was taken and used to calculate the BODIPY absorbance at 280 nm as a percentage of that at 630 nm This was found to be 9.5 % (calculation in appendix H), which was taken into account in final D.P ratio calculations. The molar extinction coefficient of  $\alpha$ -casein was also determined. A 0.1 % solution of  $\alpha$ -casein was made up, the absorbance measured, and these absorbance values applied to the Beer-Lambert Law to calculate the molar extinction coefficient. This was found to be 21 151 cm<sup>-1</sup> M<sup>-1</sup> as shown in the calculations in appendix I. A molar extinction coefficient of 100 300 cm<sup>-1</sup> M<sup>-1</sup> was used for BODIPY. Absorbance values were corrected for dye absorbance at the protein wavelength of 280 nm (9.5 %) and for protein absorbance at the dye wavelength of 630 nm (0 %).

#### 5.2.2 Results

Calculated final D:P ratios along with initial reaction mixture ratios for three BODIPY: $\alpha$ -casein conjugates are given in table 5.1.

REACTION MIXTURE RATIO	CALCULATED DYE:PROTEIN RATIO
5:1	0.55:1
15:1	0.46:1
45:1	0.52:1

#### Table 5.1 – Initial reaction mixture and calculated final dye:protein ratios for BODIPY:α-casein conjugates

The results in table 5.1 show that as the amount of BODIPY in the reaction mixture was increased relative to the amount of  $\alpha$ -casein, the actual dye:protein ratio did not change. However, this seems illogical and suggests that for some reason the rules of D:P ratio calculations have broken down.

The absorbance spectra that were taken of the free BODIPY dye (figure 5.1 (a)) and of the three BODIPY: $\alpha$ -casein conjugates (figure 5.1 (b)) for the D:P calculations are shown below. As shown in figure 5.1, upon conjugation, the absorption peak at 630 nm of the BODIPY solution (a) unexpectedly split into two peaks at 597 nm and 642 nm in the conjugates (b).



### Figure 5.1 – Absorbance spectra of BODIPY solution (a) and BODIPY conjugates (b)

Wavelength (nm)

This phenomenon may be explained by H-dimer formation. This is a nonfluorescent complex of two neighbouring and aligned dye moieties that has an absorption maximum that is typically 20 – 40 nm shorter than the usual absorption maximum for a given chromophore. Similar behaviour has been demonstrated with tetramethylrhodamine and Cy dyes [287 - 289]. If this interpretation is correct then the following should be observed:

- Upon protease digestion of the labelled protein the 597 nm band should decrease relative to the 642 nm band
- Upon denaturation of the labelled protein the 597 nm band should decrease relative to the 642 nm band
- The 597 nm band should increase relative to the 642 nm band as the number of dyes attached per protein is increased
- The 597 nm peak should be absent on the fluorescence excitation spectrum of the labelled protein because it is due to absorption by the non-fluorescent dimer

Experiments have therefore been carried out to establish whether the above criteria are observed.

#### 5.2.2.1 Effect of protease digestion on the absorption characteristics of BODIPY:α-casein

As shown in figure 5.2, upon incubation with the protease alkaline protease, the absorbance spectrum of BODIPY: $\alpha$ -casein changed such that the 597 nm band decreased relative to the 642 nm band.



The fact that protease digestion of BODIPY: $\alpha$ -casein had the above effect suggests H-dimer formation.

#### 5.2.2.2 Effect of denaturation on the absorption characteristics of BODIPY:α-casein

Figure 5.3 shows the effect of denaturation through heating and urea upon the absorbance spectrum of BODIPY: $\alpha$ -casein conjugate.



It can be seen that the 597 nm band decreased relative to the 642 nm band upon denaturation, again suggesting the formation of H-dimers between the BODIPY molecules.

### 5.2.2.3 Effect of the number of BODIPY dyes attached to α-casein on the absorbance spectra of the conjugates

The absorbance spectra of the three BODIPY conjugates – 5:1, 15:1, and 45:1, were shown above in figure 5.1 (b). From these spectra it can be seen that as the number of BODIPY molecules attached to  $\alpha$ -casein was increased, the 597 nm band increased relative to the 642 nm band. This evidence further supports the H-dimer theory and also indicates that the D:P calculations shown in table 5.1 are incorrect. If the H-dimer theory is indeed correct, it follows that as the 597 nm band increased relative to the 642 nm band with increasing amounts of BODIPY in the reaction mixture, the number of BODIPY molecules attached to the  $\alpha$ -casein molecule also increased. The D:P calculations did not suggest this, hence adding more weight to the fact that the calculations are wrong.

#### 5.2.2.4 Comparison of the fluorescence excitation spectrum with the absorbance spectrum of BODIPY:α-casein

The fluorescence excitation spectrum of BODIPY: $\alpha$ -casein was obtained using excitation and emission bandwidths of 10 nm, a scan speed of 240 nm / min, and an emission wavelength of 655 nm. The excitation spectrum (figure 5.4) showed a lack of the 597 nm peak, which was apparent on the absorbance spectrum (figure 5.1(b)). This evidence also supports the theory that BODIPY H-dimers have been formed in the conjugate, as the dimer is non-fluorescent. However, the dimer does absorb light. As a change in structure has occurred, the dimer could really be thought of as another chromophore and so a double peak appears on the absorbance spectrum. The first peak is due to absorption by the dimer, the second by the single fluorophore molecule.



#### Figure 5.4 – Excitation spectrum of BODIPY: a-casein

#### 5.2.2.5 Re-evaluation of D:P calculations

Calculation of D:P ratios makes one big assumption: that the spectral characteristics of the free dye and that of the conjugated dye are identical. Clearly this is not the case with BODIPY and BODIPY: $\alpha$ -casein, as shown by figure 5.1. The shape of the absorbance spectrum changes upon conjugation. This would greatly affect the D:P calculations if not taken into consideration and would help to explain why the D:P ratios given in table 5.1 seem so illogical.

With this in mind, the D:P calculations were altered such that a wavelength of 350 nm was chosen to calculate the dye concentration using the Beer-Lambert Law. As highlighted by figures 5.1 (a) and (b), the shape of the BODIPY

#### Chapter 5 Experimental: single labelled substrate protease and protease inhibition assays

spectrum doesn't significantly change around 350 nm upon conjugation and so this wavelength was chosen for further calculations.

D:P calculations were carried out by measuring absorbances at 280 nm for  $\alpha$ casein and 350 nm for BODIPY, and were corrected for any dye contribution at the protein wavelength and *vice-versa*.  $\alpha$ -casein absorbance at 350 nm was found to be 11 % of that at 280 nm (appendix J), and BODIPY absorbance at 280 nm was found to be 35 % of that at 350 nm (appendix K). Molar extinction coefficients of 21, 151 cm<sup>-1</sup> M<sup>-1</sup> (appendix I) and 41, 627 cm<sup>-1</sup> M<sup>-1</sup> (appendix L) were used for  $\alpha$ -casein and BODIPY respectively.

Four BODIPY:α-casein conjugates were prepared as described in section 2.2.10 using reaction mixture ratios of 2:1, 5:1, 10:1, and 15:1. Final D:P ratios were calculated using the re-evaluated D:P calculation method with a dye wavelength of 350 nm. Initial reaction mixture and calculated final D:P ratios are given in table 5.2.

REACTION MIXTURE RATIO OF DYE:PROTEIN	FINAL DYE:PROTEIN RATIO		
2:1	0.71:1		
5:1	2.35:1		
10:1	3.93:1		
15:1	5.10:1		

Table 5.2 – Initial reaction mixture and calculated final dye:protein ratios for various BODIPY:α-casein conjugates

As would be expected, as the amount of dye in the reaction mixture was increased relative to the protein concentration, a concomitant increase was seen in the final D:P ratio. As this ratio was increased, it might be expected that the suitability of the substrate as a quenched protease substrate would also have increased. In practice, however, the conjugate with a final D:P ratio of 2.35 gave the best fluorescence enhancement after treatment with enzymes, and so this conjugate was used in all subsequent work.

#### 5.2.2.6 Calculation of percentage quenching in the 2.35 BODIPY:α-casein conjugate

In order to determine the amount of quenching that there was in the BODIPY: $\alpha$ casein conjugate chosen for all further work, solutions of free BODIPY dye and conjugate were taken and absorbance readings taken at 350 nm. These solutions were adjusted such that the absorbance value of both solutions was identical i.e. the concentration of dye was the same in each.

Fluorescence emission spectra were then taken of each solution to compare the fluorescence of free and conjugated dye to calculate the amount of fluorescence quenching in the substrate. An excitation wavelength of 605 nm was used with excitation and emission bandwidths of 5 nm. Fluorescence intensities were corrected for any buffer fluorescence. Intensities of 389.14 and 1444.24 were obtained for conjugate and free dye, respectively. This indicated that the 2.35 BODIPY: $\alpha$ -casein conjugate was 73 % guenched.

73 % quenching in the conjugate indicated that BODIPY:α-casein should indeed act as an efficient intramolecularly quenched protease substrate. However, if more time were available this percentage quenching could be improved which would increase assay sensitivity.

### 5.2.2.7 Effect of altering excitation wavelength on the emission spectrum of BODIPY:α-casein

Figure 5.4 showed that the optimum excitation wavelength of BODIPY: $\alpha$ -casein was 625 nm. However, the optimum emission wavelength of BODIPY: $\alpha$ -casein is 644 nm, which gives a Stokes shift of just 19 nm. Using excitation and emission bandwidths of 10 nm and a solution of BOBIPY: $\alpha$ -casein in a standard 3 ml quartz cuvette, emission spectra were taken using a variety of excitation wavelengths.





Figure 5.5 illustrates the increasing interference from the scatter peak on the emission peak as the excitation wavelength was increased to the optimum excitation wavelength. The use of the optimal excitation wavelength of 625 nm was therefore not practical because the Stokes shift was so small. Although a loss in fluorescence intensity was seen with lower excitation wavelengths, which will therefore decrease the sensitivity of the protease assays, a balance between sensitivity and interference was required and a sub-optimal wavelength of 605 nm was chosen for use in all subsequent experiments.

#### 5.2.2.8 Investigation into the effect of additives on the absorbance and emission spectra of BODIPY:α-casein

It has previously been shown in our lab that the fluorescence properties of a fluorophore can be changed by the appropriate choice of solvent environment [290]. This paper described the dependence of the optimum excitation wavelength of naphthofluorescein on the presence of additives such as cyclodextrins and (3-[3-cholamidopropyl]-dimethylamino)-1-propane sulfonate (CHAPS). Such effects were used to red-shift the excitation wavelength to match the output of a 635 nm diode laser for use in a simple and sensitive fluorescence detector.

The effect of cyclodextrins and CHAPS on the absorbance and emission wavelengths of BODIPY was therefore investigated to see if this phenomenon could be applied to assays involving BODIPY:α-casein.

It can seen from figure 5.6 that the inclusion of different cyclodextrins and CHAPS into the solvent environment of BODIPY:α-casein had no great effect upon the peak absorbance wavelengths. The use of these additives to shift the

Chapter 5 Experimental: single labelled substrate protease and protease inhibition assays

optimum excitation wavelength of BODIPY:α-casein towards that of the output of a 635 nm diode laser is therefore not applicable.

#### Figure 5.6 – Effect of cyclodextrins and CHAPS on the absorbance spectrum of BODIPY:α-casein



From figure 5.7 it is apparent that the inclusion of neither CHAPS nor the different cyclodextrins tested had much effect upon the emission wavelength of BODIPY: $\alpha$ -casein. The greatest effect (if any) was seen with 2.0 %  $\beta$ -cyclodextrin, and this was to shift the optimum to a lower wavelength. This would have no benefit at all, as it would only serve to lower the Stokes shift and introduce more error through scatter.



#### Figure 5.7 – Effect of cyclodextrins and CHAPS on the fluorescence emission spectrum of BODIPY:α-casein

5.2.2.9 Stability of BODIPY:α-casein

In order to determine the best condition for BODIPY:α-casein storage, BODIPY:α-casein was kept in brown bottles in a cupboard at room temperature and in the fridge, and also in aliquots in microcentrifuge tubes wrapped with black tape in the freezer. Over a period of 3 months samples were taken and the fluorescence monitored. All samples were first left to equilibrate to room temperature and then incubated in a water bath at 18 °C for 5 minutes. Emission spectra were then taken using an excitation of 605 nm and excitation and emission bandwidths of 10 nm. All readings were taken first thing in the morning. The effects of these different storage conditions are shown in figure 5.8.



#### Figure 5.8 – Effect of storage conditions on the fluorescence characteristics of BODIPY:α-casein

From figure 5.8 it is evident that the ideal storage condition for BODIPY: $\alpha$ -casein was in the freezer. For the first 20 days the stability of BODIPY: $\alpha$ -casein with respect to its fluorescence properties remained relatively constant with all three storage conditions. After 20 days however, both the fridge and room temperature samples began to increase in fluorescence intensity. This was most likely due to degradation of the quenched conjugate causing an increase in fluorescence due to abolished internal energy transfer. The freezer samples however remained relatively constant over a period of 3 months. All BODIPY: $\alpha$ -casein conjugate was thus stored in the freezer in 30 µl aliquots in microcentrifuge tubes surrounded by black tape, and thawed as required.

As BODIPY: $\alpha$ -casein was made up in buffer for use in protease and protease inhibition experiments, the stability of BODIPY: $\alpha$ -casein in solution was tested.

Conjugate was prepared in glycine and carbonate buffers at different pH values. The fluorescence of the conjugate was measured and plotted as a percentage decrease in fluorescence over time after the solution had been made whilst sitting at room temperature in the lab (figure 5.9).

#### Figure 5.9 – Stability of BODIPY:α-casein in aqueous solution represented by the percentage decrease in fluorescence intensity over time



Time after solution made up (minutes)

It was noticeable from figure 5.9 that in lower pH solution the BODIPY:α-casein conjugate was unstable. At higher pH values the substrate was more stable and it was seen that BODIPY:α-casein was most stable in carbonate buffer at a pH of 10.0. BODIPY:α-casein diluted in this buffer gave a percentage decrease in fluorescence intensity of 10 % over 253 minutes, which was adequate for carrying out experiments. The drop in fluorescence over time was most likely due to photodecomposition.

#### 5.3 PROTEASE KINETICS

Simple experiments were carried out to illustrate the use of the fluorogenic BODIPY: $\alpha$ -casein substrate in enzyme kinetics. This was not part of the objective of this work and was purely to demonstrate that enzyme kinetics can be done. The results of these experiments are therefore not strictly important, the aim was to show the principle. A discussion of enzyme kinetics was given in section 1.6.3.

#### 5.3.1 Procedures

Fluorescence measurements were carried out statically in a 3 ml standard quartz cuvette using an Hitachi F-4500 fluorescence spectrophotometer with an excitation wavelength of 605 nm and excitation and emission bandwidths of 10 nm. A range of increasing substrate concentrations was used and the fluorescence intensity measured and converted to fluorescence intensity units per minute. Lineweaver-Burk graphs were then plotted for the proteolytic cleavage of BODIPY: $\alpha$ -casein by proteinase K,  $\alpha$ -chymotrypsin, and pepsin, as shown in figure 5.10. Measurements were carried out in triplicate and error bars represent plus and minus one standard deviation.

#### 5.3.2 Results

#### Figure 5.10 – Lineweaver-Burk plots for the proteolytic digestion of BODIPY:α-casein by various proteases



From figure 5.10 values of  $K_M$  and  $V_{max}$  were determined and these are given in table 5.3.

PROTEOLYTIC ENZYME	<i>К<sub>М</sub></i> (nM)	V <sub>max</sub> (fluorescence units / min)
PROTEINASE K	17.86	312.5
α-CHYMOTRYPSIN	17.24	86.21
PEPSIN	129.72	312.5

Table 5.3 –  $K_M$  and  $V_{max}$  values obtained experimentally for various proteolytic enzymes on the substrate BODIPY: $\alpha$ -casein

Figure 5.10 demonstrates that the proteolytic digestion of BODIPY: $\alpha$ -casein by proteinase K,  $\alpha$ -chymotrypsin, and pepsin obeyed classical Michaelis-Menten kinetics. Values of  $K_M$  and  $V_{max}$  have also been determined (table 5.3). Comparing  $K_M$  values indicates that BODIPY: $\alpha$ -casein is a more efficient substrate for proteinase K and  $\alpha$ -chymotrypsin than for pepsin.

#### 5.4 PROTEASE ASSAYS IN A STATIC SYSEM

#### 5.4.1 Procedures

Fluorescence measurements were made for the protease assays using an Hitachi F-4500 fluorescence spectrophotometer with excitation and emission bandwidths of 10 nm and an excitation wavelength of 605 nm. Static measurements were carried out in a standard guartz fluorescence cuvette.

Five protease enzymes, each with varying pH optima, ranging from pH 2.5 - 9.0, were investigated in these experiments – alkaline protease, proteinase K,  $\alpha$ -chymotrypsin, trypsin, and pepsin. A description of each is given in section 1.6.4.

In order to test the stability of the F-4500 fluorescence spectrophotometer, the fluorescence of a rhodamine B reference block was monitored over time. An excitation wavelength of 558 nm with excitation and emission bandwidths of 2.5 nm was used. An emission scan was performed immediately after the spectrophotometer had been switched on and then every 30 minutes thereafter. The shutter between the excitation beam and reference block was shut between readings so as not to heat or photobleach the block. A measurement was also taken of the Xe lamp output along side the fluorescence readings. These results are shown below in figure 5.11.





Figure 5.11 indicates that although the Xe lamp output remains relatively constant over 9 hours, the fluorescence reading of a reference block increases rapidly over the first hour after the F-4500 had been switched on. After this initial instability the readings were sufficiently stable. For all experiments that were carried out therefore, the F-4500 was switched on and allowed to stabilise for 1 hour before fluorescence readings were taken.

#### 5.4.2 Results

#### 5.4.2.1 Optimisation of protease assay conditions

Assay conditions for each enzyme were optimised univariately. The effect of the following parameters on the enzyme assay were studied:

- [enzyme]
- incubation time
- temperature
- pH
- buffer system
- [buffer]

The assay parameters studied and optimum conditions found for each enzyme are shown below in table 5.4.

#### KEY: CA = Carbonate, CI = Citrate, GL = Glycine, PH = Phosphate

	[ENZYME] (µg/ml)	TIME (minutes)	TEMPERATURE (°C)	рН	BUFFER	[BUFFER] (mM)
ALKALINE PROTEASE						
RANGE STUDIED OPTIMUM	0 – 7 2.65	0 - 30 10	20 – 55 40	7 – 11 9.0	PH / CA CA	5-75 10
PROTEINASE K						
RANGE STUDIED OPTIMUM	0 - 17 9.9	1 – 20 5	20 – 45 40	7 – 11 7.0	PH / CA PH	6 – 90 8
RANGE STUDIED OPTIMUM	0 – 17 3.3	1 – 20 10	20 – 45 40	7 – 9 7.5	PH PH	5 – 100 10
PEPSIN						
RANGE STUDIED OPTIMUM	0 – 17 13.2	1-20 10	20 – 55 <b>50</b>	1 – 3 2.5	GL / CI GL	10 - 100 50
TRYPSIN						
RANGE STUDIED OPTIMUM	0 – 22 4.29	1 – 20 15	20 – 50 45	5-9 7.0	CI / PH PH	8 – 90 10

### Table 5.4 - Assay parameters studied and optimum conditions for various proteolytic enzymes using the fluorogenic substrate BODIPY : α-casein in a static system

As optimisation experiments will be described in detail for the protease assays carried out in a flowing system, individual optimisation experiments for the assay conditions for the 5 proteolytic enzymes investigated here will not be described.

Under the optimised conditions given in table 5.4, BODIPY: $\alpha$ -casein was incubated with the protease proteinase K. As shown in figure 5.12, incubation of conjugate with proteinase K gave a fluorescence increase of approximately 9-fold, indicating that BODIPY: $\alpha$ -casein is promising as a fluorogenic protease substrate.



#### 5.4.2.2 Calibration curves for BODIPY:α-casein

The effect of substrate concentration was examined for each of the proteolytic enzymes under optimised conditions. Readings were taken in triplicate and error bars show plus and minus one standard deviation.

#### Figure 5.13 – Influence of BODIPY:α-casein concentration on the production of BODIPY when incubated with a variety of proteases in a static system under optimised conditions



Chapter 5

The activities of a variety of proteolytic enzymes on the substrate BODIPY: $\alpha$ casein were compared, as shown in figure 5.13. These enzymes represent proteases with activities over a wide pH range – 2.5 to 9.0. It was therefore possible to compare the activities of acid, neutral, and alkaline proteases of different classes (pepsin being an aspartic protease and alkaline protease, proteinase K, trypsin, and  $\alpha$ -chymotrypsin being serine proteases) on the same graph using the same substrate merely by changing the buffer system. It can be seen that alkaline protease is by far the most efficient protease, followed by proteinase K. Trypsin and  $\alpha$ -chymotrypsin have extremely similar efficiencies, as would be expected from these closely related enzymes. Pepsin also has a lower activity towards BODIPY: $\alpha$ -casein than both alkaline protease and proteinase K.

#### 5.4.2.3 Determination of limits of detection

Limits of detection, defined as the analyte concentration giving a signal equal to the blank signal plus 3 standard deviations of the blank [291], were determined for each of the five proteolytic enzymes in the static system. Fluorescence intensity against enzyme concentration was plotted for the linear portion of the graphs for each of the enzymes, which are shown in figures 5.14 to 5.18. Measurements were taken in triplicate and error bars are given which represent plus and minus one standard deviation.





Figure 5.15 – Fluorescence intensity against enzyme concentration to calculate the limit of detection of proteinase K



140



#### Figure 5.16 – Fluorescence intensity against enzyme concentration to calculate the limit of detection of α-chymotrypsin





141



#### Figure 5.18 – Fluorescence intensity against enzyme concentration to calculate the limit of detection of pepsin

Limits of detection for each proteolytic enzyme using the substrate BODIPY: $\alpha$ -casein are given in table 5.5.

PROTEOLYTIC ENZYME	LIMIT OF DETECTION (nM)		
ALKALINE PROTEASE	0.880		
PROTEINASE K	4.630		
α-CHYMOTRYPSIN	0.208		
TRYPSIN	0.148		
PEPSIN	0.771		

#### Table 5.5 – Limits of detection determined experimentally for various proteolytic enzymes using the substrate BODIPY:α-casein in a static system

Limits of detection in the nM region have been achieved for all five proteolytic enzymes using this system. These experiments indicate that BODIPY:α-casein is promising as a sensitive fluorogenic substrate for the assay of proteolytic enzymes. As these preliminary experiments have proven successful, BODIPY:α-casein was studied in a flow analysis manifold with four enzymes at different pH values. Such a flow set-up would be well suited for use in the high throughput screening of candidate drug molecules.

#### 5.5 PROTEASE ASSAYS IN A FLOWING SYSTEM

#### 5.5.1 Procedures

Fluorescence measurements were made using an F-4500 fluorescence spectrophotometer with excitation and emission bandwidths of 10nm and time scans were performed with excitation and emission wavelengths of 605 nm and 644 nm respectively. Flow system measurements were carried out using the setup shown in figure 5.19.



Figure 5.19 – Manifold for flow injection enzyme and enzyme inhibition assays

(A = buffer, B = peristaltic pump, C = conjugate injection valve, D = enzyme or enzyme / inhibitor injection valve, E = switching valve, F = incubation loop, G = thermostatted water bath, H = fluorescence detector, I = waste) The sequences of events that occur in these flow system experiments are depicted in figure 5.20.

### Figure 5.20 – Sequence of events for the determination of proteases and their inhibitors in a flow system



Photos are shown below depicting the actual set-up of this flow system in our laboratory (figures 5.21 (a) and (b)).

Figures 5.21 (a) and (b) – Actual layout and equipment used for flow anaysis protease and protease inhibition assays





A flow rate of 1.85 ml / min (calculations in appendix M) was used for all flow system experiments. The volume of the incubation loop was calculated to be 670  $\mu$ l (calculations in appendix N).

Four protease enzymes, each with varying pH optima, ranging from pH 2.5 - 9.5, were investigated in these experiments – alkaline protease, proteinase K,  $\alpha$ -chymotrypsin, and pepsin. A description of each is given in section 1.6.4.

#### 5.5.2 Results

Assay conditions for each enzyme were optimised. The effect of the following parameters on the enzyme assay were studied:

- [enzyme]
- incubation time
- temperature
- pH
- buffer system
- [buffer]

The assay parameters studied and optimum conditions found for each enzyme are shown below in table 5.6.
	[ENZYME] (µg/ml)	TIME (minutes)	TEMPERATURE (°C)	рН	BUFFER	[BUFFER] (mM)
ALKALINE PROTEASE RANGE STUDIED OPTIMUM	0 – 1.87 0.75	1.0 - 3.0 2.0	25 – 50 40	7 – 11 9.5	PH / CA CA	10 – 100 10
PROTEINASE K RANGE STUDIED OPTIMUM	0 – 1.87 1.49	1.0 – 3.0 1.5	25 – 50 35	7 – 11 9.5	PH / CA CA	10 – 100 20
α- CHYMOTRYPSIN RANGE STUDIED OPTIMUM	0 – 1.49 0.75	1.0 – 3.0 1.5	25 – 50 35	7 – 9 7.5	РН РН	10 – 100 10
PEPSIN RANGE STUDIED OPTIMUM	0 – 44.78 37.31	1.0 - 3.0 1.5	25 – 50 35	1 – 3 2.0	GL / CI GL	10 – 100 20

#### KEY: CA = Carbonate, CI = Citrate, GL = Glycine, PH = Phosphate

Table 5.6 - Assay parameters studied and optimum conditions for various proteolytic enzymes using the fluorogenic substrate BODIPY : α-casein in a flow system

The effect of assay parameters only on the proteolytic enzyme pepsin will be discussed in detail as it was found that similar trends were observed with all four proteolytic enzymes. Optimisation experiments were carried out in duplicate and error bars represent plus and minus one standard deviation. All readings were corrected for any effects of the assay parameters upon the conjugate.

#### 5.5.2.1 Effect of temperature

Effect of temperature on the activity of pepsin was examined over the range 25 – 50 °C using a thermostatted water bath. As shown in figure 5.22, there was an increase in the reaction rate as the temperature was increased from 25 °C to 35 °C. A maximum level was reached at 35 °C after which the reaction rate declined with further increase of temperature. At high temperature, rapid denaturation of the quaternary protein structure occurred leading to a disruption of the enzymatic activity.



Figure 5.22 – Effect of temperature on the activity of pepsin

A temperature of 35 °C was selected for subsequent experiments since a maximum reaction rate was seen at this temperature. This was expected as most biological enzymes operate at 37 °C.

# 5.5.2.2 Effect of pH and buffer system

Chapter 5

Effect of buffer system and pH on the activity of pepsin was also investigated. A pH range from 1.0 - 3.0 in 0.5 increments was investigated using citrate and glycine buffers. As shown in figure 5.23, pH and buffer system both dramatically affected the reaction rate. Higher fluorescence intensities were observed with glycine buffer than citrate buffer and a pH optimum of 2.0 was seen with glycine buffer. On both sides of this pH value, the fluorescence dropped drastically.

Figure 5.23 – Effect of pH and buffer system on the activity of pepsin



As a maximum reaction rate was achieved with glycine buffer at a pH of 2.0, this buffer system and pH was used for all subsequent experiments.

### 5.5.2.3 Effect of buffer concentration

The effect of glycine buffer pH 2.0 concentration was optimised. A concentration range of 10 to 100 mM was studied and as shown in figure 5.24, a peak in pepsin activity was seen at a concentration of 20 mM. At higher buffer concentrations a drop in the reaction rate occurred.



Figure 5.24 – Effect of buffer concentration on the activity of pepsin

Buffer concentration affected the reaction rate of pepsin on the substrate BODIPY:α-casein. A concentration of 20 mM was chosen for further work as this concentration gave an optimal reaction rate and was sufficient to maintain buffering capacity.

# 5.5.2.4 Effect of incubation time

The time of incubation of pepsin and BODIPY: $\alpha$ -casein together was investigated over the range 1.0 to 3.0 minutes in 0.5 minute increments. Figure 5.25 demonstrates that incubation time did not greatly affect the reaction rate.



Figure 5.25 – Effect of incubation time on the activity of pepsin

As incubation time had no great effect upon the enzymatic activity of pepsin, an incubation time of 1.5 minutes was selected for further experiments. This was chosen as it is beneficial to carry out experiments for high throughput screening rapidly and difficulties were found using an incubation time of 1.0 minute when carrying out injections and switching manually.

#### 5.5.2.5 Effect of pepsin concentration on the reaction rate

Pepsin concentration was varied between 0 and 44.78  $\mu$ g / ml. It can be seen from figure 5.26 that between 0 and 15  $\mu$ g / ml the formation of product proceeds at a rate which is linear with increasing enzyme concentration. In this region therefore, the reaction is independent of substrate concentration i.e. substrate is present in excess and the reaction is a 'zero order' reaction. However, as substrate is used up, the enzyme's active sites are no longer saturated, and substrate concentration becomes rate limiting and the reaction becomes first order. This is signified by the graph reaching a plateau. Ideally to measure enzyme activity, measurements must be made in that portion of the graph where the reaction is zero order because the reaction is then independent of substrate concentration of the graph where the reaction of pepsin was therefore carried out in the region 0 to 4  $\mu$ g / ml, as described in section 5.5.2.7.



# Figure 5.26 – Influence of pepsin concentration on the formation of product when incubated with BODIPY:α-casein

# 5.5.2.6 Calibration graphs for BODIPY:α-casein in the flowing system

Using the optimum assay parameters for each enzyme given in table 5.6, a calibration graph for BODIPY: $\alpha$ -casein was obtained by injecting a series of standard solutions of BODIPY: $\alpha$ -casein into the flow system with each of the enzymes. Measurements were made in triplicate and error bars show plus and minus one standard deviation. Each measurement was carried out separately. The BODIPY: $\alpha$ -casein conjugate that was used with alkaline protease was a different batch from that used with proteinase K,  $\alpha$ -chymotrypsin, and pepsin. For this reason alkaline protease has been plotted on a separate graph.

Figure 5.27 – Influence of substrate concentration on the production of BODIPY fluorescence when incubated with alkaline protease in a flowing system



154



The calibration graphs obtained for BODIPY:α-casein show good linearity in the ranges investigated. In this region enzyme is not limiting and so increases in substrate concentration are accompanied by an increase in reaction rate, as more substrate is available to be converted to product by the enzyme.

Figure 5.28 shows that different proteases act upon the substrate with different efficiencies. Proteinase K cleaves the substrate with a greater activity than either  $\alpha$ -chymotrypsin or pepsin. These results do show that the activities of a variety of enzymes with different pH optima can be compared using the same substrate merely by altering the pH and buffer system.

# 5.5.2.7 Determination of limits of detection

Limits of detection, defined as the analyte concentration giving a signal equal to the blank signal plus 3 standard deviations of the blank [291], were determined for each of the four proteolytic enzymes in the flow analysis system. Fluorescence intensity against enzyme concentration was plotted for the linear portion of the graphs for each of the enzymes, which is the analytical useful region for the determination of enzymes due to the fact that the reaction is independent of substrate concentration i.e. zero order reaction. These results are shown in figures 5.29 to 5.32. Measurements were taken in triplicate and error bars are given which represent plus and minus one standard deviation.







#### Figure 5.30 – Fluorescence intensity against enzyme concentration to calculate the limit of detection of proteinase K

Chapter 5

Figure 5.31 – Fluorescence intensity against enzyme concentration to calculate the limit of detection of α-chymotrypsin





# Figure 5.32 – Fluorescence intensity against enzyme concentration to calculate the limit of detection of pepsin

Calculated limits of detection for each proteolytic enzyme using the substrate BODIPY: $\alpha$ -casein in a flow analysis system are given in table 5.7.

PROTEOLYTIC ENZYME	LIMIT OF DETECTION
ALKALINE PROTEASE	10 pM
PROTEINASE K	1.07 nM
α-CHYMOTRYPSIN	0.21 nM
PEPSIN	4.49 nM

Table 5.7 – Limits of detection determined experimentally for various proteolytic enzymes using the substrate BODIPY:α-casein in a flowing system Sensitive detection limits have been determined for these proteolytic enzymes in the nM (proteinase K,  $\alpha$ -chymotrypsin, and pepsin) and pM (alkaline protease) range. If these are compared with those obtained in the static system (table 5.5) it can be seen that the detection limits of both alkaline protease and proteinase K have been significantly improved. Those of  $\alpha$ -chymotrypsin are identical, whereas with pepsin it was found that the static system gave a lower limit of detection than the flowing system. However, in all cases the time taken to carry out the assays was much improved in the flowing system compared with the static system as depicted in table 5.8. This has a great advantage when applying these protease assays to a high throughput screening format.

PROTEOLYTIC ENZYME	INCUBATION TIME IN STATIC SYSTEM (mins)	INCUBATION TIME IN FLOWING SYSTEM (mins)	
ALKALINE PROTEASE	10	2.0	
PROTEINASE K	5	1.5	
α-CHYMOTRYPSIN	10	1.5	
PEPSIN	10	1.5	

Table 5.8 – Comparison of assay incubation times for various proteolytic enzymes in static and flow system formats The experiments that have been carried out in the flow analysis system demonstrate that the substrate BODIPY:α-casein is a sensitive fluorogenic protease substrate and that the flow-based procedure that has been described here is well suited for use in high throughput screening.

#### 5.5.2.8 Comparison with previously published work

A quenched BODIPY dye-labelled casein substrate for the assay of proteases has previously been described by Jones *et al.* [255] and was described in section 1.8.3.2.2. In this study a 96-well microtiter plate format was used with the BODIPY dyes BODIPY FL and BODIPY TR-X. These had excitation / emission wavelengths of 485 nm / 530 nm, and 590 nm / 645 nm, respectively. The limits of detection (LOD) and the times taken to carry out the assays are given in table 5.9 and compared with those that have been found in this study. Limits of detection have been given in terms of enzyme units as defined by the supplier. To convert the limits of detection given in table 5.7 from molar concentrations to enzyme units, enzyme activities defined by the supplier of 21 Units / mg protein, 10.2 Units / mg protein, 51 Units / mg protein, and 4500 Units / mg protein were used for alkaline protease, proteinase K,  $\alpha$ -chymotrypsin, and pepsin respectively. Use of detection limits in enzyme units allows for a fair comparison.

	M.L.Beaumont		Jones <i>et al</i> .		
	LOD (Units)	TIME (mins)	LOD (Units)	TIME (mins)	
ALKALINE PROTEASE	2.1 X 10 <sup>-6</sup>	2.0	1 X 10 -4	60	
PROTEINASE K	6.3 X 10 <sup>-5</sup>	1.5	1 X 10 -4	60	
α-CHYMOTRYPSIN	5.5 X 10 <sup>-5</sup>	1.5	2 X 10 <sup>-4</sup>	60	
PEPSIN	1.4 X 10 <sup>-1</sup>	1.5	5 X 10 <sup>-3</sup>	60	

Table 5.9 – Comparison of enzyme limits of detection between two different methods using similar protease substrates

Comparing the limits of detection for the work carried out in our lab with those of Jones *et al.*, it can be seen that in each of the cases apart from pepsin the limit of detection has been significantly improved. The time taken to carry out the assays has also been vastly improved. The limits of detection that have been achieved in our lab could also be improved further. A sub-optimal wavelength of 605 nm was used to excite the BODIPY: $\alpha$ -casein substrate. With the use of laser induced fluorescence and cut-off filters an optimal excitation wavelength of 625 nm could be used, which would improve the sensitivity of the system giving improved limits of detection.

# 5.6 PROTEASE INHIBITION ASSAYS

### 5.6.1 Procedures

The effect of a variety of reversible inhibitors on the proteolytic enzymes was investigated. Enzyme – inhibitor pairs were chosen from the enzymes already studied and a selection of inhibitors consisting of 4-hydroxycoumarin, warfarin, 3-nitrophenylboronic acid, aprotinin, leupeptin, and pepstatin A. A description of the inhibitors used during this work is given in section 1.9.2.

Protease inhibition assays were carried out using the set-up that was shown in figures 5.19 and 5.20. Enzyme and inhibitor were pre-mixed before injection and then mixed with substrate in the incubation loop.

#### 5.6.2 Results

Preliminary experiments were carried out by incubating each proteolytic enzyme with increasing amounts of each inhibitor. It was found that neither 4-hydroxycoumarin nor warfarin had an inhibitory effect upon any of the proteolytic enzymes tested. However, it was found that the remaining inhibitors – 3-nitrophenylboronic acid (3-NPBA), aprotinin, leupeptin, and pepstatin A – each inhibited the four proteolytic enzymes to varying degrees. Enzyme – inhibitor pairs were selected depending on which inhibitor exerted the best inhibitory effect upon which enzyme.

The following enzyme – inhibitor pairs were used for all subsequent experiments

- 3-nitrophenylboronic acid and proteinase K
- aprotinin and α-chymotrypsin
- leupeptin and alkaline protease
- pepstatin A and pepsin

In addition to the enzyme assay parameters previously optimised (table 5.6), additional inhibition assay conditions were optimised using univariate methods. All other conditions such as pH, buffer system, and temperature, were as those conditions previously optimised.

The effect of the following parameters on the inhibition assays were optimised:

- pre-incubation time
- incubation time
- [substrate]
- [enzyme]

The assay parameters studied and the optimum values found for each enzyme – inhibitor pair are given below in table 5.10.

	PRE-INCUBATION TIME (mins)	INCUBATION TIME (minutes)	[SUBSTRATE] (ng ml <sup>-1</sup> )	[ENZYME] (µg ml <sup>-1</sup> )
3-NPBA + PROTEINASE K				
RANGE STUDIED OPTIMUM	0 – 5 2	1-5 2	53.3 – 266.4 106.6	0 – 1.866 0.746
APROTININ + α-CHYMOTRPYSIN				
RANGE STUDIED OPTIMUM	0-4 2	1 – 5 1	53.3 – 266.4 106.6	0 – 1.119 0.373
LEUPEPTIN + ALKALINE PROTEASE RANGE STUDIED OPTIMUM	0-5 4	1-5 1	53.3 – 266.4 213.1	0 – 0.233 0.093
PEPSTATIN A + PEPSIN BANGE STUDIED	0-4	1-5	53 3 - 266 4	0 - 37.31
OPTIMUM	2	1	106.6	0.373

Table 5.10 - Assay parameters studied and optimum conditions for various enzyme – inhibitor pairs using the fluorogenic substrate BODIPY:α-casein in a flow system

The optimisation of inhibition parameters will be discussed only for pepsin and pepstatin A. Optimisation experiments were carried out in duplicate and error bars represent plus and minus one standard deviation.

#### 5.6.2.1 Effect of pre-incubation time

Enzyme and inhibitor were pre-mixed in a glass vial and incubated for a set amount of time before injection into the flowing system and incubation with substrate. The effect of varying this pre-incubation time was investigated over 0 to 4 minutes in 1 minute increments. The results of this are shown in figure 5.33.



Figure 5.33 – Influence of pre-incubation time on the inhibition of pepsin by pepstatin A

Pre-incubation time (minutes)

The effect of pre-incubation time on the inhibition of pepsin by pepstatin A wasn't significant. This was expected as pepstatin A has been reported as an uncompetitive inhibitor of pepsin [281] and as described in section 1.9.1.1, uncompetitive inhibitors bind only with the ES complex and not with the free enzyme. Increasing the time spent for the enzyme and inhibitor together will therefore not increase inhibition. A pre-incubation time of 2 minutes was therefore chosen for further studies.

# 5.6.2.2 Effect of incubation time

Chapter 5

Incubation time is defined as the time spent for the substrate, enzyme, and inhibitor together in the incubation loop in the thermostatted water bath after injection of each into the flowing system. This assay parameter was investigated over the range 1 to 5 minutes in 1 minute increments. Figure 5.34 shows the effect of increasing incubation time on the amount of enzyme inhibition seen.





From figure 5.34 it is evident that the inhibition of pepsin by pepstatin A was not affected by incubation time. As increases in the incubation time of the substrate and inhibitor together did not cause an increase in the percentage inhibition, this suggests that the binding of inhibitor to the enzyme – substrate complex proceeds instantaneously. An incubation time of 1 minute was thus selected for further work so as to keep the assay time to a minimum.

#### 5.6.2.3 Effect of substrate concentration

Chapter 5

Increasing concentrations of substrate were injected into the flowing system with enzyme and inhibitor to determine the effect of increasing the substrate concentration on the inhibition of pepsin by pepstatin A. Figure 5.35 shows the results of this experiment.



The effect of increasing the substrate concentration was to increase the inhibition of pepsin by pepstatin A (figure 5.35). As pepstatin A is an uncompetitive inhibitor of pepsin, the inhibitor binds only with the enzyme – substrate complex, not with free enzyme. In this case therefore, as described in section 1.9.1.1 increased substrate concentration increases the inhibition. A high concentration of substrate will not reverse the inhibition, as is the case with reversible competitive inhibition. A substrate concentration of 106.6 ng ml<sup>-1</sup> was used for further work.

Chapter 5 Experimental: single labelled substrate protease and protease inhibition assays

### 5.6.2.4 Effect of enzyme concentration

The effect of increasing the concentration of pepsin was determined by injecting a series of increasing enzyme concentrations into the flowing system and detecting the amount of fluorescence (figure 5.36).





Concentration of pepsin (µg / ml)

As shown in figure 5.36, as the concentration of pepsin was increased the percentage inhibition also increased, although not to a great extent. Since the inhibitor only binds with the enzyme – substrate complex and not with free enzyme the effect will be minimal as increasing the enzyme concentration would only cause an increase in the inhibition through the formation of more enzyme – substrate, and not through the increased free enzyme. A concentration of 18.66  $\mu$ g / ml was used for further work as after this concentration increase in percentage inhibition is minimal.

# 5.6.2.5 Problems with inhibitors

A number of delays were introduced into this work due to problems encountered with the inhibitors.

Leupeptin hemisulfate salt was very problematic with respect to its stability. In powder form leupeptin is stable for 2 years if stored frozen and kept very dry. However, a 10 mM aqueous solution is stable for a week at 4 °C and at working concentrations (10 to 100  $\mu$ M) a solution is stable only for a few hours and the stock solution should be stored on ice for intermitted use over several hours.

Leupeptin was purchased in 5 mg vials, which is equivalent to 2.1 mM. A stock solution was made up by adding 1 ml 10 mM carbonate buffer pH 9.5 to the vial to give a concentration of 5 mg / ml. This was to prevent loss of leupeptin that would occur if weighing out and to prevent inaccuracies due to weighing out of very small quantities. These solutions were therefore stable for just one week. All experiments involving leupeptin therefore needed to be carried out rapidly and this is reflected in the poorer accuracy found with the leupeptin data than with the other inhibitors. This accuracy was also affected by the fact that individual experiments needed to be carried out in a few hours, as the solutions were stable for just several hours when kept on ice.

Leupeptin has working concentrations of 10 to 100  $\mu$ M. As shown in figure 5.45, the leupeptin inhibition curve was carried out using concentrations ranging from 0 to 3.73  $\mu$ g / ml. This is equivalent to molar concentrations of 0 to 7.8  $\mu$ M. The inhibition curve therefore has been carried out at low inhibitor concentrations, which is reflected in the low percentage inhibition. However due to the instability of leupeptin, which therefore makes it an extremely difficult inhibitor to work with, it was felt that spending more money on higher concentrations was not valid as it

#### Chapter 5 Experimental: single labelled substrate protease and protease inhibition assays

was only really important to obtain the linear portion of the graph to determine the limit of detection.

A second source of difficulty was found with pepstatin A. This inhibitor has poor solubility and does not dissolve in aqueous solutions. Pepstatin A is soluble in both dimethyl sulphoxide (DMSO) and methanol. Experiments were carried out and it was found that in the presence of inhibitor, the fluorescence intensity increased, the opposite of what was expected. The fluorescence of the substrate was therefore compared in buffer with that in methanol. As shown in figure 5.37, methanol dramatically affected the fluorescence of BODIPY. With increasing methanol concentration a concomitant increase in BODIPY fluorescence was noted. This response was unexpectedly non-linear.





The two plots in figure 5.37 represent the fluorescence intensity at the emission wavelength of BODIPY: $\alpha$ -casein and also at the peak emission wavelength found. This was because it was also found that the addition of increasing amounts of methanol caused a small wavelength shift of the emission maximum towards lower wavelengths, as shown in figure 5.38.

Chapter 5





It has been noted in the past that methanol dramatically enhances serine protease activity under anhydrous conditions [292]. Certain solvents or 'additives' including dimethylsulfoxide, formamide, and ethylene glycol have also been reported in the past to increase enzyme activity when added into the organic media along with small amounts of water [293, 294].

In order to determine whether methanol affects the fluorophore only or whether the enzyme is also affected by the presence of methanol, methanol concentration was increased in the presence of substrate and enzyme (figure 5.39).



# Figure 5.39 – Influence of percentage methanol on the fluorescence of BODIPY:α-casein and pepsin

If the shapes of the two graphs shown in figures 5.37 and 5.39 are compared it can be seen that the shapes are extremely similar. This suggests that the effect of methanol is purely on the BODIPY: $\alpha$ -casein substrate and not on pepsin, as if pepsin were also affected, the shape of the graph would have altered when compared to that of figure 5.37.

The solvent dependent effect that has been observed with BODIPY has been documented in the past by a number of researchers for various fluorophores [295, 296]. In fact, the Molecular Probes Inc. website records many of the BODIPY spectra in methanol as it greatly enhances the fluorescence [297].

The effect of incubating BODIPY: $\alpha$ -casein and pepsin over time in the presence of a constant concentration of methanol was also investigated. As shown in figure 5.40, the presence of methanol has a constant effect over time demonstrated by the similar shapes of the curves representing the presence of methanol and the absence of methanol (buffer).





As DMSO is also a solvent for pepstatin A, the effect of inclusion of DMSO in the medium on the fluorescence intensity and emission wavelength of BODIPY:α-casein was also studied.

From figure 5.41 it can be deduced that DMSO also causes an enhancement of the fluorescence of BODIPY, although at lower percentage concentrations the effect is not as dramatic as that seen with methanol. However, the effect of DMSO on the optimum emission wavelength was opposite to that caused by methanol as shown in figure 5.42. Whilst methanol caused a blue shift of around 5 nm at 99.6 %, DMSO caused a red shift of about 7 nm at 99.6 %.









As DMSO is more toxic than methanol and as pepstatin A was found to dissolve more easily in methanol than DMSO, the decision was taken to use methanol as the solvent for DMSO even though the fluorescence enhancement effects were greater with methanol than with DMSO.

A 10 mg / ml stock solution of pepstatin A in methanol was stored in aliquots in the freezer where they were stable for months. An aliquot was used for each experiment and was made up in 20 mM glycine buffer pH 10.0, which was stable for at least one day at room temperature. Accounting for dilution in buffer and for additional dilution in the flow system, the actual percentage of methanol present during optimisation experiments amounted to only 0.09 %. To take the effect of this methanol upon BODIPY:α-casein into account, the same percentage of methanol was also included in the blank reading. For the generation of the pepstatin A inhibition curve, the amount of methanol in the series of standard solutions of pepstatin A injected into the flowing system was kept constant.

# 5.6.2.6 Generation of inhibition curves

Under the optimum conditions already described, a series of inhibitor standard solutions were mixed with enzyme and injected into the flow analysis system to be incubated with substrate. The reaction products were detected and the fluorescence measured. Measurements were taken in triplicate and error bars represent plus and minus one standard deviation. Inhibition curves were obtained for each of the four enzyme – inhibitor pairs.





Concentration of 3-nitrophenylboronic acid (µg / ml)





Figure 5.45 - Inhibition of alkaline protease by leupeptin





Figures 5.43 to 5.46 show the inhibition of the enzymes proteinase K,  $\alpha$ the inhibitors 3chymotrypsin, alkaline protease. and pepsin by nitrophenylboronic acid, aprotinin, leupeptin, and pepstatin A respectively. As inhibitor concentration was increased there was an increase in percentage inhibition until a plateau point was reached. For the competitive inhibitors (3nitrophenylboronic acid, aprotinin, and leupeptin), as inhibitor concentration is increased there is more competition against the substrate for the active site of the enzyme and so percentage inhibition increases. At the plateau point of the inhibition curves, enzyme becomes limiting and further increases in inhibitor produce no increases in inhibition because there are no more enzyme sites to compete for. In the case of pepstatin A, which is an uncompetitive inhibitor, increases in inhibitor concentration cause more inhibitor to bind with the enzyme - substrate complex, preventing the formation of product. This is reflected by an increase in the percentage inhibition. However, at the plateau point the formation of the enzyme - substrate complex becomes limiting and further increases in

inhibitor concentration produce no further increases in inhibition because there is no more enzyme – substrate complex to inhibit.

For the competitive inhibitors at any given inhibitor concentration, if the amount of substrate is increased a drop in percentage inhibition would be seen as there would be more substrate to compete with the inhibitor for the enzyme active site.

#### 5.6.2.7 Determination of limits of detection

Limits of detection, defined as the analyte concentration giving a signal equal to the blank signal plus 3 standard deviations of the blank [291], were determined for each of the four inhibitors in the flow analysis system. Percentage inhibition against inhibitor concentration was plotted for the linear portion of the graphs for each of the inhibitors. These results are shown in figures 5.47 to 5.50. Measurements were taken in triplicate and error bars are given which represent plus and minus one standard deviation.









Figure 5.49 – Percentage inhibition against inhibitor concentration in linear portion to calculate limit of detection





Calculated limits of detection for each inhibitor using the substrate BODIPY: $\alpha$ -casein are given in table 5.11.

INHIBITOR	LIMIT OF DETECTION		
3-NITROPHENYLBORONIC ACID	2.504 μM		
APROTININ	0.161 nM		
LEUPEPTIN	0.140 μM		
PEPSTATIN A	7.20 nM		

Table 5.11 - Limits of detection determined experimentally for various protease inhibitors using the substrate BODIPY:α-casein in a flowing system

Limits of detection in the  $\mu$ M (3-nitrophenylboronic acid and leupeptin) and nM (aprotinin and pepstatin A) regions were achieved using the substrate BODIPY: $\alpha$ -casein, showing that this fluorogenic substrate can successfully be applied to the sensitive determination of protease inhibitors.

From graphs 5.47 to 5.50 it is obvious that both aprotinin and pepstatin A are both potent serine and aspartate protease inhibitors respectively. This was seen by the high percentage inhibition achieved at low inhibitor concentrations. This was reflected in the nM limits of detection. Conversely, 3-nitrophenylboronic acid and leupeptin were less efficient serine protease inhibitors giving much lower percentage inhibitions at higher inhibitor concentrations. This was again reflected in the limits of detection, which were obtained in the  $\mu$ M region.

The time taken to carry out these inhibition assays was rapid taking at most 5 minutes for leupeptin and again the flow-based procedure that has been described would be easily automatable for use in high throughput screening assays.

The method that has previously been described in section 1.8 3 2 2 by Jones *et al.* [255] was also applied to the determination of aprotinin. This work was able to achieve limits of detection in the  $\mu$ g / ml range using an assay that took 80 minutes. The method that has been described in this thesis has improved on this significantly. Aprotinin has been detected at a concentration of 1.1 ng / ml in just 3 minutes.

# 5.7 DISCUSSION

Calculations of dye protein ratios in fluorescent or fluorogenic conjugates usually assume that the absorbance characteristics of the conjugated dye are the same as those of the free dye. This was clearly not the case at wavelengths at ca. 600 nm in the present work. Experiments were carried out to establish whether the shape change that was observed between the absorbance spectra of the free and conjugated dye was due to the formation of covalent H-dimers between BODIPY molecules. The results of these experiments strongly suggested that this theory was indeed plausible. D:P calculations were therefore carried out using absorbance values at 280 nm and 350 nm to estimate protein and dye concentrations respectively. The latter wavelength was chosen because the absorbance spectra of the free dye and conjugated dye in this region are very similar. Using this method of calculation, it was found that a D:P ratio of 2.35 BODIPY molecules to 1  $\alpha$ -casein was optimal for use in these types of assays. This was in general agreement with the findings of Jones et al., [255] who showed that conjugation of an average of 4.5 to 6 BODIPY FL or even fewer BODIPY TR-X dye groups to each casein molecule was optimal. Presumably, the formation of non-fluorescent dimers within the substrate would further help to produce a quenching effect due to the fact that the dimers can absorb the exciting light but will not fluoresce. Larger D:P ratios could be achieved by increasing the levels of dye in the reaction mixtures, but this resulted in an increase in background fluorescence, i.e. in the intensity of the unhydrolysed substrate, limiting the sensitivity of the assays.

The relatively low molar ratio of dye to protein used in the present work provides an ideal substrate for proteolytic enzymes. Worries that the use of highly derivatised molecules might sterically hinder the activity of some proteases are minimised. In addition, BODIPY dyes have no overall charge, so dye conjugation
#### Chapter 5 Experimental single labelled substrate protease and protease inhibition assays

should not greatly affect the protein substrate conformation and enzyme-active site interactions.

The application of BODIPY: $\alpha$ -casein to the determination of different proteolytic enzymes with varying pH optima has been demonstrated. Enzyme assays were carried out in just 1.5 minutes with the exception of alkaline protease (2 minutes). The pH insensitivity of BODIPY means that assays can be carried out in a single step without the need to carry out the proteolytic reaction at one pH and the measurement at another pH, as would be necessary sometimes in the case of fluorescein. The procedure does not require any process such as separation, precipitation or centrifugation.

The suitability of BODIPY: $\alpha$ -casein as a substrate in enzyme inhibition assays has also been investigated. Limits of detection of 2.50 µM, 0.14 µM, and 7.20 nM were obtained for 3-nitrophenylboronic acid, leupeptin, and pepstatin A respectively. The inhibitor aprotinin has been found to be of benefit during paediatric cardiac surgery to reduce blood loss with cardiopulmonary bypass [274]. The limit of detection for this inhibitor was 0.161 nM (1.1 ng ml<sup>-1</sup>) using the current setup. This is extremely sensitive compared to the assay using a similar substrate described by Jones *et al* [255] who observed inhibition in the µg ml<sup>-1</sup> range. In addition, the assay described by Jones *et al.* took 1 hour 20 minutes to carry out, whereas the assay described here was carried out in just 3 minutes.

The fluorogenic assays that have been described here are rapid, taking at most 2 minutes for the enzyme assays and 5 minutes for the inhibition assays. They combine the sensitivity of long-wavelength fluorescence with the ease of automation that can be applied to flow injection analysis and could be incorporated into a simple portable fluorescence instrument. They have many clinical applications, since proteases are involved in virtually all biological functions and dysfunctions and therefore are promising as drug targets.

The application of BODIPY: $\alpha$ -casein in a flow injection analysis system is a rapid, simple and sensitive method for the assay of proteases, essential requirements for a routine assay. The whole assay is easily automated and would be well suited for use in high throughput screening.

## **CHAPTER 6**

## Experimental: multi-determination protease inhibition assays

#### 6.1 INTRODUCTION

The aim of these experiments was to apply those protease inhibition assays that had previously been optimised in chapter 5 to a flow injection analysis system such that the system would allow the determination of more than one inhibitor in the same assay i.e. multi-determination.

The need to improve productivity is imperative for drug discovery companies. One strategy is: test more discover more. The availability therefore of a simple, sensitive, and efficient method for the multi-determination of proteolytic enzymes and their inhibitors would be of much use in the high throughput screening of candidate drug molecules. This set-up would allow a much higher throughput accompanied by cost advantages.

An excellent review discusses the determination of two or more species in the same sample with a single flow injection analysis system [71]

#### 6.2 MULTI-DETERMINATION PROTEASE INHIBITION ASSAYS USING CONTROLLED-PORE GLASS AS A SUPPORT

#### 6.2.1 Procedures

Fluorescence measurements were made using an F-4500 fluorescence spectrophotometer with excitation and emission bandwidths of 10nm. Time scans were performed with excitation and emission wavelengths of 605 nm and 644 nm respectively for BODIPY and BODIPY: $\alpha$ -casein, 280 nm and 350 nm for  $\alpha$ -casein, 597 nm and 660 nm for naphthofluorescein, and 640 nm and 664 nm for Cy5.

As a starting point for these assays, it was decided that the simplest approach would be a dual protease inhibition assay using one buffer system. As it was shown that both alkaline protease and proteinase K operated at pH 9.5 (section 5.5 2, table 5.6), these two enzymes seemed like the obvious choice. The serine protease inhibitor 3-nitrophenylboronic acid was shown to inhibit proteinase K (section 5.6.2.6, graph 5 43) and so this inhibitor was selected for use in this dual system. The ideal set-up for a simple assay was to flow buffer slowly through two enzyme reactors, each containing either immobilised alkaline protease or immobilised proteinase K, as shown in figure 6.1. A substrate / inhibitor solution would be injected and the fluorescence detected through the time-dependent separation of the peaks emerging from each of the reactors by having different lengths of flow tubing between the enzyme reactors and the detector.

The other alternative to this set-up was to capture the assay components in each column using an incubation loop and 4-port manual switching valve. However, this approach seemed more problematic due to having two switching valves to

operate at exact times, and so the above approach depicted in figure 6.1 was chosen as the proposed system.





(A = buffer, B = peristaltic pump, C = substrate or substrate / inhibitor injection valve, D = enzyme reactor 1, E = enzyme reactor 2, F = thermostatted water bath, G = fluorescence detector, H = waste)

Temperature, pH and buffer system conditions for alkaline protease had previously been optimised (section 5.5.2, table 5.6) to be 10 mM carbonate buffer pH 9.5 and 40 °C. Those for proteinase K were 20 mM carbonate buffer pH 9.5, 35 °C. By comparing results for buffer concentration and temperature for each enzyme, it was decided that the best compromise for multi-determination assay conditions was 20 mM carbonate pH 9.5, 35 °C. These conditions were used in all experiments.

The sequence of events using this flow analysis set-up is shown in figure 6.2.

# Substrate only or substrate and inhibitor injected through a manual injection valve (25 µl each) and split at a T-junction to flow through two enzyme reactors Products from the enzyme reactor with the shorter distance to the F-4500 detected first Products from the enzyme reactor with the longer distance to the F-4500 detected second

#### Figure 6.2 – Sequence of events for multi-determination protease inhibition assays using immobilised enzyme reactors

Controlled-pore glass (CPG) was purchased from Cambio Ltd. and evaluated as a support for use in these assays. Immobilisation of alkaline protease and proteinase K onto controlled pore glass was carried out as previously described by Masoom and Townshend [120], a description of which is given in section 2.2.12. 2.5, 5.0, and 7.5 mg alkaline protease and 4.0, 8.0, and 12.0 mg proteinase K were used for immobilisation onto 0.5 g CPG. Immobilised enzyme was stored in buffer at 4 °C when not in use. The reaction scheme for enzyme immobilisation is given in figure 6.3.

Figure 6.3 – Reaction scheme for the immobilisation of enzyme onto controlled pore glass



CPG was chosen as a support due to its high surface area, high mechanical strength, high flow rate, and rigid glass structure. Initial studies were carried out on aminopropyl CPG with a mean pore diameter of 182 Å, a particle size of 37 – 74  $\mu$ m, and a capacity of 701.9  $\mu$ moles amine per g glass. CPG with a mean pore diameter of 972 Å, a particle size of 37 – 74  $\mu$ m, and a capacity of 67  $\mu$ moles amine per g glass.

A picture of the 182 Å aminopropyl CPG was taken using a Scanning Electron Microscope (SEM) as shown in figure 6.4.

### Figure 6.4 - SEM pictures showing (a) 182 Å aminopropyl glass and (b) an individual glass chip



It can be seen that CPG has the appearance of fine glass chips of irregular shape. The best resolution for the SEM is at distances of  $100 - 200 \ \mu m$  and so the pores of the CPG were beyond the capability of the SEM.

Immobilised enzyme was packed into Omnifit glass columns with a bore of 3 mm, and a length of either 2.5 cm or 5.0 cm corresponding to reactor volumes of 175  $\mu$ l and 350  $\mu$ l respectively. Aluminium frits with 2  $\mu$ m pores were used at both ends to prevent leakage of the CPG. The appearance of the enzyme reactor 5 cm in length is shown in figure 6.5. Columns were packed by mounting the columns vertically, and connecting the bottom of the column to a peristaltic pump that allowed suction of loaded CPG through the top end of the column downwards. The glass column was then incorporated into the flowing system and clamped vertically in the water bath. Buffer was flowed upwards through the column, which was opposite to the direction of packing. This ensured maximum contact of the buffer with the CPG and tighter packing.



#### Figure 6.5 – Appearance of the enzyme reactors: top – packed glass column, bottom – empty glass column

#### 6.2.2 Results

Injections of BODIPY: $\alpha$ -casein and naphthofluorescein were made through a column packed with 12.0 mg proteinase K - CPG. A high concentration of BODIPY: $\alpha$ -casein was used to give sufficient fluorescence intensity, and naphthofluorescein was also used as it is extremely fluorescent and therefore easy to detect whilst being cheap. When each of these was injected, nothing was detected i.e. these dyes appeared to be retained on the column.

As a control experiment therefore, BODIPY: $\alpha$ -casein was mixed in a standard 3 ml fluorescence cuvette with a random amount of proteinase K – CPG by inverting the cuvette five times. The cuvette was incubated at 35 °C and at time intervals the liquid was decanted off and the fluorescence measured. The liquid was then again mixed with the beads and returned to the water bath for a further time before the liquid was again decanted off and the fluorescence measured. This was repeated over time.

As shown by figure 6.6, it was observed that over time the fluorescence of BODIPY: $\alpha$ -casein decreased in the presence of the proteinase K – CPG when compared to the fluorescence of BODIPY: $\alpha$ -casein in the absence of CPG.





Interestingly it was found that incubation of BODIPY: $\alpha$ -casein for 2 minutes with proteinase K – CPG produced an enhancement of fluorescence when compared to BODIPY: $\alpha$ -casein incubated with CPG with no proteinase K immobilised (figure 6.7). This suggested that proteinase K was definitely successfully immobilised onto the CPG as the fluorescence enhancement indicated that the substrate was subjected to proteolytic cleavage.



Conversely it was found that when naphthofluorescein was incubated with the proteinase K - CPG the fluorescence intensity remained relatively constant over time. (It is shown later in section 6.2.2.4 that naphthofluorescein does not appear out of the column either. However, this is due to column length, not due to sticking to the CPG, which explains the fact that when naphthofluorescein was incubated with proteinase K - CPG the fluorescence intensity remained constant).

The column was re-packed with 7.5 mg alkaline protease – CPG to see whether a change in the immobilised enzyme would alter the flow of BODIPY: $\alpha$ -casein and naphthofluorescein through the column. However, both were also retained in this column.

Injections of BODIPY: $\alpha$ -casein were then made through a column containing raw aminopropyl CPG i.e. without activation or immobilisation of enzyme. From the results shown in figure 6.8, it was observed that some BODIPY: $\alpha$ -casein was able to flow through this column, although most was retained.





On the basis of the results that had been found it was decided that a number of possibilities could account for the fact that the enzyme reactor retained both BODIPY:α-casein and naphthofluorescein.

It was seen that BODIPY: $\alpha$ -casein was stable with respect to its fluorescence intensity over time whereas in the presence of proteinase K – CPG the fluorescence dropped. This could be because the stability of the product was different to that of the substrate, perhaps due to the fluorophore molecules being more exposed in the product. It could also be explained by sticking of the substrate or product to the proteinase K – CPG chips, or by the CPG somehow quenching the fluorescence of BODIPY: $\alpha$ -casein.

Another possibility was product inhibition. Product binds to the active site by the same bonds that bind the substrate. As a result, enzyme-product complexes are formed which prevent the substrate from binding the enzyme. As both substrate and product compete for the active site and both cannot bind simultaneously, this is a type of competitive inhibition.

One explanation is that substrate was cleaved into peptides and so enhancement of fluorescence was seen. However, with increasing time these peptides may have been attracted to each other so that the fluorophore molecules again came into close contact and the internal energy transfer effects were again seen.

One other potential reason was that the pore sizes were too small and therefore became blocked. The average pore size was 182 Å and if it is assumed that an enzyme has a rough size of 60 Å, then actual space for the substrate to flow through becomes limited.

A combination of any of these possibilities may also have been true.

## 6.2.2.1 Absorbance experiments to investigate sticking of α-casein to glutaraldehyde activated CPG

An experiment was conducted to determine whether  $\alpha$ -casein would stick to glutaraldehyde-activated CPG. Absorbance measurements were made at 278 nm using an Unicam 8700 UV / Visible Spectrometer. A standard 3 ml quartz absorbance cuvette was used and baselines were taken with buffer prior to measurements. A 0.1 %  $\alpha$ -casein solution was made up in 0.1 M phosphate buffer pH 6.0 and a random amount of glutaraldehyde-activated beads were then added to this solution in a centrifuge tube, the mixture agitated, and then spun down using a Clandon MLW T52.1 centrifuge for 2 minutes. The liquid was then decanted off and an absorbance reading was taken. A fresh solution of 0.1 M phosphate buffer pH 6.0 was then added to the beads and the method repeated. Repetitions were made until protein was no longer detected in the buffer. The results are given in figure 6.9.



#### Figure 6.9 – Absorbance of 0.1 % α-casein solution before (1) and after (2) mixing with glutaraldehyde-activated CPG and subsequent readings of freshly added buffer

Comparing samples 1 and 2, it appeared that 11.28 % of the  $\alpha$ -casein was lost. However, when fresh buffer was added and then the absorbance measured, traces of  $\alpha$ -casein were detected, indicating that the  $\alpha$ -casein was getting caught up in the CPG chips. After fresh buffer was added until  $\alpha$ -casein was no longer detected, a loss of 2.85 % was found which could be attributed to experimental error. This experiment therefore showed that  $\alpha$ -casein was not sticking to the CPG.

#### 6.2.2.2 Blocking of free aldehyde groups using ethanolamine

Free aldehyde groups on the proteinase K – CPG without attached enzyme were blocked using ethanolamine as described in section 2.2.11. This was carried out to make sure all aldehyde groups were blocked so that substrate could not bind to any free groups. Ethanolamine was used as an alternative to Bovine Serum Albumin (BSA) or  $\alpha$ -casein that commonly are used for blocking because these proteins would be used as a substrate by the immobilised enzyme. Tris (hydroxymethyl)amino methane (TRIS) could also have been used, but this is more bulky and so 3.0 M ethanolamine pH 9.0 was used. Sodium cyanoborohydride (NaCNBH<sub>3</sub>) at a concentration of 10 mM was used to make the linkage permanent.

It was found, however, that blocking had little effect and BODIPY:α-casein was still retained in the enzyme reactor.

#### 6.2.2.3 Evaluation of CPG with a larger pore size of 972 Å

CPG with a larger mean pore diameter was evaluated as a support to investigate whether pore size was limiting the flow of BODIY: $\alpha$ -casein through the enzyme column. Using a larger pore size it was found that whilst 2.95 % naphthofluorescein was detected, which was still very poor, both BODIPY and BODIPY: $\alpha$ -casein were still retained within the column. This suggested that something other than pore size was responsible, although the result with naphthofluorescein was a slight improvement.

#### 6.2.2.4 Evaluation of shorter enzyme reactors

The 5 cm long glass columns that had been used up to this point were replaced with 2.5 cm long columns to see whether this would affect the amount of material that was retained. Columns were packed with blocked 8 mg proteinase K – 972 Å CPG and injections of the dyes naphthofluorescein, Cy5, and BODIPY were made as well as injections of BODIPY: $\alpha$ -casein and  $\alpha$ -casein.

As shown in figures 6.10 to 6.14, the use of a shorter column made a noticeable difference. When naphthofluorescein, Cy 5, and  $\alpha$ -casein were injected, all were detected. When BODIPY and BODIPY: $\alpha$ -casein were injected, nothing was detected.





Figure 6.11 – Injection of Cy5 with and without 2.5 cm column





Figure 6.12 – Injection of  $\alpha$ -casein with and without 2.5 cm column







Figure 6.14 – Injection of BODIPY:α-casein with and without 2.5 cm column

These results suggested that for some reason the BODIPY fluorophore was sticking to the CPG retaining both the fluorophore and the conjugate, since  $\alpha$ -casein flowed through freely. BODIPY, as described in section 1.1.4.3, is relatively non-polar whereas both naphthofluorescein and Cy 5 are polar molecules. It is therefore possible that the polarity of the molecule influences its ability to flow through the CPG. One possibility is that polar molecules dissolve in the buffer more easily and are pulled through the column in solution. However, BODIPY with no ionic charge may stick to the CPG instead of being pulled along by the buffer.

The use of CPG as a support in these assays has proved to be incompatible with the BODIPY fluorophore. CPG was therefore not pursued as a support for immobilisation.

#### 6.3 MULTI-DETERMINATION PROTEASE INHIBITION ASSAYS USING PIERCE ULTRALINK MEDIUM AS A SUPPORT

#### 6.3.1 Procedures

UltraLink<sup>™</sup> Biosupport Medium, 0.25 g (approximately 2 ml of gel), was purchased from Pierce Chemical Company to investigate whether this medium could be used as an alternative to controlled-pore glass for immobilisation of proteolytic enzymes and incorporation into a glass column. The immobilisation of enzyme onto the UltraLink Medium was carried out according to the manufacturer's instructions and is described in section 2.2.13. An 8 mg / ml solution of proteinase K (lot 90K8604, 11.3 Units / mg) was used for immobilisation onto 0.1254 g UltraLink Medium. A quench solution of 3.0 M ethanolamine pH 9.0 was used to block unreacted azlactone sites.

Pierce UltraLink Biosupport Medium is a preactivated, ready-to-use bead with a bis-acrylamide / azlactone copolymer composition. The copolymer is slightly hydrophobic and highly crosslinked and because the azlactone functionality is copolymerised with the matrix material, the binding capacity is an integral part of the bead. This results in a high functionality throughout the porous bead matrix. The support rapidly reacts with nucleophiles, allowing coupling to a wide range of proteins. The beads have an average particle size of  $50 - 80 \mu m$  in diameter with a pore size of 1000 Å and a functionality of  $250 \mu moles$  per g of beads. The matrix is stable over a pH range of 1 - 13 and due to the rigid polymeric nature of the support, it has excellent utility in medium to low pressure applications. This support was thus chosen for these attractive features and the ease of use. The reaction scheme for enzyme immobilisation is given below in figure 6.15.





The amount of proteinase K actually immobilised onto the Biosupport Medium was determined by measuring the amount of proteinase K left in the filtrate using the Bicinchoninic Acid (BCA) protein assay, as described in section 2.2.14. The use of Triton® X-100 surfactant in the bead production could have interfered with an assay using absorbance at 280 nm for the measurement of uncoupled enzyme. Hence, the BCA protein assay was used. By subtracting the amount of proteinase K left in the filtrate from the total amount of proteinase K added, the amount of proteinase K actually immobilised onto the support could be deduced.

Evaluation of the proteinase K immobilised onto UltraLink was carried out in a flow injection analysis system similar to that depicted in figure 6.1, but using a single stream with one packed column. A flow rate of 1.06 ml / min was used with 20 mM carbonate pH 9.5 as the buffer system at room temperature. An Omnifit glass column 5 cm in length with a reactor volume of 350  $\mu$ l (figure 6.5) was used with 2  $\mu$ m aluminium frits. Columns were packed as described above in section 6.2.1.

All fluorescence measurements were made using an F-4500 fluorescence spectrophotometer with excitation and emission bandwidths of 10nm. Time scans were performed with excitation and emission wavelengths of 605 nm and 644 nm for BODIPY solution and BODIPY: $\alpha$ -casein conjugate, and 280 nm and 350 nm for  $\alpha$ -casein solution.

#### 6.3.2 Results

#### 6.3.2.1 Determination of amount of proteinase K immobilised onto UltraLink Biosuppport Medium

The BCA assay was used to calculate the amount of proteinase K left in the filtrate after the immobilisation procedure had been carried out. A set of standard proteinase K solutions of concentrations 0.25 mg / ml, 0.50 mg / ml, 1.00 mg / ml, and 2.00 mg / ml was used to generate a calibration graph. As shown in figure 6.16, the calibration graph showed good linearity over this concentration range with an  $R^2$  of 0.9900. The absorbance of the unknown proteinase K solution was 1.064 absorbance units. By substituting this absorbance into the equation of the calibration graph, the amount of proteinase K left in the filtrate was determined. Hence by deduction from the total amount of proteinase K used, the amount of proteinase K immobilised onto the UltraLink Biosupport Medium was deduced. This was 13.603 mg proteinase K, which corresponds to a coupling efficiency of 85 %.



## Figure 6.16 - Calibration graph of proteinase K concentration using

#### 6.3.2.2 Evaluation of UltraLink as a support for multi-determination protease inhibition assays in a flowing system

Injections of BODIPY, BODIPY: a-casein, and a-casein were made through the column containing proteinase K immobilised onto UltraLink in order to determine whether these materials would flow through the column. Injections were also made through an empty column as a blank to compare the results. As shown in figures 6.17 to 6.19, all three compounds were able to flow through the column freely.



Figure 6.17 – Injection of BODIPY through both an empty column and a column packed with proteinase K - UltraLink

Figure 6.18 – Injection of BODIPY:α-casein through both an empty column and a column packed with proteinase K - UltraLink







As peak broadening occurred, the peak areas were compared in order to calculate the percentage of fluorescent material that was flowing through the UltraLink Medium compared with that flowing through the empty column. These percentages are given in table 6.1.

MATERIAL	AMOUNT FLOWING THROUGH ULTRALINK COLUMN (%)
BODIPY	118
BODIPY:α-CASEIN	369
α-CASEIN	107

Table 6.1 – Amount of different materials flowing through a packed proteinase-UltraLink column compared with an empty column Experimental: multi-determination protease inhibition assays

Chapter 6

It is apparent from table 6.1 that BODIPY, BODIPY: $\alpha$ -casein, and  $\alpha$ -casein were all capable of flowing through the column containing proteinase K immobilised onto UltraLink Biosupport Medium. This was a huge improvement when compared to controlled-pore glass and these initial results with UltraLink were extremely promising. It also certainly confirms that a BODIPY-CPG interaction was responsible for the problems that were found when using CPG.

It can be seen that both BODIPY and  $\alpha$ -casein flow through the medium freely with all the material being detected. The percentage of material flowing through the packed column was derived by comparing the peak areas obtained through packed and empty columns. However, peak areas were calculated using the F-4500 software, which does not give an absolutely accurate value. This may account for the fact that the percentages for BODIPY and  $\alpha$ -casein are slightly above 100 % when in theory a maximum of 100 % should have been recorded.

The fact that 369 % BODIPY:α-casein appeared to flow through the packed column is very encouraging. This suggested that not only was BODIPY:α-casein able to flow through this support, unlike controlled-pore glass, but that the conjugate was being digested by the immobilised proteinase K into highly fluorescent peptides.

These preliminary results with UltraLink Biosupport Medium indicate that this support is more conducive than controlled pore glass for use with BODIPY:α-casein. No sticking of either dye or protein to UltraLink was observed.

Unfortunately, as time did not permit, these experiments were not continued and would therefore constitute further work. Nevertheless, on the basis of the preliminary results presented here, it seems logical to suggest that the application of this medium to the set-up shown in figure 6.1 would produce an effective multi-determination protease inhibition assay for high throughput screening.

#### 6.4 MULTI-DETERMINATION PROTEASE INHIBITION ASSAYS USING SOLUBLE ENZYME

As lack of time prevented further investigation into the use of UltraLink Medium in flow-based multi-determination protease inhibition assays, the use of a soluble enzyme system was very simply demonstrated in order to show the principle behind these multi-determination assays. On this kind of macro-system, the soluble method would not be preferred due to the high amounts of enzyme required and the level of complexity of the flow system when compared to that of the flow system incorporating immobilised enzymes.

#### 6.4.1 Procedures

All fluorescence measurements were made using an F-4500 fluorescence spectrophotometer with excitation and emission bandwidths of 10 nm. Time scans were performed with excitation and emission wavelengths of 605 nm and 644 nm respectively.

For these assays pepsin and proteinase K were chosen to represent two enzymes with pH extremes and from different classes: proteinase K is a serine protease working at pH 9.5, whereas pepsin is an aspartate protease working at 2.0. Pepstatin A was chosen as the model inhibitor as it is active over a wide pH range and so would work in either buffer system.

Enzyme solutions were made up by dissolving enzyme into 500 ml of the optimised buffer systems for each enzyme. 0.50 mg / ml pepsin was made up in 20 mM glycine pH 2.0, and 0.02 mg / ml proteinase K in 20 mM carbonate buffer pH 9.5. N<sub>2</sub> was then bubbled through each of the enzyme solutions to remove  $O_2$ . Assays were carried out at 35 °C using a thermostatted water bath.

The flow set-up that was used for these experiments is shown in figure 6.20 and the individual steps that were carried out are depicted in figure 6.21.

Figure 6.20 – Schematic of flow system used for soluble



(A = enzyme solution 1, B = enzyme solution 2, C = peristaltic pump, D = substrate or substrate / inhibitor injection valve 1, E = substrate or substrate / inhibitor injection valve 2 1, F = thermostatted water bath, G = T-junction, H = fluorescence detector, I = waste)





#### 6.4.2 Results

## 6.4.2.1 Determination of length of flow tubing required for separation of peaks

As the principle behind this multi-determination assay was to inject two lots of substrate with or without inhibitor and detect the two lots of product formed in each of the enzyme streams by time-dependent separation, it was important to determine the length of tubing required to cause such a separation. The pepsin enzyme solution was flowed through the short length of tubing whereas proteinase K was flowed through the longer length of tubing. The consequence of this was that the first detected peak was due to the pepsin component and the second peak was due to the proteinase K component. The tubing that was used for these experiments was Omnifit Teflon tubing with an internal diameter of 0.8 mm. As a minimum length of tubing was required for the shorter length pepsin section, a length of 25 cm was used between the injection valve connector and the T-junction. This was chosen so that the assay components would flow through the thermostatted water bath before detection and to cover the physical distance between the injection valve and T-junction. Using this length of tubing, the time between injection and detection was 9 seconds.

A longer length of tubing of 65 cm for the proteinase K part was firstly added. However, using this length a time of 16 seconds was observed between injection and detection, which was too short for peak separation. In order for a separation of the peaks to occur the first peak has to return to the baseline first, so sufficient time must be allowed for this to occur. A length of 355 cm was then added which equated to a time of 45 seconds between injection and detection. This length was enough to separate the two peaks, but the first peak had not quite reached the baseline properly. This length was therefore increased to 399 cm, which allowed a 50 second separation, and the two peaks were properly resolved as Experimental: multi-determination protease inhibition assays

demonstrated by figure 6.22. For further experiments therefore a shorter length of 25 cm and a longer length of 399 cm were used.



Figure 6.22 – Separation of two enzyme peaks using 25 cm and 399 cm length tubing

The length of tubing that was required for separation of the two peaks was not ideal. The length of 399 cm had to be coiled for placement in the water bath. This soluble set-up would be much more applicable to a miniaturised system where such long lengths would not be needed.

#### 6.4.2.2 Multi-determination protease inhibition assay

As time was very short the idea of this investigation was to demonstrate the principle, so the results that are presented here are of a qualitative nature rather than quantitative. Triplicate injections of substrate in the absence and presence of pepstatin A were made using the system shown in figure 6.20 with tubing lengths as described in section 6.4.2.1. The time scans showing an example of these injections are given in figure 6.23.



From figure 6.23 it can be seen that the effect of an inhibitor on two proteolytic enzymes has been determined in the same assay. From the pepsin peaks it is evident that inclusion of pepstatin A inhibited pepsin, signified by the reduction in fluorescence intensity. However, if the proteinase K peaks are examined, these are identical whether pepstatin A was present or not. This indicated that pepstatin A did not inhibit proteinase K. This was as expected as pepstatin A is an aspartate protease inhibitor.

Using the substrate BODIPY:α-casein it has been shown that multi-determination assays using the described system were possible and these assays have been used to show the selective inhibition of two proteolytic enzymes by pepstatin A. Not only two proteolytic enzymes, but two enzymes that operate at opposite pH values – pH 2.0 for pepsin and pH 9.5 for proteinase K. In addition these enzymes represented two different classes of protease – pepsin being an aspartate protease and proteinase K being a serine protease. This system would be suitable for the high throughput screening of compounds against more than one target enzyme.

#### 6.5 DISCUSSION

The suitability of applying the long-wavelength fluorogenic substrate BODIPY: a-case in to multiplexed protease and protease inhibition assays has The use of immobilised enzymes in these assays much been investigated simplifies the flow system needed, and due to the advantages of immobilised enzymes as discussed in section 1.7.1, it was decided that immobilisation of proteolytic enzymes to a suitable support was required. However, studies on the use of controlled-pore glass as a support were unsuccessful due to an incompatibility with the BODIPY fluorophore. It appeared that BODIPY, which has no ionic charge, stuck to the support thereby retaining the fluorophore within the enzyme reactor. One hypothesis on the cause of this sticking was based on polarity. It was found that polar dyes flowed through the CPG column freely, and it was thus postulated that the more polar the fluorophore, the less sticking would occur. This possibly may have been due to polar molecules dissolving into solution with more ease.

An alternative support was explored for immobilisation of enzymes and subsequent use with BODIPY' $\alpha$ -casein. Pierce UltraLink Biosupport Medium was chosen for the reasons discussed and was shown to be extremely promising for use in these assays. Enzyme was successfully immobilised onto this medium and UltraLink was shown to support the flow of BODIPY: $\alpha$ -casein. Initial results also suggested that as BODIPY: $\alpha$ -casein flowed through the enzyme reactor, the immobilised enzyme cleaved the substrate into fluorescent peptides. These results were extremely encouraging and promising and with more time this system could potentially be developed for multi-determination. This would constitute future work.
Chapter 6

Soluble enzyme experiments were carried out as a proof of principle determination and it was shown that BODIPY:α-casein could be used as a pH-independent substrate for use in flow-based multiplexed screening programmes. However, this format was less favourable than that using immobilised proteases, as the system was more complex. In addition, due to the high concentrations of enzyme needed and the long lengths of tubing that were required for peak separation, this soluble format would be much more suited to a miniaturised system. The application of these assays to a chip-based system would comprise future work and will therefore be discussed as such. However, further work on the effects of flow rate ought to be carefully studied. At a lower flow rate the dispersion of the zones will be less and they will overlap less, so it may be possible to have shorter tubes for the second enzyme. In that case it might be possible to get more samples per hour through even though the flow rate has been reduced.

## **CHAPTER 7**

# Conclusions and future work

#### 7.1 CONCLUSIONS

The objective of this investigation was to develop new fluorescence methods for the rapid and sensitive determination of proteolytic enzymes and their inhibitors. In addition, the possibilities of applying these enzyme and inhibition assays to a flow injection analysis system and the development of multi-determination assays were also to be examined. These assays were aimed at a high throughput screening application.

To a large extent these objectives have been successfully fulfilled. Fluorescence polarisation and energy transfer methods have both been explored for use in protease and protease inhibition assays. Of these two approaches, energy transfer methods have proven superior. As discussed in section 3.4, this is of no great surprise as inherent problems in polarisation methods include (a) polarisation filters reducing exciting and emitted light intensities; (b) polarised light pre-selecting only a fraction of the potentially excitable molecules; and (c) polarisation changes being small and temperature sensitive. It was found that for fluorescence polarisation assays a lightly labelled substrate was required, and these offer an advantage due to less steric hindrance of the proteolytic enzyme over those heavily labelled substrates used in energy transfer methods. These substrates using fluorescence polarisation have successfully been used to analyse different proteases with broad specificity and fluorescence polarisation has the additional benefits of simplicity and relative immunity to the inner filter effect.

However, these pros of fluorescence polarisation are outweighed by the fact that the fluorescence polarisation changes are very small which greatly limits the sensitivity of the assay. The pH sensitivity of fluorescein as a label was also extremely limiting as easy analysis of acid proteases was prevented. The use of

long wavelength fluorophores, which are usually preferred due to the decrease in background fluorescence found within the long wavelength region (ca. 600 – 900 nm), was also unsuccessful.

The usefulness of this technique was further limited by the requirement of a cuvette-based approach, with the associated low throughput and high reagent usage

Energy transfer methods have been investigated in two different formats (a) a single fluorophore used as an excess label attached to a protein substrate, with proteolytic action fragmenting it and producing enhanced fluorescence through abolition of internal energy transfer, and (b) use of two fluorophores attached to one substrate, with energy transfer from one to another and proteolytic cleavage causing a fluorescence enhancement through disruption of the energy transfer.

Double-labelled energy transfer methods have been investigated using the donor / acceptor pairs of FITC / TMRITC and Cy 5 / Cy 5.5 One major advantage that this method possesses over the single-labelled energy transfer methods is that intensity ratio measurements are made using double-labelled substrates, and so instrument fluctuations will largely cancel out. The use of two fluorophores should also produce a larger Stokes shift, helping to solve a real problem in long wavelength measurements.

For reasons discussed in chapter 4, it was found in the double-labelled section of work that lightly labelled donor and heavily labelled acceptor conjugates were required for use in FRET protease assays.

FITC / TMRITC, although promising in theory, were found not to be perfect. The phenomenon of energy transfer was shown to exist in these conjugates, however the results were extremely difficult to interpret and by no means ideal for use in protease assays. Initial results with Cy 5 and Cy 5.5 as donor and acceptor

respectively were unsuccessful as the two emission peaks were unable to be resolved. However, Cy 5 and Cy 5.5 have previously been shown to be a good donor / acceptor pair [287, 288].

It is obvious that the synthesis of double-labelled substrates is more complex both to investigate and to achieve in routine practice than single-labelled substrates, although with more work spent this energy transfer method may give advantages over other energy transfer methods.

The use of a single excess-labelled substrate has, by far, been the most successful fluorescence method for the determination of proteolytic enzymes and inhibitors. A substrate utilising  $\alpha$ -casein labelled with a long wavelength BODIPY dye has been synthesised. The relatively low molar ratio of dye to protein used in this work provides an ideal substrate for proteolytic enzymes as worries that the use of highly derivatised molecules might sterically hinder the activity of some proteases are minimised. In addition, BODIPY dyes have no overall charge, so dye conjugation should not greatly affect the protein substrate conformation and enzyme-active site interactions.

This substrate has been shown to be a suitable intramolecularly quenched conjugate for the determination of proteolytic enzymes and their inhibitors in flowbased screening programmes. The application of BODIPY: $\alpha$ -casein to the determination of different proteolytic enzymes with varying pH optima (ranging from pH 2.0 to 9.5) has been demonstrated. The suitability of BODIPY: $\alpha$ -casein as a substrate in enzyme inhibition assays has also been investigated. Limits of detection of 2.50  $\mu$ M, 0.14  $\mu$ M, 7.20 nM, and 0.161 nM were obtained for 3-nitrophenylboronic acid, leupeptin, pepstatin A, and aprotinin respectively. The assays are rapid, taking at most 2 minutes for the enzyme assay and 5 minutes for the inhibition assays. The flow-based procedure that has been described is particularly well suited to the high throughput screening of candidate drug molecules.

The progression of this work was to aim towards the adaptation of the flow-based procedure to a set-up that would allow the multi-determination of proteolytic enzymes and their inhibitors. This would further improve the throughput of the system – an essential requirement of high throughput screening. As time was short, only the feasibility of this set-up was explored. The use of immobilised enzymes was investigated, as this far simplified the set-up required. Two different supports were studied – controlled pore glass (CPG) and Pierce UltraLink Biosupport Medium. It was found that CPG was unsuitable for use with the BODIPY fluorophore, thereby preventing its use in these assays. BODIPY was found to stick to the CPG resulting in the fluorophore being retained within the enzyme column. This was possibly related to the ionic charge as polar dyes were found to flow freely through the column whereas the non-polar BODIPY molecule did not.

Pierce UltraLink Biosupport Medium was therefore investigated as an alternative to CPG and was shown to be extremely promising as a support in these assays. Enzyme was successfully immobilised onto this medium and UltraLink was shown to support the flow of BODIPY: $\alpha$ -casein. Initial results also suggested that as BODIPY: $\alpha$ -casein flowed through the enzyme reactor, the immobilised enzyme cleaved the substrate into fluorescent peptides. With more time this system could potentially be developed for multi-determination.

For proof of principle soluble enzyme experiments were carried out. However, this format was less favourable than that using immobilised proteases, as the system was more complex. In addition, due to the high concentrations of enzyme needed and the long lengths of tubing that were required for peak separation, this soluble format was much more suited to a miniaturised system.

#### 7.2 FUTURE WORK

#### 7.2.1 Fluorescence polarisation

Although fluorescence polarisation was found to be less sensitive than energy transfer methods, polarisation does possess several attractive features. The cuvette-based approach hampered the usefulness of this technique due to the associated low throughput and high reagent usage. A flow-based system would improve this throughput significantly. It would therefore be extremely useful to transfer the assay to a flow injection analysis system similar to that described for the quenched substrate, if only to test the sensitivity of polarisation measurements in systems where small volumes are illuminated and observed in the instrument.

Another area that would be worth pursuing is the use of BODIPY and other long wavelength fluorophores. Experiments using BODIPY:a-casein with fluorescence polarisation were unsuccessful. However, these substrates were highly quenched, causing significant depolarisation through non-radiative energy transfer and thus severely limiting the assay span and sensitivity. A study of much more lightly labelled substrates containing little or no quenching and their applicability to fluorescence polarisation may prove to be more promising. In addition assessment of other long wavelength fluorophores would be useful. For use in fluorescence polarisation the dye requires a suitable fluorescence lifetime for appreciable Brownian rotation to occur. There are worries about the extent of polarisation effects with short-lifetime long-wavelength fluorophore molecules (section 1.2.1.3). This may be one of the few areas where long-wavelength fluorophores are not very useful. However, if a suitable long-wavelength fluorophore could be found this could prove valuable.

#### 7.2.2 Double-labelled substrate

As double-labelled substrates possess potential advantages over other energy transfer methods, more work in this area could be fruitful It is evident that much more work is required to determine the optimum donor and acceptor ratios required for the substrate to be used as a protease substrate. The use of Cy 5 and Cy 5 5 as donor and acceptor respectively would be worth further study. The effect of bandwidth is one experimental parameter that could be investigated. The use of narrower bandwidths would result in less direct acceptor excitation, although presumably at the expense of sensitivity. Laser excitation may therefore be more promising as presumably laser excitation would allow less direct excitation of the acceptor along with more sensitivity

#### 7.2.3 Single-labelled substrate

This type of assay has been most suited for the determination of proteolytic enzymes and their inhibitors for use in flow-based screening programmes. In addition to the further work on multi-determination that needs to be carried out, which is described in section 7.2.4, the enhancement of the operation of the flow-based procedure described should be considered. Computer-controlled valves could replace the manually operated switching and injection valves, thereby providing a semi-automated system.

#### 7.2.4 Multi-determination

Multi-determination allows a higher throughput and thus is extremely important in the future development of the single-labelled substrates for high throughput screening. The work that was carried out showed that Pierce UltraLink

Biosupport Medium was extremely promising as a support for protease immobilisation. To begin, enzyme columns containing this support could be integrated into the proposed dual inhibition assay described in section 6.2.1 and depicted in figure 6.1. Using the proteolytic enzymes and inhibitors from the assays described in chapter 5, numerous set-ups are possible. Figure 7.1 illustrates just a couple of examples of possible set-ups out of the many combinations that are possible.

Numerous other set-ups are possible for multi-determination and various other supports are available for enzyme immobilisation. Much more work can therefore be carried out in the area of multi-determination.

However, although parallel channels allow multi-determination, which in turn produces a greater throughput of samples, the current macro-system has a limit. It would be impractical to keep paralleling up because of the amount of tubing that would be required and the number of injection valves needed. This is particularly problematic when carrying out the assays manually. Space requirements would also increase, as well as the cost associated with high reagent consumption. This macro system therefore becomes limiting. The designs of such assays are much more suited to miniaturised assay formats, and hence a discussion of the transferral of these protease and protease inhibition assays to a miniaturised system is given.

## Figure 7.1 – Possible set-ups for multi-determination assays: (a) one inhibitor, three enzymes with three different pH optima, (b) one inhibitor, three enzymes with two different pH optima



Substrate (S = BODIPY:α-casein) and inhibitor (I = aprotinin) are injected into three streams of buffer with neutral, alkaline, and acidic pH corresponding to streams flowing through α-chymotrypsin, proteinase K, and pepsin enzyme reactors respectively, to the detector which will detect selective inhibition of the first peak (α-chymotrypsin), and no inhibition of the second or third peaks (proteinase K and pepsin respectively).



Substrate (S = BODIPY:α-casein) and inhibitor (I = aprotinin) are injected into two streams of buffer with alkaline and neutral pH corresponding to streams flowing through alkaline protease and proteinase K (alkaline) and α-chymotrypsin (neutral) enzyme reactors, to the detector which will detect no inhibition of the first and second peaks (alkaline protease and proteinase K), and selective inhibition of the third peak (α-chymotrypsin).

#### 7.2.5 Miniaturised protease and protease inhibition assays

Section 1.5.1 discussed the pressures that currently face drug discovery companies and the role of HTS to harness the discovery potential of large numbers of compounds efficiently, by reducing time and cost and increasing information gained. The need to minimise the cost of screening has been addressed primarily by reducing the volume of sample to be screened, which, in turn, has resulted in the miniaturisation of HTS technology as a whole.

The benefits of the miniaturisation of HTS were discussed in section 1.5.4, and due to the potential improvements that could be achieved by miniaturising the protease and protease inhibition assays, this seems an obvious step in their continuing development.

Many discussions on the general concepts and techniques, and the background of the emergence of this technology can be found [298 - 306].

Chip-based systems have been described previously for enzyme assays including  $\beta$ -galactosidase [307]. Its inhibition by various inhibitors, performed in a microfabricated channel network using electrokinetic flow to precisely mix substrate, enzyme and inhibitor, was also illustrated. A schematic of the enzyme analysis chip taken from the paper in shown in figure 7.2.



### Figure 7.2 – Schematic diagram of the chip designed for enzyme assays in [307]

Microchip-based assays have also been described for protein kinase A [308] and ACLARA BioSciences, Inc., have reported the design, fabrication and characterisation of plastic microfluidic devices for conducting highly parallel enzyme-based assays [309]. Caliper Technologies Corporation have commercialised lab-on-a-chip technology [310] and have fabricated high throughput systems to perform on the order of tens of thousands of experiments per day per chip, using nanoliter volumes of reagents [311]. Chips have been designed for assays with fluorogenic substrates, e.g. protease assays, for biochemical assays, e.g. kinase assays, and for cellular assays, e.g. G-protein coupled receptor assays [312].

Using a chip-based system, the ability to manufacture a set-up with many parallel channels is much simpler than with a macro system. The soluble enzyme multi-determination system that was described in section 6.4 would be much better suited to a miniaturised system. Many parallel channels of different lengths would be easily manufactured in such a system, allowing the determination of two or more activities in nanoliter volumes using time separation. The need for such small volumes of reagents means that the soluble enzyme set-up could be used without the huge expenses found with such a set-up on a macro scale. The small amounts of sample required in miniaturised systems are also much better suited to the small amounts of compounds that are synthesised by combinatorial chemistry methods, allowing a better connection of combinatorial chemistry to HTS.

Alternatively, the immobilised enzyme format could also be used. The fabrication of micro-porous silica structures providing an efficient method of retaining a catalyst within a reactor system has been demonstrated by researchers at the University of Hull [313].

One area that needs to be considered in the design of chip-based protease and protease inhibition assays is the method of heating the assay components. From an examination of the literature, it appears that several methods are suitable, although the following overview is by no means exhaustive. A simple and common method to heat fluids in a chip is by the use of heating blocks [314 - 316]. This is an example of contact heating.

Another commonly used method is that of resistive heating [317, 318] Heating accomplished using a resistive heater and thermocouple for temperature monitoring has been used for DNA analysis where the heater and thermocouple are mounted to the bottom of the chip [317], and for the polymerase chain reaction (PCR) mounted on the top [318]. An integrated microchip device for the

digestion, separation and postcolumn labelling of protein and peptides [319] has made use of a small detachable resistive heater. A thin Thermofol<sup>TM</sup> Heater (Minco Products, Inc.) was used to control the temperature in the reaction channel. The heater was attached to the glass chip using a C-clamp and thermal contact between the heater and the microchip was provided using a piece of thermally conductive silicone elastomer. The heater temperature and the glass temperature on top of the microchip were measured using thermocouples, the temperatures of which did not differ more than 2 °C in the temperature range 20 to 80 °C down the length of the reaction channel. Simplified calculations showed that heating less than 2 mm length of the reaction channel (11.4  $\mu$ m deep and 43.6  $\mu$ m wide) is necessary to heat the fluid inside to the desired temperature.

Microfabricated devices constructed from glass and PDMS with integral heaters have been described [320], which were used for heterogeneous catalysis reactions. The *in situ* heater was fabricated from Nichrome wire (0.25 mm diameter) carefully formed into a serpentine shape and immobilised in the PDMS top plate and covering the channels. In the finished chips, the heating element was 2 mm from the catalyst surface and was able to heat 15 mm of channel length. Heating was achieved by a potentiostat (0 – 270 V) and monitored via a digital thermometer with the temperature probe located close to the reaction channel.

The methods already discussed have all been examples of contact heating However, non-contact heating allows for fast switching of temperature and reactions, as opposed to the conventional contact devices.

A non-contact heating method has been demonstrated for control of a chemical reaction in a microchip by utilising the photothermal effect produced by a compact diode laser [321]. A light absorbing target on top of a glass microchip above a microchannel through which reactants flowed was utilised to heat the reaction media. In order to increase the temperature locally in the microchannel,

a 10mW diode laser was used to heat an absorbing target (black ink point, produced by felt tip marker) Heat flux was conducted through the glass cover plate and an increase in the temperature of the mixed solution in the main microchannel was noted. The reaction media were heated in a stopped flow mode. Reaction volume was estimated to be 10 nl and with laser powers of 2, 4, 7, 8, and 10 mW, the temperature change induced was estimated to be 1.9, 3.2, 5.0, 5.6, and 7 °C, respectively for 5 s of irradiation. The small heat capacity of the heated reaction space allowed the equilibrium temperature to be reached within 2 s

Non contact heating was also applied to PCR, which was carried out in capillary tubes rather than on a microchip [322]. Infrared light was produced by a tungsten lamp, which was focussed on the optically transparent PCR reaction chamber where the sample was heated due to the absorption of radiation. The temperature was sensed in a reference capillary with a thermocouple. A mirror placed behind the capillaries enhanced the effect of heating. A workable solution is provided by the construction of a reference probe that exhibits identical thermal properties as the sample. Using this method, a heating rate of 65 °C / s was seen.

Three distinct methods of heating are apparent: heating blocks (contact), resistive heating (contact), and using a light source (non-contact). Heating blocks are simple, but bulky. Resistive heating seems promising, with the advantage of different combinations. The Integral heaters immobilised in the PDMS top plate used in [320] are easy to produce and the Thermofoil<sup>™</sup> heater (Minco Products, Inc.) that was used as a detachable heater [319] also appears promising. The non-contact methods, although faster, are expensive However, the speed obtained with these methods is not crucial for protease and protease inhibition assays.

A potentially tricky part of the design of a miniaturised system would be the excitation of the fluorophore and subsequent detection. Detection of fluorescence on a chip is described briefly in many papers [323 - 330], which possess a general scheme of using laser excitation by focusing the laser with a lens into the reaction channel and collection of fluorescence with a microscope objective and detection using a photo-multiplier tube (PMT).

Ocvirk, G. *et al.*, [331] reported a detailed confocal detection scheme for subpicomolar detection. Excitation from an air-cooled argon ion laser (488 nm) passed through a 485 nm bandpass filter, was reflected by a dichroic mirror and was focused onto the glass device using a microscope objective. Fluorescence emission was collected, passed through a dichroic mirror and measured with a PMT. A schematic of this set-up is shown in figure 7.3, taken from the paper.



### Figure 7.3 – Schematic diagram showing the set-up for a confocal fluorescence detection on a chip, taken from [331]

A completely integrated microfluidic / micro-optic device has been described [332], in which refractive microlens arrays and aluminium aperture arrays were integrated on both sides of a chemical chip.

A disposable sensor chip and optical readout device well suited for medical pointof-care diagnostics used an evanescent field generated by total internal reflection of a laser beam [333] This field was then used for excitation of fluorophore markers. The sensor chip was optically connected to a prism of the optical readout device by a refractive-index-matching fluid.

Two different ways to get light into and fluorescence out of a chip seem to be promising. Firstly, the use of lasers (LEDs etc.), focusing lenses / microscope objectives, and filters in order to accurately focus the excitation light into the channel and then detect using a PMT. Secondly, the fabrication of waveguides in the chip coupled to fiber optics to get the light both in and out. These methods would require much thought and planning in order to develop a realistic miniaturised system for the determination of proteolytic enzymes and their inhibitors.

### References

#### REFERENCES

- Bechtol, K.B., Hanzel, D.K., and Liang, B.C. Molecular Dynamics, Inc. (1999). "Technical Note 57. Understanding Fluorescence". Accessed via www. <u>http://www.mdyn.com/application\_notes/technotes/TN57.htm</u> (13 August 1999).
- [2] MacTaylor, C.E., and Ewing, A.G. (1997). Critical Review of Recent Developments in Fluorescence Detection for Capillary Electrophoresis. Electrophoresis 18, 2279 – 2290.
- [3] Bright, F.V. (1988). Bioanalytical Applications of Fluorescence Spectroscopy. Analytical Chemistry 60 (18), 1031A – 1039A.
- [4] Pesce, A.J., Rosén, C-G., and Pasby, T.L., eds. (1971). Fluorescence Spectroscopy. An Introduction for Biology and Medicine. New York: Marcel Dekker Inc.
- [5] Guilbault, G.G. (1985). Principles of Luminescence Spectroscopy. Luminescent Determination of Clinically and Agriculturally Important Samples. *Pure and Applied Chemistry* 57 (3), 495-514.
- [6] Lakowicz, J.R. (1999). Principles of Fluorescence Spectroscopy. Second Edition. New York: Kluwer Academic / Plenum Publishers.
- [7] Soper, S.A., Warner, I.M., and McGown, L.B. (1998). Molecular Fluorescence, Phosphorescence, and Chemiluminescence Spectrometry. *Analytical Chemistry* **70**, 477R – 494R.

- [8] Oldham, P.B., McCarroll, M E., McGown, L.B., and Warner, 1 M. (2000).
  Molecular Fluorescence, Phosphorescence, and Chemiluminescence Spectrometry. *Analytical Chemistry* 72, 197R – 209R.
- [9] Evans, E.H., Dawson, J B, Fisher, A, Hill, S.J., Price, W.J, Smith, C.M.M., Sutton, K.L, and Tyson, J,F. (2001). Atomic Spectrometry Update. Advances in Atomic Emission, Absorption, and Fluorescence Spectrometry, and Related Techniques. *Journal of Analytical Atomic Spectrometry* 16, 672 – 711.
- [10] Hill, S J., Chenery, S., Dawson, J.B., Evans, E.H., Fisher, A., Price, W J., Smith, C.M.M., Sutton, K L., and Tyson, J F. (2000). Advances in Atomic Emission, Absorption, and Fluorescence Spectrometry, and Related Techniques. *Journal of Analytical Atomic Spectrometry* **15**, 763 – 805.
- [11] Nuijens, M.J., Zomer, M., Mank, A J.G , Gooijer, C., Volthorst, N.H., and Hofstraat, J W. (1995). A Cyanine Fluorophore With a Hydrazine Functionality as<sup>7</sup> Labelling Reagent for Aldehydes in Liquid Chromatography. *Analytica Chimica Acta* 311, 47 - 55
- [12] Passwater, R A. (1973). Scatter Interference. *Fluorescence News* 7(3), 17-23.
- [13] Patonay, G., and Antoine, M.D. (1991). Near-Infrared Fluorogenic Labels New Approach to an Old Problem Analytical Chemistry 63, 321A – 327A.
- [14] Heitler, W. (1954). *The Quantum Theory of Radiation*. Oxford: Clarendon Press.

- [15] Miller, J.N. (1995). "Long Wavelength Fluorescence Spectroscopy". In Burgess, C., and Jones, D.G., eds. Spectrophotometry, Luminescence and Colour; Science and Compliance. Elsevier Science B.V., pp 247 – 253.
- [16] Miller, J.N., Brown, M.B., Seare, N.J., and Summerfield, S. (1993)
  "Analytical Applications of Very Near-IR Fluorimetry". In *Fluorescence Spectroscopy New Methods and Applications* Springer-Verlag, Chapter 14.
- [17] Sigma-Aldrich Company, Ltd. (1999). "Fluorescein Isothiocyanate Isomer I". Accessed via www. http://www.sigmaaldrich.com/sigma/proddata/f7250.htm (25 May 1999).
- [18] Lakowicz, J.R. (1999). In *Principles of Fluorescence Spectroscopy*. Second Edition. New York: Kluwer Academic / Plenum Publishers, pp 67 - 69.
- [19] Treibs, A., and Kreuzer, F-H. (1968). Difluorboryl-komplexe von di und tripeyrrylmethenen. *Liebigs Annalen der Chemie* **718**, 203 223.
- [20] Worries, H.J., Koek, J.H., Lodder, G., Lugtenberg, J., Fokkens, R., Driessen, O., and Mohn, G.R. (1985). A Novel Water-Soluble Fluorescent Probe: Synthesis, Luminescence and Biological Properties of the Sodium Salt of the 4-Sulfonate-3, 3', 5, 5'-tetramethyl-2, 2'pyrromethen-1, 1'-BF<sub>2</sub> complex *Recueil des Travaux Chimiques des Pays-Bas* **104**, 288 - 291.
- [21] Haugland, R.P., Kang, H.C. (1988). "Chemically Reactive Dipyrrometheneboron Difluoride Dyes". US Patent 4, 774, 339.

- [22] Karolin, J., Johansson, L.B-A, Strandberg, L., and Ny, T. (1994) Fluorescence and Absorption Spectroscopic Properties of Dipyrrometheneboron Difluoride (BODIPY) Derivatives in Liquids, Lipid Membranes, and Proteins Journal of the American Chemical Society 116, 7801 – 7806
- [23] Johnson, I.D., Kang, H.C., and Haugland, R.P. (1991). Fluorescent Membrane Probes Incorporating Dipyrrometheneboron Difluoride Fluorophores. *Analytical Biochemistry* **198**, 228 – 237.
- [24] Kang, H.C., and Haugland, R.P. (1995). "Dibenzopyrrometheneboron Difluoride Dyes". US Patent 5, 433, 896.
- [25] Mujumdar, R.B., Ernst, L.A., Mujumdar, S.R., Lewis, C.J., and Waggoner,
  A.S. (1993). Cyanine Dye Labeling Reagents: Sulfoindocyanine
  Succinimidyl Esters. *Bioconjugate Chemistry* 4 (2), 105 109.
- [26] Gruber, H.J., Kada, G, Pragl, B., Riener, C., Hahn, C D., Harms, G.S., Ahrer, W., Dax, T.G., Hohenthanner, K., and Knaus, H-G. (2000). Preparation of Thiol-Reactive Cy5 Derivatives from Commercial Cy5 Succinimidyl Ester. *Bioconjugate Chemistry* 11, 161 – 166.
- [27] Briggs, M S., Burns, D.D., Cooper, M E., and Gregory, S.J. (2000). A pH Sensitive Fluorescent Cyanine Dye for Biological Applications. *Chemical Communications*, 2323 – 2324
- [28] Seitz, W.R. (1982). Analytical Derivatization Reactions Based on Changes in Fluorescence Polarization: A Worthwhile Idea?. Applied Spectroscopy 36 (2), 161 – 167.

- [29] Roemelt, P.M. et al. (1980). Selective Analysis of Binary Fluorophor Mixtures by Fluorescence Polarization. Analytical Chemistry 52, 769 – 771.
- [30] Nasir, M S, and Jolley, M E (1999). Fluorescence Polarization: An Analytical Tool for Immunoassay and Drug Discovery. *Combinatorial Chemistry and High Throughput Screening* 2, 177 – 190.
- [31] Jolley, M.E. (1999). "Principles of Fluorescence Polarization. What is Fluorescence Polarization ?" Accessed via www. www.jolley.com/jolleyfiles/learning.htm (16 August 1999).
- [32] Pesce, A.J., Rosén, C-G., and Pasby, T.L., eds. (1971). "Chapter 4. Polarized Fluorescence". In *Fluorescence Spectroscopy. An Introduction* for Biology and Medicine. New York: Marcel Dekker Inc., pp 87 – 111.
- [33] Perrin, F. (1926). Polarization de la lumiere de fluorescence. Vie moyenne de molecules dans l'etat excite. Journal de Physique et la Radium 7, 390 – 401.
- [34] Wu, P., and Brand, L. (1994). Review. Resonance Energy Transfer: Methods and Applications. *Analytical Biochemistry* 218, 1 – 13.
- [35] Pesce, A.J., Rosén, C-G., and Pasby, T.L., eds. (1971) "Chapter 5. Energy Transfer". In *Fluorescence Spectroscopy*. An Introduction for Biology and Medicine. New York: Marcel Dekker Inc., pp 131 – 148.
- [36] "Fluorescence Resonance Energy Transfer". Accessed via www. <u>http://www.idtdna.com/techbulletins/Fluorescence\_Resonance\_Energy\_Tr</u> <u>ansfer asp</u> (29 November 2000).

- [37] Lakowicz, J.R. (1999). "Chapter 13. Energy Transfer". In Principles of Fluorescence Spectroscopy. Second Edition. New York: Kluwer Academic / Plenum Publishers, pp 368 – 394
- [38] Forster, T. (1948). Zwischenmolekulare Energiewanderung und Fluoreszenz. Annalen der Physik (Leipzig) 2, 55 75.
- [39] Stryer, L., and Haugland, R.P. (1967). Energy Transfer: A Spectroscopic Ruler. *Proceedings of the National Academy of Science USA* 58, 719 – 726.
- [40] Matayoshi, E.D., Wang, G.T., Krafft, G.A., and Erickson, J. (1990). Novel Fluorogenic Substrates for Assaying Retroviral Proteases by Resonance Energy Transfer. Science 247, 954 – 958.
- [41] Lim, C.S., Miller, J.N., and Bridges, J.W. (1980). Energy-Transfer Immunoassay: A study of the Experimental Parameters in an Assay for Human Serum Albumin. *Analytical Biochemistry* **108**, 176 – 184.
- [42] Burghart, A., Thorsen, L.H., Chen, J., Burgess, K., Bergström, F., and Johansson, L.B. (2000). Energy Transfer Cassettes Based on BODIPY® Dyes. *Chemical Communications*, 2203 – 2204.
- [43] Mitsui, T., Nakano, H., and Yamana, K. (2000) Coumarin-Fluorescein Pair as a New Donor-Acceptor Set for Fluorescence Energy Transfer Study of DNA. *Tetrahedron Letters* 41, 2605 – 2608.
- [44] Jean, F., Basak, A., DiMaio, J., Seidah, N.G., and Lazure, C. (1995). An Internally Quenched Fluorogenic Substrate of Prohormone Convertase 1 and Furin Leads to a Potent Prohormone Convertase Inhibitor. *Biochemical* Journal 307, 689 – 695.

- [45] Oswald, B., Lehmann, F., Simon, L., Terpetschnig, E., and Wolfbeis, O.S.
  (2000). Red Laser-Induced Fluorescence Energy Transfer in an Immunosystem. *Analytical Biochemistry* 280, 272 – 277.
- [46] Hung, S-C., Mathies, R.A., and Glazer, A.N. (1998). Comparison of Fluorescence Energy Transfer Primers with Different Donor-Acceptor Dye Combinations. *Analytical Biochemistry* 255, 32 – 38.
- [47] Ruzicka, J., and Hansen, E.H. (1975). Flow Injection Analysis. Part 1. A New Concept of Fast Continuous Flow Analysis. *Analytica Chimica Acta* 78, 145 – 157.
- [48] Nagy, G., Feher, Z., and Pungor, E. (1970). Application of Silicone Rubber-Based Graphite Electrodes for Continuous Flow Measurements. Part II. Voltammetric Study of Active Substances Injected Into Electrolyte Streams. Analytica Chimica Acta 52, 47 - 54.
- [49] Stewart, K K., Beecher, G R., and Hare, P.E. (1976). Rapid Analysis of Discrete Samples: the Use of Nonsegmented, Continuous Flow. *Analytical Biochemistry* 70, 167 – 173.
- [50] Ranger, C B. (1981). Flow Injection Analysis. Principles, Techniques, Applications, and Design. *Analytical Chemistry* **53**(1), 20A – 32A.
- [51] Hansen, E.H (1995). Flow-Injection Analysis: Leaving its Teen-Years and Maturing. A Personal Reminiscence of its Conception and Early Development. Analytica Chimica Acta 308, 3 – 13.
- [52] Snyder, L.R. (1980). Continuous-Flow Analysis: present and Future. Analytica Chimica Acta 114, 3 – 18.

- [53] Ruzicka, J, and Hansen, EH. (1980). Flow Injection Analysis. Principles, Applications and Trends. *Analytica Chimica Acta* **114**, 19 – 44.
- [54] Ruzicka, J., and Christian, G.D. (1990). Flow Injection Analysis and Chromatography: Twins or Siblings? *Analyst* **115**, 475 – 486.
- [55] Puchades, R., and Maquieira, A. (1996). Recent Developments in Flow Injection Immunoanalysis. *Critical Reviews in Analytical Chemistry* 26 (4), 195-218.
- [56] Kolev, S D. (1995). Mathematical Modelling of Flow-Injection Systems. Analytica Chimica Acta **308**, 36 – 66.
- [57] Ruzicka, J., and Hansen, E.H. (2000). Flow Injection Analysis. *Analytical Chemistry*, 212A 217A.
- [58] Jaromir Ruzicka and Elo H. Hansen (1988). Flow Injection Analysis. Second Edition. Volume 62 in Chemical Analysis: a Series of Monographs on Analytical Chemistry and its Applications, Series Editor J.D. Winefordner, Editor Emeritus I.M. Kolthoff. New York: John Wiley and Sons Inc.
- [59] Bo Karlberg and Gil E. Pacey (1989). Techniques and Instrumentation in Analytical Chemistry volume 10. Flow Injection Analysis A Practical Guide. Amsterdam: Elsevier Science Publishers B.V.
- [60] Hansen, E.H., Ruzicka, J., and Animesh, K. (1978). Flow Injection Analysis for Calcium in Serum, Water and Waste Waters by Spectrophotometry and by Ion-Selective Electrode. *Analytica Chimica Acta* 100, 151 - 165.

- [61] Baban, S., Beetalstone, D., Betteridge, D., and Sweet, P. (1980). The Determination of Sulphate by Flow-Injection Analysis With Exploitation of pH Gradients and EDTA. *Analytica Chimica Acta* **114**, 319 - 323.
- [62] Ruzicka, J., and Marshall, G D. (1990). Sequential Injection a New Concept for Chemical Sensors, Process Analysis and Laboratory Assays. *Analytica Chimica Acta* 237, 329 – 343.
- [63] Workman, Jr., J., Veltkamp, D.J., Doherty, S., Anderson, B.B., Creasy, K.E., Koch, M., Tatera, J.F., Robinson, A.L., Bond, L., Burgess, L.W., Bokerman, G.N., Ullman, A.H., Darsey, G.P., Mozayeni, F., Bamberger, J.A., and Greenwood, M.S. (1999). Process Analytical Chemistry. Flow and Sequential Injection Analysis Analytical Chemistry 71, 156R – 158R.
- [64] Solich, P., Polášek, M., and Karlíček, R. (1995) Sequential Flow-Injection Spectrophotometric Determination of Coumarins Using a Double-Injection Single-Line System. *Analytica Chimica Acta* 308, 293 – 298
- [65] Ruzicka, J., and Scampavia, L. (1999). From Flow Injection to Bead Injection. *Analytical Chemistry News and Features*, 257A – 263A
- [66] Itabashi, H., Kawamoto, H., and Kawashima, T. (2001). A Novel Flow Injection Technique: All Injection Analysis. *Analytical Sciences* 17, 229 – 231.
- [67] Antiochia, R., and Palleschi, G. (1997). A Tri-Enzyme Electrode Probe for the Sequential Determination of Fructose and Glucose in the Same Sample Analytical Letters 30, 683 – 697.

- [68] Marcos, J., Rios, A., and Valcárcel, M. (1992). Automated Simultaneous Determination of Metal Ions by Use of Variable Flow Rates in Unsegmented Systems. *Analyst* **117**, 1629 - 1633.
- [69] Spohn, U., Van der Pol, J., Eberhardt, R., Joksch, B., and Wandrey, C. (1994). An Automated System for Multichannel Flow-Injection Analysis. *Analytica Chimica Acta* 292, 281 - 295.
- [70] Luque de Castro, M.D., and Cases, M.V. (1984). Simultaneous Determinations in Flow Injection Analysis. A Review Analyst 109, 413 – 419.
- [71] Kubáň, V. (1992). Simultaneous Determination of Several Components by Flow Injection Analysis. *Critical Reviews in Analytical Chemistry* 23 (1), 15 – 53.
- [72] Chen, R L.C., and Matsumoto, K. (1995). Sequential Enzymatic Monitoring of Glucose, Ethanol and Glutamate in Bioreactor Fermentation Broth Containing a High Salt Concentration by a Multi-Channel Flow-Injection Analysis Method. *Analytica Chimica Acta* 308, 145 – 151.
- [73] Bailey, D., and Brown, D. (2001). High-Throughput Chemistry and Structure-Based Design: Survival of the Smartest. *Drug Discovery Today* 6 (2), 57 – 59.
- [74] Silverman, L, Campbell, R., and Broach, J.R. (1998). New Assay Technologies for High-Throughput Screening. *Current Opinion in Chemical Biology* 2, 397 – 403.

- [75] Sittampalam, G S., Kahl, S.D., and Janzen, W.P. (1997) High-Throughput Screening: Advances in Assay Technologies Current Opinion in Chemical Biology 1, 384 – 391.
- [76] Drews, J. (2000). Drug discovery: a Historical Perspective. Science 287, 1960 – 1964.
- [77] Editor Rebecca Lawrence (2001). Genomics. New Tools for Drug Discovery. A Supplement to *Drug Discovery Today* **6** (5). S 63 SS118.
- [78] Jain, K K. (2001). Proteomics: Delivering New Routes to Drug Discovery
   Part 1. Drug Discovery Today 6 (15), 772 774.
- [79] Jain, K K (2001). Proteomics<sup>-</sup> Delivering New Routes to Drug Discovery - Part 2. Drug Discovery Today 6 (16), 829 – 832.
- [80] Rademann, J., and Jung, G. (2000). Integrating Combinatorial Synthesis and Bioassays. *Science* **287**, 1947 1848.
- [81] Burbaum, J.J. (1998). Miniaturization Technologies in HTS: How Fast, How Small, How Soon? *Drug Discovery Today* **3** (7), 313 – 322.
- [82] Burbaum, J. (1998). Engines of Discovery. Chemistry in Britain June Issue, 38 – 41.
- [83] Wölcke, J., and Ullmann, D. (2001). Miniaturized HTS Technologies uHTS. Drug Discovery Today 6 (12), 637 – 646.
- [84] Ehret, R., Baumann, W., Brishwein, M., Lehmann, M., Henning, T.,
  Freund, I., Drechster, S., Friedrich, U., Hubert, M-L., Motrescu, E., Kob,
  A., Palzer, H, Grothe, H., and Wolf, B. (2001). Multiparametric

Microsensor Chips for Screening Applications. *Fresenius Journal of Analytical Chemistry* **369**, 30 – 35.

- [85] Lubert Stryer (1995). *Biochemistry*. Fourth Edition New York<sup>-</sup> W.H. Freeman and Company, Chapter 8.
- [86] Lehninger, A.L., Nelson, D.L., and Cox, M.M. (1993) *Principles of Biochemistry*. Second Edition. New York<sup>-</sup> Worth Publishers, Inc
- [87] Trevor Palmer (1991). "Specificity of Enzyme Action". In Understanding Enzymes Third Edition. Ellis Harwood, Chapter 4.
- [88] Lehninger, A L., Nelson, D.L., and Cox, M M. (1997). "Enzymes". In Principles of Biochemistry. Second edition. USA: Worth Publishers Inc, pp 198 – 239.
- [89] Stryer, L. (1995). "Enzymes' Basic Concepts and Kinetics". In Biochemistry. Fourth Edition. USA: Freeman, pp181 – 206.
- [90] Palmer, T. (1991). In Understanding Enzymes. Third Edition. Ellis Harwood, pp118 – 138.
- [91] Hartley, B.S. (1960). Proteolytic Enzymes. Annual Review in Biochemistry 29, 45 – 72.
- [92] Nduwimana, J., Guenet, L., Dorval, I, Blayau, M., Le Gall, J.Y., and Treut, A.L. (1995). Proteases. *Annales de Biologie Clinique* **53**, 251 264.
- [93] Neurath, H. (1989). "The Diversity of Proteolytic Enzymes". In R.J. Beynon, and J A Bond, eds. *Proteolytic Enzymes A Practical Approach*. Oxford: Irl Press at Oxford University Press, Chapter 1

- [94] Barrett, A.J. (1986). "An Introduction to the Proteinases". In Barrett, A J and Salvesen, G., eds. *Proteinase Inhibitors*. Amsterdam, New York. Elsevier Science Publishers BV (Biomedical Division), Chapter 1.
- [95] Dunn, B.M. (1989). "Determination of Protease Mechanism". In Beynon,
  R.J., and Bond, J.S., eds. *Proteolytic Enzymes A Practical Approach*.
  Oxford: Irl Press at Oxford University Press, Chapter 4.
- [96] Alan J. Barrett, Neil D. Rawlings, and J. Fred Woessner, eds. (1998). Handbook of Proteolytic Enzymes. London: Academic Press.
- [97] Anwar, A., and Saleemuddin, M. (1998). Alkaline Proteases: A Review. Bioresource Technology 64, 175 – 183.
- [98] Ebeling, W, Hennrich, N., Klockow, M., Metz, H., Orth, H.D., and Lang, H. (1974). Proteinase K from *Tritirachium album* Limber. *European Journal* of Biochemistry 47, 91 – 97.
- [99] Bajorath, J., Saenger, W., and Pal, G P. (1988) Autolysis and Inhibition of Proteinase K, a Subtilisin-Related Serine Proteinase Isolated from the Fungus *Tritirachium album* Limber. *Biochimica et Biophysica Acta* 954, 176 – 182.
- [100] Nelson, J.M., and Griffin, E.G. (1916). Adsorption of Invertase. *Journal of the American Chemical Society* **38**, 1109 1115.
- [101] Tischer, W., and Wedekind, F. (19). Immobilized Enzymes: Methods and Applications. *Topics in Current Chemistry* **200**, 96 126.

- [102] Rao, S.V., Anderson, KW, and Bachas, L.G. (1998). Fundamental Review. Oriented Immobilization of Proteins. *Mikrochimica Acta* 128, 127 – 143.
- [103] Zaborsky, O. (1974). Immobilized Enzymes. Ohio: CRC Press.
- [104] Liang, J.F., Li, Y.T., and Yang, V.C. (2000). Biomedical Application of Immobilized Enzymes. *Journal of Pharmaceutical Sciences* 89 (8), 979 – 990.
- [105] Lam, S., and Malıkin, G. (1993). *Analytical Applications of Immobilised Enzymes.* London: Academic Press.
- [106] D'Souza, S.F. (1999). Immobilized Enzymes in Bioprocess. *Current Science* 77, 69 79.
- [107] Masoom, M., and Townshend, A. (1984). Determination of Glucose in Blood by Flow Injection Analysis and an Immobilized Glucose Oxidase Column. Analytica Chimica Acta 166, 111 – 118. (NB same as ref 120).
- [108] Peters, K., and Richards, F.M. (1977). Chemical Cross-linking. Glutaraldeyde. Annual Review of Biochemistry 46, 537 – 539.
- [109] Vidal, M., Prata, M., Santos, S., Tavares, T., Hossfeld, J., Preininger, C., and Oliva, A. (2000). Fluorescence IgG Immunosensor Based on a Micro Flow Cell Containing Controlled Pore Glass as Immobilisation Support. *Analyst* 125, 1387 – 1391.
- [110] Ghous, T., and Townshend, A (1996). Flow Injection Determination of Metrifonate by Immobilised Acetylcholinesterase Inhibition. *Analytica Chimica Acta* 332, 179 – 185. (NB same as ref 119).

- [111] Sánchez-Cabezuso, M., Fernández-Romero, J.M., Luque de Castro, M D. (1995). Fluorimetric-Flow Injection Determination of Theophylline Based on its Inhibitory Effect on Immobilised Alkaline Phosphatase. *Analytica Chimica Acta* 308, 159 – 163. (NB same as ref 122).
- [112] Phillips, T M, Queen, W.D., More, N S., and Thompson, A.M. (1985). Protein A-Coated Glass Beads. Universal Support Medium for High-Performance Immunoaffinity Chromatography. *Journal of Chromatography* 327, 213 – 219.
- [113] Ukeda, H., Ishii, T., Sawamura, M., and Kusunose, H. (1995). Dynamic Analysis of the Binding Process of Bovine Serum Albumin on Glutaraldeyde-Activated Controlled Pore Glass. *Analytica Chimica Acta* 308, 261 – 268.
- [114] Becker, T.M., and Schmidt, H-L. (2000). New Ways of Enzymatic Twosubstrate Determinations in Flow Injection Systems. *Analytica Chimica Acta* 421, 7 – 18.
- [115] Clark, D. (1994) Can Immobilization be Exploited to Modify Enzyme Activity? *Trends In Biotechnology* **12**, 439 – 443.
- [116] Goldstein, L., Levin, Y., and Katchalski, E. (1964). A Water-Insoluble Polyanionic Derivative of Trypsin. II. Effect of the Polyelectrolyte Carrier on the Kinetic Behaviour of the Bound Trypsin. *Biochemistry* 3, 1913 – 1919.
- [117] Line, W.F., Kwong, A., and Weetall, H.H. (1971). Pepsin Insolubilized by Covalent Attachment to Glass. Preparation and Characterization. *Biochimica et Biophysica Acta* 242, 194 – 202.

- [118] Masoom, M., and Townshend, A. (1986). Flow-Injection Determination of Sulphite and Assay of Sulphite Oxidase Analytica Chimica Acta 179, 399 - 405.
- [119] Ghous, T., and Townshend, A. (1996) Flow Injection Determination of Metrifonate by Immobilised Acetylcholinesterase Inhibition. *Analytica Chimica Acta* 332, 179 – 185. (NB same as ref 110).
- [120] Masoom, M., and Townshend, A. (1984). Determination of Glucose in Blood by Flow Injection Analysis and an Immobilized Glucose Oxidase Column. Analytica Chimica Acta 166, 111 – 118 (NB same as ref 107).
- [121] Murakami, Y., Takeuchi, T, Yokoyama, K, Tamiya, E., and Karube, I.
  (1993). Integration of Enzyme-Immobilized Column with Electrochemical Flow Cell Using Micromachining Techniques for a Glucose Detection System. Analytical Chemistry 65 2731 – 2735
- [122] Sánchez-Cabezuso, M., Fernández-Romero, J M, Luque de Castro, M.D. (1995). Fluorimetric-Flow Injection Determination of Theophylline Based on its Inhibitory Effect on Immobilized Alkaline Phosphatase. *Analytica Chimica Acta* 308, 159 – 163. (NB same as ref 111).
- [123] Tang, L.X., Rowell, F.J., and Cumming, R H. (1995) Rapid and Sensitive Assay for Some Protease Enzymes Using a Fluorosubstrate-Immobilized Bioreactor. *Analyst* 120, 1949 - 1952.
- [124] Frebel, H., Chemnitius, G C., Cammann, K, Kakerow, R., Rospert, M., and Mokwa, W. (1997). Multianalyte Sensor for the Simultaneous Determination of Glucose, L-Lactate and Uric Acid Based on a Microelectrode Array. Sensors and Actuators 43, 87 - 93.

- [125] Ensafi, AA, and Rezaei, B. (1998). Flow Injection Analysis Determination of Ascorbic Acid With Spectrofluorimetric Detection. *Analytical Letters* **31**, 333 - 342.
- [126] Jurkiewicz, M., Alegret, S., Almirall, J., Garcia, M., and Fàbrigas, E. (1998). Development of a Biparametric Bioanalyser for Creatinine and Urea. Validation of the Determination of Biochemical Parameters Associated with Hemodialysis. *Analyst* **123**, 1321 - 1327.
- [127] Hassan, S.S.M., Mahmoud, W.H., and El-Mosallamy, M.A.F. (1999). Determination of Metformin in Pharmaceutical Preparations Using Potentiometry, Spectrofluorimetry and UV-Visible Spectrophotometry. *Analytica Chimica Acta* 378, 299 - 311.
- [128] Mayer, C., Frauer, A., Schalkhammer, T., and Pittner, F. (1999). Enzyme-Based Flow Injection Analysis System for Glutamine and Glutamate in Mammalian Cell Culture Media. *Analytical Biochemistry* 268, 110 - 116.
- [129] Tsukatani, T, and Matsumoto, K. (2000). Flow-Injection Fluorometric Quantification of Succinate in Foodstuffs Based on the Use of an Immobilized Enzyme Reactor. *Analytica Chimica Acta* 416, 197 - 203.
- [130] Sugaya, T., Naito, S, Yonezawa, S., Morishita, F., and Kojima, T. (1988). Simultaneous Assay of Aspartate Aminotransferase and Alanine Aminotransferase by Flow Injection Method Using Immobilized Enzyme Reactors. Analytical Sciences 4, 579 – 582.

- [131] Matsumoto, K., Asada, K., and Mori, R. (1998). Simultaneous Biosensing of Inosine Monophosphate and Glutamate by Use of Immobilized Enzyme Reactors. *Analytica Chimica Acta* 358, 127 - 136.
- [132] Becker, T.M., and Schmidt, H-L. (2000). New Ways of Enzymatic Two-Substrate Determinations in Flow Injection Systems. *Analytica Chimica Acta* 421, 7 – 18.
- [133] Umoh, E.F., von Putten, A.B., and Schugerl, K. (1996). Simultaneous On-Line Monitoring of Glucose and Total Malto Sugar in Fermentation Processes Using an FIA System. *Journal of Chemical Technology and Biotechnology* 67, 276 - 280
- [134] Matsumoto, K., Matsubara, H., Handa, M., Ukeda, H., and Osajima, J. (1990). Simultaneous Determination of Glucose, Ethanol and Lactate in Alcoholic Beverages and Serum by Amperometric Flow Injection Analysis With Immobilized Enzyme Reactors. *Journal of Biotechnology* 14, 115 -126.
- [135] Chen, R.L.C., and Matsumoto, K. (1995). Sequential Enzymatic Monitoring of Glucose, Ethanol and Glutamate in Bioreactor Fermentation Broth Containing a High Salt Concentration by a Multi-Channel Flow-Injection Analysis Method. Analytica Chimica Acta 308, 145 – 151.
- [136] Tetley, T.D. (1993). Proteinase Imbalance: Its Role in Lung Disease. Thorax 48, 560 – 565.
- [137] Bradbury, J. (2001). Neutrophil Elastase Inhibitors for Cystic Fibrosis. Drug Discovery Today 6 (9), 441 – 442.
- [138] Hay, D.W.P. (2000). Chronic Obstructive Pulmonary Disease: Emerging Therapies. *Current Opinion in Chemical Biology* **4**, 412 – 419.
- [139] Axys. (2000). "The Protease Tour page 3. Bone Building". Accessed via www. http://www.axyspharm.com/protease/pg3.html (27 September 2000).
- [140] Maeda, H., and Yamamoto, T. (1996). Pathogenic Mechanisms Induced by Microbial Proteases in Microbial Infections. *Biological Chemistry Hoppe-Seyler* 377, 217 – 226.
- [141] Lively, M.O. (1999).. "Bacterial Type I Signal Peptidases". In Benn M. Dunn, ed. Proteases as Infectious Agents. USA: Academic Press, pp 219 231.
- [142] Qiu, XD., and Abdel-Meguid, S.S. (1999). "Human Herpesvirus Proteases". In Ben M. Dunn, ed. *Proteases of Infectious Agents*. USA: Academic Press, pp 93 – 115.
- [143] Walker, M.A. (1999). Hepatitis C Virus: an Overview of Current Approaches and Progress. *Drug Discovery Today* **4** (11), 518 529.
- [144] Corvas (2000). "Corvas Programs and Markets Other Diseases. Hepatitis C Viral Protease Inhibitor". Accessed via www. http://www.corvas.com/other-diseases.html (27 September 2000).
- [145] Colin Berry (1999). "Proteases as Drug Targets for the Treatment of Malaria". In Ben M Dunn, ed. Proteases of Infectious Agents. USA: Academic Press, pp 165 – 188.

- [146] Health Central (2000). "Health Central Articles. Fewer Achoos? Cure for the Common Cold may be in Sight". Accessed via www. http://www.healthcentral.com/drdean/deanfulltexttopics.cfm?id=8096 (20 April 2000).
- [147] Schnebli, H.P., and Braun, N.J. (1986). "Proteinase Inhibitors as Drugs"
   In Barrett, A.J., and Salvesen, G, eds. *Proteinase Inhibitors*. Elsevier
   Science Publishers BV (Biomedical Division), pp 613 627.
- [148] Southan, C. (2001). A Genomic Perspective on Human Proteases as Drug Targets. *Drug Discovery Today* **6** (13), 681 – 688.
- [149] Shafer, J.A. (1998). Cardiovascular Chemotherapy: Anticoaguiants. *Current Opinion in Chemical Biology* **2**, 458 – 465.
- [150] Ripka, W.C, and Vlasuk, G.P. (1997). Chapter 8. Antithrombotics / Serine Proteases. Annual Reports in Medicinal Chemistry 32, 71 – 89.
- [151] Vacca, J.P. (2000). New Advances in the Discovery of Thrombin and Factor Xa Inhibitors. *Current Opinion in Chemical Biology* **4**, 394 400.
- [152] Corvas (2000). "Corvas Programs and Markets Cardiovascular". Accessed via www. http://www.corvas.com/cardiovascular.html (27 September 2000).
- [153] Oefner, C, Binggeli, A., Breu, V., Bur, D., Clozel, J-P., D'Arcy, A., Dorn, A., Fischli, W., Grüninger, F., Guller, R., Hirth, G., Märki, H, P., Mathews, S, Müller, M., Ridley, R.G, Stadler, H., Vieira, E., Wilhelm, M., Winkler, F.K., and Wostl, W. (1999). Renin Inhibition by Substituted Piperidines: a Novel Paradigm for the Inhibition of Monomeric Aspartic Proteinases? *Chemistry and Biology* 6, 127 131.

- [154] DeFrancesco, L. (2000). "Serine Protease Inhibitor Prevents HIV Infection". Accessed via www. http://www.bioresearchonline.com/content/news/article.asp (9 January 2001).
- [155] Foda, H.D., and Zucker, S. (2001). Matrix Metalloproteinases in Cancer Invasion, Metastasis and Angiogenesis. *Drug Discovery Today* 6 (9), 478 – 482.
- [156] Coussens, L.M., and Werb, Z. (1996). Matrix Metalloproteinases and the Development of Cancer. *Chemistry and Biology* **3**, 895 904.
- [157] Moss, M.L., White, J.M., Lambert, M.H., and Andrews, R.C. (2001). TACE and Other ADAM Proteases as Targets for Drug Discovery. Drug Discovery Today 6 (8), 417 – 426.
- [158] Brown, T.L., Yet, M-G., and Wold, F. (1982). Substrate-Containing Gel Electrophoresis: Sensitive Detection of Amylolytic, Nucleolytic, and Proteolytic Enzymes. *Analytical Biochemistry* **122**, 164 – 172.
- [159] Kelleher, P.J., and Juliano, R.L. (1984). Detection of Proteases in Polyacrylamide Gels Containing Covalently Bound Substrates. Analytical Biochemistry 136, 470 – 475.
- [160] Ohlsson, B.G., Westström, B.R., and Karlsson, B.W. (1986). Enzymoblotting: A Method for Localizing Proteinases and Their Zymogens Using para-Nitroanilide Substrates after Agarose Gel Electrophoresis and Transfer to Nitrocellulose. *Analytical Biochemistry* **152**, 239 – 244.

- [161] Roffman, S., and Troll, W. (1974). Microassay for Proteolytic Enzymes
   Using a New Radioactive Anilide Substrate. *Analytical Biochemistry* 61, 1 5.
- [162] Chang, B-W., Chen, R.L.C., Huang, I-J., and Chang, H-C. (2001). Assays for Angiotensin Converting Enzyme Inhibitory Activity. *Analytical Biochemistry* 291, 84 – 88.
- [163] Schlage, W.K. (1988). A Rapid and Sensitive Microassay for Bacterial Collagenase and Other Proteolytic Activities on Collagenous Substrates. *Biological Chemistry Hoppe-Seyler* 369, 357 – 363
- [164] Fournout, S., Roquet, F., Salhi, S.L., Seyer, S L., Valverde, V., Masson, J.M., Jouin, P., Pau, B., Nicolas, M., and Hanin, V. (1997). Development and Standardization of an Immuno-Quantified Solid Phase Assay for HIV-1 Aspartyl Protease Activity and Its Application to the Evaluation of Inhibitors. *Analytical Chemistry* 69, 1746 1752.
- [165] Gan, Z., Marquardt, R R., and Xiao, H. (1999). Protease and Protease Inhibitor Assays Using Biotinylated Casein Coated on a Solid Phase. *Analytical Biochemistry* 268, 151 – 156.
- [166] O'Driscoll, B.M., Rattray, F.P., McSweeney, P.L.H., and Kelly, A.L.
   (1999). Protease Activities in Raw Milk Determined Using a Synthetic Heptapeptide Substrate. *Journal of Food Science* 64 (4), 606 611.
- [167] Kuwada, M., and Katayama, K. (1984). Differentiation of Endopeptidases and Aminopeptidases by High-Performance Liquid Chromatography of Reaction Products from Chromogenic Peptide p-Nitroanilines as Substrates. Analytical Biochemistry 139, 438 – 443.

- [168] Pazhanisamy, S, Stuver, C.M., and Livingston, D.J (1995). Automation of a High-Performance Liquid Chromatography-Based Enzyme: Evaluation of Inhibition Constants for Human Immunodeficiency Virus-1 Protease Inhibitors. *Analytical Biochemistry* 229, 48 – 53.
- [169] Antal, J., Pal, G, Asboth, B, Buzas, Z., Patthy, A., and Graf, L. (2001). Specificity Assay of Serine Proteinases by Reverse-Phase High-Performance Liquid Chromatography Analysis of Competing Oligopeptide Substrate Library. *Analytical Biochemistry* 288, 156 – 167.
- [170] Deo, S.K., Lewis, J.C., and Daunert, S. (2000). Bioluminescence Detection of Proteolytic Bond Cleavage by Using Recombinant Aequorin. *Analytical Biochemistry* 281, 87 – 94.
- [171] Edwards, R., and Townshend, A (1995). Chemiluminescence Determination of Proteases by Flow Injection Using Immobilized Isoluminol. Analyst 120, 117 – 120.
- [172] Branchini, B.R., Hermes, J.D., Salituro, F.G., Post, N J., and Claeson, G.
   (1981). Sensitive Enzyme Assays Based on the Production of Chemiluminescent Leaving Groups. *Analytical Biochemistry* 111, 87 – 96.
- [173] Branchini, B.R., Salituro, F.G., Hermes, J.D., and Post, N.J. (1980).
   Highly Sensitive Assays for Proteinases Using Immobilized Luminogenic Substrates. *Biochemical and Biophysical Research Communications* 97 (1), 334 – 339.
- [174] Bronstein, I., Edwards, B., Martin, C., Sparks, A, and Voyta, J.C N. US Patent Abstracts 5591591. Dioxetane Compounds for the Chemiluminescent Detection of Proteases, Methods of Use and Kits Therefore, p711.

- [175] Zhang, R., Beyer, B.M., Durkin, J., Ingram, R., Njoroge, F.G., Windsor, W.T., and Malcolm, B.A. (1999). A Continuous Spectrophotometric Assay for the Hepatitis C Virus Serine Protease. *Analytical Biochemistry* 270, 268 275.
- [176] Stebbins, J, and Debouck, C. (1997). A Microtiter Colorimetric Assay for the HIV-1 Protease. Analytical Biochemistry 248, 246 – 250.
- [177] Chang, B-W, Chen, R.L.C., Huang, I-J., and Chang, H-C. (2001). Assays for Angiotensin Converting Enzyme Inhibitory Activity. Analytical Biochemistry 291, 84 – 88.
- [178] Kourteva, I., Sleigh, R.W., and Hjertén, S. (1987). Assay for Enzyme Inhibition: Detection of Natural Inhibitors of Trypsin and Chymotrypsin. *Analytical Biochemistry* **162**, 345 – 349.
- [179] Sarath, G, De La Motte, R.S. and Wagner, F.W. (1989). "Protease assay Methods". In Beynon, R.J., and Bond, J.S., eds. *Proteolytic Enzymes – A Practical Approach* England: Irl Press, Chapter 3.
- [180] Knight, C.G. (1995). Fluorimetric Assays of Proteolytic Enzymes. Methods in Enzymology 248, 18 – 34.
- [181] Gershkovich, A.A, and Kholodovych, V.V. (1996). Minireview. Fluorogenic Substrates for Proteases Based on Intramolecular Fluorescence Energy Transfer (IFETS). Journal of Biochemical and Biophysical Methods 33, 135 – 162.

- [182] Lottenberg, R., Christensen, U., Jackson, C.M., and Coleman, P.L (1981). Assay of Coagulation Proteases Using Peptide Chromogenic and Fluorogenic Substrates. *Methods in Enzymology* 80, 341 – 361.
- [183] Manafi, M., Kneifel, W., and Bascomb, S. (1991). Fluorogenic and Chromogenic Substrates Used in Bacterial Diagnostics. *Microbiological Reviews* 55 (3), 335 – 348
- [184] Spencer, P.W., Titus, J.S., and Spencer, R.D. (1975). Direct Fluorimetric Assay for Proteolytic Activity Against Intact Proteins. *Analytical Biochemistry* 64, 556 – 566.
- [185] Zimmerman, M, Yurewicz, E, and Patel, G. (1976). A New Fluorogenic Substrate for Chymotrypsin. *Analytical Biochemistry* **70**, 258 262.
- [186] Wiesner, R., and Troll, W. (1982). A New Assay for Proteases Using Fluorescent Labeling of Proteins. *Analytical Biochemistry* **121**, 290 294.
- [187] DeLumen, BO, and Tappel, AL. (1970). Fluorescein-hemoglobin as a Substrate for Cathepsin D and Other Proteases. *Analytical Biochemistry* 36, 22 – 29.
- [188] Twining, S.S. (1984). Fluorescein Isothiocyanate-Labeled Casein Assay for Proteolytic Enzymes. *Analytical Biochemistry* **143**, 30 – 34.
- [189] Maeda, H. (1979). Assay of Proteolytic Enzymes by the Fluorescence Polarization Technique. *Analytical Biochemistry* **92**, 222 – 227.
- [190] Bolger, R., and Checovich, W. (1984). A New Protease Activity Assay Using Fluorescence Polarization *BioTechniques* **17** (3), 585 589.

- [191] Khalfan, H., Abuknesha, R., and Robinson, D. (1983). Fluorigenic Method for the Assay of Proteinase Activity With the Use of 4-Methylumbelliferyl-Casein. *Biochemical Journal* 209, 265 – 267.
- [192] Weber, G. (1953) Rotational Brownian Motion and Polarization of Solutions. Advances in Protein Chemistry 8, 415 – 459.
- [193] Weber, G. (1952). Polarization of the Fluorescence of Macromolecules.
  1. Theory and Experimental Method. *Biochemical Journal* 51, 145 155.
- [194] Weber, G. (1952). Polarization of the Fluorescence of Macromolecules
  2. Fluorescent Conjugates of Ovalbumin and Bovine Serum Albumin. Biochemical Journal 51, 155 – 167.
- [195] Weber, G. (1956). Photoelectric Method for the Measurement of the Polarization of the Fluorescence of Solutions. *Journal of the Optical Society of America* 46 (11), 962 – 970.
- [196] Dandliker, W.B. and Feigen, G.A. (1961). Quantification of the Antigen-Antibody Reaction by the Polarization of Fluorescence. *Biochemical and Biophysical Research Communications* 5 (4), 299 – 304
- [197] Dandliker, W.B. and de Saussure, V.A. (1970). Fluorescence Polarization in Immunochemistry. *Immunochemistry* **7**, 799 – 828.
- [198] Dandliker, W.B., Kelly, R.J., Dandliker, J., Farquhar, J., and Levin, J. (1973).
   Fluorescence Polarization Immunoassay. Theory and Experimental Method. Immunochemistry 10, 219 227.

- [199] Levison, S.A., Dandliker, W.B., Brawn, R.J., and Vanderlaan, W.P. (1976). Fluorescence Polarization Measurement of the Hormone-Binding Site Interaction. *Endocrinology* 99 (4), 1129 – 1143.
- [200] Jolley, M.E. Fluorescence Polarization Immunoassay for the Determination of Therapeutic Drug Levels in Human Plasma. *Journal of Analytical Toxicology* **5**, 236 240.
- [201] Wu, P., Brasseur, M., Schindler, U. (1997) A High-Throughput STAT Binding Assay Using Fluorescence Polarization. *Analytical Biochemistry* 249, 29 – 36.
- [202] Jolley, M E. (1996) Fluorescence Polarization Assays for the Detection of Proteases and Their Inhibitors. *Journal of Biomolecular Screening* 1, 33 – 38.
- [203] Lynch, B A, Loiacono, K.A., Tiong, C.L., Adams, S.E., and MacNeil, I.A. (1997). A Fluorescence Polarization Based Src-SH2 Binding Assay. *Analytical Biochemistry* 247, 77 – 82.
- [204] Parker, G J., Law, T.L., Lenoch, F.J., and Bolger, R.E. (2000). Development of High Throughput Screening Assays Using Fluorescence Polarization: Nuclear Receptor-Ligand-Binding and Kinase / Phosphatase Assays. *Journal of Biomolecular Screening* 5 (2), 77 – 88.
- [205] Allen, M., Reeves, J., and Mellor, G. (2000). High Throughput Fluorescence Polarization: A Homogeneous Alternative to Radioligand Binding for Cell Surface Receptors. *Journal of Biomolecular Screening* 5 (2), 63 – 69

264

- [206] Bergstrom, K. (1977). Determination of Trypsin in Duodenal Fluid Using a New Chromogenic Substrate and a Reaction Rate Instrument. Science Tools, The LKB Instrument Journal 24 (1), 8 – 9.
- [207] Schade, SZ, Jolley, M.E., Sarauer, B.J., and Simonson, L.G. (1996).
   BODIPY-α-casein, a pH-Independent Protein Substrate for Protease Assays Using Fluorescence Polarization. *Analytical Biochemistry* 243, 1 – 7.
- [208] Molecular Probes Inc. (1999). EnzChek™ Polarization Assay Kits for Proteases.
- [209] Levine, L.M., Michener, M.L., Toth, M.V., and Holwerda, B.C. (1997). Measurement of Specific Protease Activity Utilizing Fluorescence Polarization. *Analytical Biochemistry* 247, 83 – 88.
- [210] Grys, E L., Schade, S Z., Cohen, M E., Geivelis, M., Robinson, P.J., and Simonson, L.G. (2000). Measurement of Proteases in Human Subgingival Dental Plaque by Fluorescence Polarization. Archives of Oral Biology 45, 1101 – 1106.
- [211] Khalfan, H., Abuknesha, R., and Robinson, D. (1983). Fluorigenic Method for the Assay of Proteinase Activity With the Use of 4-Methylumbelliferyl-Casein. *Biochemical Journal* 209, 265 – 267.
- [212] Mallya, S.K., Meyer, S., Bozyczko, D., Siman, R., and Ator, M.A. (1998).
   A Sensitive, Continuously Recording Fluorogenic Assay for Calpain.
   Biochemical and Biophysical Research Communications 248, 293 296.
- [213] Van Noorden, C.J F., Boonacker, E., Bissell, E.R., Meijer, A.J., van Marle, J., and Smith, R.E. (1997). Ala-Pro-Cresyl Violet, a Synthetic Fluorogenic

Substrate for the Analysis of Kinetic Parameters of Dipeptidyl Peptidase IV (CD26) in Individual Living Rat Hepatocytes. *Analytical Biochemistry* **252**, 71 – 77.

- [214] Ramjee, M.K. (2000). The Use of Fluorogenic Substrates to Monitor Thrombin Generation for the Analysis of Plasma and Whole Blood Coagulation. *Analytical Biochemistry* 277, 11 – 18.
- [215] Latt, S.A., Auld, D.S., and Vallee, B.L. (1972). Fluorescence Determination of Carboxypeptidase A Activity Based on Electronic Energy Transfer. *Analytical Biochemistry* 50, 56 – 62.
- [216] Wang, G.T., and Krafft, G.A. (1992). Automated Synthesis of Fluorogenic Protease Substrates: Design of Probes for Alzheimer's Disease-Associated Proteases. *Bioorganic and Medicinal Chemistry Letters* 2(12), 1665 – 1668.
- [217] Krafft, G.A., and Wang, G.T. (1993). Fluorescent Probes in Studies of Proteases. *Fluorescent Chemosensors for Ion and Molecule Recognition* 538, 183 – 195.
- [218] Taliani, M., Bianchi, E., Narjes, F., Fossatelli, M., Urbani, A., Steinkuhler,
   C., De Francesco, R., and Pessi, A. (1996). A Continuous Assay of
   Hepatitis C Virus Protease Based on Resonance Energy Transfer
   Depsipeptide Substrates. *Analytical Biochemistry* 240, 60 67.
- [219] Kakiuchi, N., Nishikawa, S., Hattori, M, and Shimotohno, K. (1999). A High Throughput Assay of the Hepatitis C Virus Nonstructural Protein 3 Serine Proteinase. *Journal of Virological Methods* 80, 77 – 84.

- [220] Fitzgerald, M.C., Laco, G.S., Elder, J.H., and Kent, S.B H. (1997). A Continuous Fluorometric Assay for the Feline Immunodeficiency Virus Protease. *Analytical Biochemistry* 254, 226 – 230.
- [221] Wang, G.T., Matayoshi, E., Huffaker, J., and Krafft, G.A. (1990). Design and Synthesis of New Fluorogenic HIV Protease Substrates Based on Resonance energy Transfer. *Tetrahedron Letters* 31(45), 6493 – 6496.
- [222] Zhong, W., and Benkovic, S J. (1998). Development of an Internally Quenched Fluorescent Substrate for Escherichia coli Leader Peptidase. Analytical Biochemistry 255, 66 – 73.
- [223] Oliveira, M C F, Hirata, I.Y., Chagas, J.R., Boschcov, P., Gomes, R.A.S., Figueiredo, A.F.S., and Juliano, L. (1992). Intramolecularly Quenched Fluorogenic Peptide Substrates for Human Renin. *Analytical Biochemistry* 203, 39 – 46.
- [224] Wang, G.T., Chung, C.C., Holzman, T.F., and Krafft, G.A. (1993). A Continuous Fluorescence Assay of Renin Activity. *Analytical Biochemistry* 210, 351 - 359.
- [225] Beebe, K.D., and Pei, D. (1998). A Continuous Fluorimetric Assay for Tail-Specific Protease. *Analytical Biochemistry* **263**, 51 – 56.
- [226] Persson, A., and Wilson, I.B. (1977). A Fluorogenic Substrate for Angiotensin-Converting Enzyme. *Analytical Biochemistry* **83**, 296 – 303.
- [227] Jean, F., Basak, A., DiMaio, J., Seidah, N.G., and Lazure, C. (1995). An Internally Quenched Fluorogenic Substrate of Prohormone Convertase 1 and Furin Leads to a Potent Prohormone Convertase Inhibitor. *Biochemical Journal* 307, 689 – 695.

- [228] Melo, R.L., Alves, L C., Del Nery, E., Juliano, L., and Juliano, M.A. (2001). Synthesis and Hydrolysis by Cysteine and Serine Proteases of Short Internally Quenched Fluorogenic Peptides. Analytical *Biochemistry* 293, 71 – 77.
- [229] Peng, S-B, Zheng, F., Angleton, E L., Smiley, D., Carpenter, J., and Scott, J.E. (2001). Development of an Internally Quenched Fluorescent Substrate and a Continuous Fluorimetric Assay for Streptococcus pneumoniae Signal Peptidase I. *Analytical Biochemistry* 293, 88 – 95.
- [230] Anne, C., Cornille, F., Lenoir, C., and Roques, B.P. (2001). High-Throughput Fluorogenic Assay for Determination of Botulinum Type B Neurotoxin Protease Activity. *Analytical Biochemistry* 291, 253 – 261.
- [231] Washiya, K., Furuike, T., Nakajima, F., Lee, Y.C., and Nishimura, S-I.
   (2000). Design of Fluorogenic Substrates for Continuous Assay of Sialyltransferase by Resonance Energy Transfer. *Analytical Biochemistry* 283, 39 – 48.
- [232] Castillo, M.J., Kurachi, K., Nishino, N., Ohkubo, I., and Powers, J.C. (1983). Activity of Bovine Blood Coagulation Factor  $IX_{a\beta}$ , Factor  $X_{a\beta}$ , and Factor XI<sub>a</sub> Toward Fluorogenic Peptides Containing the Activation Site Sequences of Bovine Factor IX and Factor X. *Biochemistry* **22**, 1021 1029.
- [233] Yaron, A., Carmel, A., and Katchalski-Katzir, E. (1979). Intramolecularly Quenched Fluorogenic Substrates for Hydrolytic Enzymes. *Analytical Biochemistry* 95, 228 – 235.

- [234] Grahn, S., Ullmann, D., and Jakubke, H-D. (1998). Design and Synthesis of Fluorogenic Trypsin Peptide Substrates Based on Resonance Energy Transfer. Analytical Biochemistry 265, 225 – 231.
- [235] Csuhai, E., Juliano, M.A., St. Pyrek, J., Harms, A.C., Juliano, L., and Hersh, L.B. (1999). New Fluorogenic Substrates for N-Arginine Dibasic Convertase. *Analytical Biochemistry* 269, 149 – 154.
- [236] Johnson, G.D., and Ahn, K. (2000). Development of and Internally Quenched Fluorescent Substrate Selective for Endothelin-Converting Enzyme-1. Analytical Biochemistry 286, 112 – 118.
- [237] Yoon, H.Y., Choi, KY, and Song, BD. (2000). Fluorometric Assay of Turnip Mosaic Virus NIa Protease. Analytical Biochemistry 277, 228 – 231.
- [238] Vencill, C.F., Rasnick, D., Crumley, KV, Nishino, N., and Powers, JC. (1985). Clostridium histolyticum Collagenase: Development of New Thio Ester, Fluorogenic, and Depsipeptide Substrates and New Inhibitors. Biochemistry 24, 3149 – 3157.
- [239] Stack, M S., and Gray, R.D. (1989). Comparison of Vertebrate Collegenase and Gelatinase Using a New Fluorogenic Substrate Peptide. *Journal of Biological Chemistry* 264(8), 4277 – 4281.
- [240] Angliker, H., Neumann, U., Molloy, S.S., and Thomas, G. (1995). Internally Quenched Fluorogenic Substrate for Furin. Analytical Biochemistry 224, 409 – 412.
- [241] Capobianco, J.O., Lerner, C.G., and Goldman, R.C. (1992). Application of a Fluorogenic Substrate in the Assay of Proteolytic Activity and in the

Discovery of a Potent Inhibitor of *Candida albicans* Aspartic Proteinase. *Analytical Biochemistry* **204**, 96 – 102.

- [242] Nishino, N., and Powers, J.C. (1980). *Pseudomonas aeruginosa* Elastase. *Journal of Biological Chemistry* **255** (8), 3482 – 3486.
- [243] Gulnik, S.V., Suvorov, L.I., Majer, P., Collins, J., Kane, B.P., Johnson,
   D.G., and Erickson, J.W. (1997) Design of Sensitive Fluorogenic
   Substrates for Human Cathepsin D. *FEBS Letters* 413, 379 384.
- [244] Rush, R.S., Mitas, M., Powers, J.C., Tanaka, T., and Hersh, L.B. (1984).
   Fluorogenic Substrates for the Enkephalin-Degrading Neutral Endopeptidase (Enkephalinase). Archives of Biochemistry and Biophysics 231 (2), 390 – 399.
- [245] Bonneau, P.R., Plouffe, C., Pelletier, A., Wernic, D., and Poupart, M-A.
   (1998). Design of Fluorogenic Peptide Substrates for Human Cytomegalovirus Protease Based on Structure-Activity Relationship Studies. *Analytical Biochemistry* 255, 59 – 65.
- [246] Del Nery, E., Juliano, M.A., Meldal, M., Svendsen, I., Scharfstein, J., Walmsley, A, and Juliano, L (1997). Characterization of the Substrate Specificity of the Major Cysteine Protease (cruzipain) from *Trypanosoma cruzi* Using a Portion-Mixing Combinatorial Library and Fluorogenic Peptides. *Biochemical Journal* 323, 427 – 433.
- [247] Maggiora, L.L., Smith, C.W., and Zhang, Z-Y. (1992). A General Method for the Preparation of Internally Quenched Fluorogenic Protease Substrates Using Solid-Phase Peptide Synthesis. *Journal of Medicinal Chemistry* 35, 3727 – 3730.

- [248] Geoghegan, K.F., Emery, M.J., Martin, W.H., McColl, A S., and Daumy, G.O (1993). Site-Directed Fluorescent Tagging of Human Renin and Collagenase (MMP-1) Substrate Peptides Using the Periodate Oxidation of N-Terminal Serine. An Apparently General Strategy for Provision of Energy-Transfer Substrates for Proteases. *Bioconjugate Chemistry* 4, 537 544.
- [249] Charitos, C., Tzougraki, C., and Kokotos, G. (2000). Synthesis and Fluorescence Properties of Intramolecularly Quenched Fluorogenic p-Nitroanilides containing Coumarin or Quinolinone Derivatives as Fluorophores Journal of Peptide Research 56, 360 – 372.
- [250] Bickett, D.M., Green, M.D., Berman, J., Dezube, M., Howe, A.S., Brown, P J, Roth, J.T., and McGeehan, G.M. (1993). A High Throughput Fluorogenic Substrate for Interstitual Collegenase (MMP-1) and Gelatinase (MMP-9). Analytical Biochemistry 212, 58 - 64.
- [251] Ambrose, W.P., Semin, D.J., Robbins, D.L., Van Orden, A., Kashem, M.A., Hamilton, S.A., Nelson, R.M., Jett, J.H., and Keller, R.A. (1998).
   Detection System for Reaction-Rate Analysis in a Low-Volume Proteinase-Inhibition Assay. *Analytical Biochemistry* 263, 150 157.
- [252] Cummings, R.T., McGovern, H.M., Zheng, S., Park, Y.W., and Hermes, J.D. (1999). Use of a Phosphotyrosine-Antibody Pair as a General Detection Method in Homogeneous Time-Resolved Fluorescence: Application to Human Immunodeficiency Viral Protease. *Analytical Biochemistry* 269, 79 – 93.
- [253] Meldal, M., Svendsen, I.B., Breddam, K, and Auzanneau, F-I. (1994). Portion-Mixing Peptide Libraries of Quenched Fluorogenic Substrates for

Complete Subsite Mapping of Endoprotease Specificity. *Proceedings of the National Academy of Sciences USA* **91**, 3314 - 3318.

- [254] Rosse, G., Kueng, E., Page, M.G.P., Schauer-Vukasinovic, V., Giller, T., Lahm, H-W., Hunziker, P., and Schlatter, D. (2000). Rapid Identification of Substrates for Novel Proteases Using a Combinatorial Peptide Library. *Journal of Combinatorial Chemistry* 2, 461 – 466.
- [255] Jones, L J., Upson, R.H., Haugland, R P., Panchuk-Voloshina, N., Zhou,
   M., and Haugland, R.P. (1997). Quenched BODIPY Dye-Labeled Casein
   Substrates for the Assay of Protease Activity by Direct Fluorescence
   Measurement. Analytical Biochemistry 251 (2), 144 152.
- [256] Homer, K.A., and Beighton, D. (1990). Fluorometric Determination of Bacterial Protease Using Fluorescein Isothiocyanate-Labeled Proteins as Substrates. Analytical Biochemistry 191, 133 – 137.
- [257] Voss, E.W., Workman, C.J., and Mummert, M E. (1996). Detection of Protease Activity Using a Fluorescence-Enhancement Globular Substrate. *BioTechniques* 20 (2), 286 – 291.
- [258] Thompson, V.F, Saldana, S., Cong, J., and Goll, D E. (2000). A BODIPY Fluorescent Microplate Assay for Measuring Activity of Calpains and Other Proteases. *Analytical Biochemistry* 279, 170 – 178.
- [259] Reis, R.C M, Sorgine, M H F., Coelho-Sampaio, T. (1998) A Novel Methodology for the Investigation of Intracellular Proteolytic Processing in Intact Cells. *European Journal of Cell Biology* 75, 192 – 197.
- [260] Molecular Probes, Inc. (1999). EnzChek<sup>™</sup> Protease Assay Kits, product information, 1 – 4.

- [261] Jiang, P., and Mellors, A. (1998). Membrane Protein Proteolysis Assayed by Fluorescence Quenching: Assay of O-Sialoglycoprotein Endopeptidase. *Analytical Biochemistry* 259, 8 – 15
- [262] Trevor Palmer, ed. (1991). "Enzyme Inhibition". In Understanding Enzymes. Third Edition. Chichester: Ellis Horwood, Chapter 8
- [263] C.Graham Knight. (1986) "The Characterization of Enzyme Inhibition". In Barrett, A.J. and Salvesen, G., eds. *Proteinase Inhibitors*. Elsevier Science Publishers BV(Biomedical Division), Chapter 2.
- [264] Manfred E. Wolff, ed (1995). "Approaches to the Rational Design of Enzyme Inhibitors". In Burger's Medicinal Chemistry and Drug Discovery. Fifth Edition, volume 1:Principles and Practice. New York: John Wiley and Sons, Inc., Chapter 18.
- [265] Schnebli, H.P., and Braun, N.J (1986). "Proteinase Inhibitors as Drugs".
   In Barrett, A.J. and Salvesen, G., eds. *Proteinase Inhibitors*. New York:
   Elsevier Science Publishers BV (Biomedical Division), Chapter 21.
- [266] Barrett, A.J. and Salvesen, G., eds. (1986). *Proteinase Inhibitors*. New York: Elsevier Science Publishers BV (Biomedical Division).
- [267] Trautschold, I., Werle, E., and Zickgraf-Rüdel, G. (1967). Trasylol. Biochemical Pharmacology 16, 59 – 72.
- [268] Gebhard, W., Tschesche, H., and Fritz, H. (1986). "Biochemistry of Aprotinin and Aprotinin-Like Inhibitors". In Barrett, A J. and Salvesen, G., eds. *Proteinase Inhibitors* New York: Elsevier Science Publishers BV (Biomedical Division), Chapter 10.

- [269] Blow, D.M., and Wright, C.S. (1972). A Model for the Association of Bovine Pancreatic Trypsin Inhibitor with Chymotrypsin and Trypsin. *Journal of Molecular Biology* 69, 137 – 144.
- [270] Antonini, E., Ascenzi, P., Menegatti, E., and Guarneri, M. (1983). Multiple Intermediates in the Reaction of Bovine β-Trypsin with Bovine Pancreatic Trypsin Inhibitor (Kunitz). *Biopolymers* 22, 363 – 375.
- [271] van Mierlo, C.P.M., Darby, N J., and Creighton, T.E. (1992). The Partially Folded Conformation of the Cys-30 Cys-51 Intermediate in the Disulfide Folding Pathway of Bovine Pancreatic Trypsin Inhibitor. *Proceedings of the National Academy of Science USA* 89, 6775 – 6779.
- [272] Goldenberg, D.P., Bekeart, L.S., Laheru, D.A., and Zhou, J.D. (1993).
   Probing the Determinants of Disulfide Stability in Native Pancreatic
   Trypsin Inhibitor. *Biochemistry* 32, 2835 2844
- [273] Peters, D.C., and Noble, S. (1999). Aprotinin An update of its Pharmacology and Therapeutic Use in Open Heart Surgery and Coronary Artery Bypass Surgery. *Drugs* 57 (2), 233 – 260.
- [274] D'Errico, C.C., Munro, H.M., and Bove, E.L. (1999). Pro: The Routine Use of Aprotinin During Pediatric Cardiac Surgery Is a Benefit. *Journal of Cardiothoracic and Vascular Anesthesia* **13** (6), 782 – 784.
- [275] Stassen, J.M., Lambeir, A-M., Matthyssens, G., Ripka, W.C., Nyström, Å., Sixma, J.J., and Vermylen, J. (1995). Characterisation of a Novel Series of Aprotinin-Derived Anticoagulants. *Thrombosis and Haemostasis* 74 (2), 646 – 654.

- [276] Umezawa, H. (1976). Structures and Activities of Protease Inhibitors of Microbial Origin. *Methods in Enzymology* 45, 678 – 683.
- [277] Knight, C.G. (1980). Human Cathepsin B. Application of the Substrate N-Benztloxycarbonyl-L-arginyl-L-arginine 2-Naphthylamide to a Study of the Inhibition by Leupeptin. *Biochemical Journal* 189, 447 – 453.
- [278] Suenaga, H., Yamamoto, H., and Shinkai, S. (1996). Screening of Boronic Acids for Strong Inhibition of the Hydrolytic Activity of α-Chymotrypsin and for Sugar Sensing Associated With a Large Fluorescence Change. *Pure and Applied Chemistry* 68 (11), 2179 – 2186.
- [279] Kettner, C A., Bone, R., Agard, D.A., and Bachovchin, W.W. (1988). Kinetic Properties of the Binding of α-Lytic Protease to Peptide Boronic Acids. *Biochemistry* 27, 7682 – 7688.
- [280] Bachovchin, W.W., Wong, W.Y.L., Farr-Jones, S., Shenvi, A.B., and Kettner, C.A. (1988). Nitrogen-15 NMR Spectroscopy of the Catalytic-Triad Histidine of a Serine Protease in Peptide Boronic Acid Inhibitor Complexes. *Biochemistry* 27, 7689 – 7697.
- [281] Umezawa, H., Aoyagi, T., Morishima, H., Matsuzaki, M., Hamada, M., and Takeuchi, T. (1970). Pepstatin, a New Pepsin Inhibitor Produced by Actinomycetes. *The Journal of Antibiotics* 23 (5), 259 – 262.
- [282] Fujinaga, M., Chernaia, M.M., Tarasova, N I., Mosimann, S.C., and James, M N.G. (1995). Crystal Structure of Human Pepsin and its Complex with Pepstatin. *Protein Science* 4, 960 – 972.

- [283] Sarpara, G.H. (2000). Fluongenic Enzyme Assays Using Long-Wavelength Substrates. Loughborough University, Unpublished Ph D. Thesis.
- [284] Hitachi. (1991). Instruction Manual Model F-4500 Fluorescence Spectrophotometer. Section 2 4.
- [285] Perkin-Elmer. (1991). The Model LS 50B Luminescence Spectrometer User's Manual. Chapter 8.
- [286] Smith, P.K., Krohn, R.I., Hermanson, G.T., Mallia, A.K., Gartner, F H, Provenzano, M.D., Fujimoto, E.K., Goeke, N.M., Olson, B.J., Klenk, D.C. (1985). Measurement of Protein Using Bicinchoninic Acid. Analytical Biochemistry 150, 76 – 85.
- [287] Schobel, U., Egelhaaf, H-J., Brecht, A, Oelkrug, D., and Gauglitz, G. (1999). New Donor-Acceptor Pair for Fluorescent Immunoassays by Energy Transfer. *Bioconjugate Chemistry* **10**, 1107 – 1114.
- [288] Schobel, U., Egelhaaf, H-J., Frohlich, D., Brecht, A, Oelkrug, D., and Gauglitz, G. (2000). Mechanisms of Fluorescence Quenching in Donor-Acceptor Labeled Antibody-Antigen Conjugates. *Journal of Fluorescence* 10(2), 147 – 154.
- [289] Ravdin, P., and Axelrod, D. (1977). Fluorescent Tetramethyl Rhodamine Derivatives of α-Bungarotoxin: Preparation, Separation, and Characterization. Analytical Biochemistry 80, 585 – 592.
- [290] Sarpara, G.H., Hu, S.H., Palmer, D.A., French, M.T., Evans, M., and Miller, J.N. (1999). A New Long-Wavelength Fluorigenic Substrate for

Alkaline Phosphatase: Synthesis and Characterisation. *Analytical Communications* **36**, 19 – 20.

- [291] Miller, J.N., and Miller, J.C. (1993) Statistics and Chemometrics for Analytical Chemistry (Fourth Edition). UK: Pearson Education Limited, pp 120 – 123.
- [292] Hutcheoun, G.A., Parker, M.C., James, A., and Moore, B.D. (1997). Methanol Dramatically Enhances Serine Protease Activity Under Anhydrous Conditions. *Chemical Communications*, 931 – 932.
- [293] Reslow, M., Adlercreutz, P., and Mattiasson, B. (1992). Modification of the Microenvironment of Enzymes in Organic Solvents. Substitution of Water by Polar Solvents. *Biocatalysis* 6, 307 - 318.
- [294] Almarsson, Ö., and Klibanov, A.M. (1996) Remarkable Activation of Enzymes in Nonaqueous Media by Denaturing Organic Solvents. *Biotechnology and Bioengineering* 49, 87 – 92.
- [295] Zeng, H-H., Wang, K-M., Li, D., and Yu, R-Q. (1994). Development of an Alcohol Optode Membrane Based on Fluorescence Enhancement of Fluorescein Derivatives. *Talanta* 41 (6), 969 – 975.
- [296] Stryer, L. (1965). The Interaction of a Naphthalene Dye With Apomyoglobin and Apohemoglobin. A Fluorescent Probe of Non-Polar Binding Sites. *Journal of Molecular Biology* 13, 482 – 495.
- [297] Molecular Probes, Inc. (2001). "BODIPY Dyes Spanning the Visible<br/>Spectrum".Accessedviawww.<a href="http://www.probes.com/handbook/sections/0103.html">http://www.probes.com/handbook/sections/0103.html</a> (5 September 2001).

- [298] Bogue, R. (1995). MicroTAS a Revolution in the Making for Analytical Instrument Technology. *Laboratory Equipment Digest, February*, 14 16.
- [299] Effenhauser, C.S., and Manz, A. (1994). Miniaturizing a Whole Analytical Laboratory Down to Chip Size. *American Laboratory, September*, 15 18.
- [300] Manz, A., Graber, N., and Widmer, H.M. (1990). Miniaturized Total Chemical Analysis Systems: a Novel Concept for Chemical Sensing. Sensors and Actuators B 1, 244 – 248.
- [301] Figeys, D., and Pinto, D. (2000). Lab-on-a-Chip: A Revolution in Biological and Medical Sciences. *Analytical Chemistry* 330A 335A.
- [302] McCreedy, T. (2000). Fabrication Techniques and Materials Commonly Used for the Production of Microreactors and Micro Total Analytical Systems. *Trends in Analytical Chemistry* 19(6), 396 – 401.
- [303] Guetens, G., Van Cauwenberghe, K., De Boeck, G., Maes, R., Tjaden, U.R, van der Greef, J, Highley, M., van Oosterom, A.T., and Bruijn, E.A. (2000). Nanotechnology in Bio / Clinical Analysis. *Journal of Chromatography B* 739, 139 150.
- [304] Vo-Dinh, T., Cullum, B M, and Stokes, D.L. (2001). Nanosensors and Biochips: Frontiers in Biomolecular Diagnostics. Sensor and Actuators B 74, 2 – 11.
- [305] Erbacher, C., Bessoth, F G, Busch, M, Verpoorte, E., and Manz, A. (1999). Towards Integrated Continuous-Flow Chemical Reactors. *Mikrochimica Acta* 131, 19 – 24.

- [306] Weigl, B.H. (2000). Microfluidics-Based Lab-on-a-Chip Systems. *IVD* Technology November / December, 47 – 55.
- [307] Hadd, A.G., Raymond, D.E., Halliwell, J.W., Jacobson, S.C., and Ramsey,
   M. (1997). Microchip Device for Performing Enzyme Assays. *Analytical Chemistry* 69, 3407 3412.
- [308] Cohen, C.B., Chin-Dixon, E., Jeong, S., and Nikiforov, T.T. (1999). A Microchip-Based Enzyme Assay for Protein Kinase A. Analytical Biochemistry 273, 89 – 97.
- [309] Boone, T.D., Ricco, A.J., Gooding, P., Björnson, T.O., Singh, S., Xiao, V., Gibbons, I., Williams, S.J., and Tan, H. (2000). "Sub-Microliter Assays and DNA Analysis on Plastic Microfluidics". In A. van den Berg *et al.*, eds. *Micro Total Analysis Systems 2000.* Netherlands: Kluwer Academic Publishers, pp 541 – 544.
- [310] Kopf-Sill, A. (2000). "Commercializing Lab-on-a-Chip Technology". In A. van den Berg et al., eds. Micro Total Analysis Systems 2000 Netherlands: Kluwer Academic Publishers, pp 233 238.
- [311] Caliper (2001). "High Throughput Systems". Accessed via www. http://www.calipertech.com.html (09 July 2001).
- [312] Chow, A., Kopf-Sill, A., Nikiforov, T., Zhou, A., Coffin, J., Wada, G., Alajoki, L., Spaid, M., Yurkovetsky, Y., Sundberg, S., and Parce, J.W. (2000). "High-Throughput Screening on Microchips". In A. van den Berg *et al.*, eds. *Micro Total Analysis System 2000*. Netherlands: Kluwer Academic Publishers, pp 489 – 492.

- [313] Christensen, P.D., Johnson, S.W.P., McCreedy, T., Skelton, V, and Wilson, N.G. (1998). The Fabrication of Micro-Porous Silica Structures for Micro-Reactor Technology. *Analytical Communications* 35, 341 – 343.
- [314] Przekwas, A., Makhijani, V., Athavale, M., Klein, A., and Bartsch, P. (2000). "Computational Simulation of Bio-Microfluidic Processes in Integrated DNA Biochips". In A. van den Berg *et al.*, eds. *Micro Total Analysis System 2000*. Netherlands: Kluwer Academic Publishers, pp 561 – 564.
- [315] Badal, M.Y., Tang, T., Lee, W.E., Dickinson-Laing, T., Bader, D.E., and Harrison, D.J. (2000). "An Integrated System for Gene Detection Using Cycling Probe Technology". In A. van den Berg *et al.*, eds. *Micro Total Analysis System 2000*. Netherlands: Kluwer Academic Publishers, pp 423 – 426.
- [316] Kopp, M U., de Mello, A.J., and Manz, A. (1998). Chemical Amplification: Continuous-Flow PCR on a Chip. *Science* **280**, 1046 – 1048.
- [317] Lagally, E T., Paegal, B M., and Mathies, R.A. (2000). "Microfabrication Technology for Chemical and Biochemical Microprocessors". In A. van den Berg et al., eds. Micro Total Analysis System 2000. Netherlands: Kluwer Academic Publishers, 217 – 220.
- [318] Lao, A.I.K., Lee, T.M.H., Carles, M.C., and Hsing, I-M. (2000). "Thermal Management and Surface Passivation of a Miniaturized PCR Device for Traditional Chinese Medicine". In A. van den Berg *et al.*, eds. *Micro Total Analysis System 2000*. Netherlands: Kluwer Academic Publishers, pp 139 – 142.

- [319] Gottschlich, N., Culbertson, C.T., McKnight, T.E., Jacobson, S.C., and Ramsey, J.M. (2000). Integrated Microchip-Device for the Digestion, Separation and Postcolumn Labeling of Proteins and Peptides. *Journal of Chromatography B* 745, 243 – 249.
- [320] McCreedy, T., and Wilson, N.G. (2001). Microfabricated Reactors for On-Chip Heterogeneous Catalysis. *Analyst* **126**, 21 – 23
- [321] Tanaka, Y., Slyadnev, M.N., Hibara, A, Tokeshi, M., Kitamori, T. (2000). Non-Contact Photothermal Control of Enzyme Reactions on a Microchip by Using a Compact Diode Laser. *Journal of Chromatography A* 894, 45 – 51.
- [322] Huhmer, A.F.R., and Landers, J.P. (2000). Noncontact Infrared-Mediated Thermocycling for Effective Polymerase Chain Reaction Amplification of DNA in Nanoliter Volumes. *Analytical Chemistry* 72, 5507 – 5512.
- [323] Chiem,N., and Harrison, D.J. (1997). Microchip-Based Capillary Electrophoresis for Immunoassays: Analysis of Monoclonal Antibodies and Theophylline. Analytical Chemistry 69 (3), 373 – 378.
- [324] Hadd, A.G., Jabobson, S.C., and Ramsey, J.M. (1999). Microfluidic Assays of Acetylcholinesterase Inhibitors. *Analytical Chemistry* 71, 5206 – 5212.
- [325] Hadd., A.G., Raymond, D.E., Halliwell, J.W., Jacobson, S.C., and Ramsey, M. (1997) Microchip Device for Performing Enzyme Assays. *Analytical Chemistry* 69 (17), 3407 – 3412.
- [326] Hirano, K., Matsuzawa, Y., Yasuda, H., Katsura, S., and Mizuno, A. (2000). "Molecular Processing of a Single DNA Inside Channels for

Genome Analysis". In A. van den Berg *et al*, eds. *Micro Total Analysis System 2000*. Netherlands: Kluwer Academic Publishers, pp 439 – 442.

- [327] Oleschuk, R.D., Schultz-Lockyear, L.L., Ning, Y., and Harrison, D.J.
   (2000). Trapping of Bead-Based Reagents within Microfluidic Systems: On-Chip Solid-Phase Extraction and Electrochromatography. *Analytical Chemistry* 72, 585 – 590.
- [328] Lu, Q., Callahan, J.H, and Collins, G.E. (2000). The Selective Detection of Uranium (VI) on a Microchip Using a Derivatized 4-Sulfonic Calix[6]arene. *Chemical Communications*, 1913 – 1914.
- [329] Owens, C.V., Davidson, Y.Y., Kar, S., and Soper, S.A. (1997). High-Resolution Separation of DNA Restriction Fragments Using Capillary Electrophoresis with Near-IR, Diode-Based, Laser-Induced Fluorescence Detection. *Analytical Chemistry* 69, 1256 – 1261.
- [330] Wallenborg, S.R. and Bailey, C.G. (2000). Separation and Detection of Explosives on a Microchip Using Micellar Electrokinetic Chromatography and Indirect Laser-Induced Fluorescence. *Analytical Chemistry* 72, 1872 – 1878.
- [331] Ocvirk, G., Tang, T., and Harrison, D.J. (1998). Optimization of Confocal Epifluorescence Microscopy for Microchip-Based Miniaturized Total Analysis Systems. *Analyst* 123, 1429 – 1434.
- [332] Roulet, J-C., Herzig, H.P., Verpoorte, E., de Rooij, N.F., and Dandliker, R. (2000). "Integration of Micro-Optical Systems for Fluorescence Detection in μTAS Applications". In A. van den Berg *et al.*, eds. *Micro Total Analysis System 2000*. Netherlands: Kluwer Academic Publishers, pp 163 166.

[333] Schult, K., Katerkamp, A, Trau, D., Grawe, F., Cammann, K, and Meusel,
 M. (1999). Disposable Optical Sensor Chip for Medical Diagnostics: New
 Ways in Bioanalysis. *Analytical Chemistry* **71**, 5430 – 5435.



# Appendices

È,

# APPENDIX A

## Supplier names and addresses

SUPPLIER	ADDRESS
AD Instruments, Ltd.	Grove House
	Grove Road
	Hastings
	TN35 4JS, UK
Amersham Pharmacia Biotech UK, Ltd.	Amersham Place
	Little Chalfont
	Buckinghamshire
	HP7 9NA, UK
Anachem, Ltd.	20 Charles Street
	Luton
	Bedfordshire
	LU2 0EBG, UK
Cambio, Ltd.	34 Newnham Road
	Cambridge
	CB3 9EY, UK
Elkay Laboratory Products (UK), Ltd.	Unit 4, Marlborough Mews
	Crockford Lane
	Basingstoke, Hampshire
-	RG24 8NA, UK

Eppendorf Scientific, Inc	1 Cantiague Road	
	Westbury	
	New York, USA	
Fisher Scientific UK	Bishop Meadow Road	
	Loughborough	
	Leicestershire	
	LE11 5RG, UK	
Grant Instruments (Cambridge), Ltd.	29 Station Road	
	Shepreth	
	Royston	
	Hertforshire	
	SG8 6PZ, UK	
Hellma (England), Ltd.	Cumberland House	
	24 – 28 Baxter Avenue	
	Southend-on-Sea	
	Essex	
	SS2 6HZ, UK	
Hitachi Scientıfic Instruments	7 Ivanhoe Road	
	Hogwood Industrial Estate	
	Finchampstead	
	Berkshire	
	RG40 4QQ, UK	
Molecular Probes, Inc.	Cambridge Bioscience	
	24 – 25 Signet Court	
	Newmarket Road	
	Cambridge	
	CB5 8LA, UK	
	1	

Omnifit, Ltd.	2 College Park	
	Coldhams Lane	
	Cambridge	
	CB1 3HD, UK	
Perkin-Elmer, Ltd.	Post Office Lane	
	Beaconsfield	
	Buckinghamshire	
	HP9 1QA, UK	
Pierce (Perbio Science LIK I td.)	Century House	
	Tettenhall	
	Chashira	
	CH3 9RG, UK	
Precisa Balances Ltd.	4 Vermont Place	
	Tongwell	
	Milton Keynes, Buckinghamshire	
	MK15 8JA, UK	
Sigma-Aldrich Company, Ltd.	Fancy Road	
	Poole	
	Dorset	
	BH12 4QH, UK	
USF Elga Ltd.	High Street	
	Lane End	
	High Wycombe	
	Bucks.	
	HP14 35H, UK	

Unicam Limited	York Street	
	Cambridge	
	CB1 2PX, UK	
Whatman International Ltd.	St Leonard's Road	
	20 / 20 Maidstone	
	Kent	
	ME16 0LS, UK	

Table A - Supplier names and addresses

#### APPENDIX B

Appearance of the model F-4500 fluorescence spectrophotometer (Hitachi) located in our laboratory



# Figure B – Appearance of the F-4500 in our laboratory

#### APPENDIX C

Block diagram of the layout of the model F-4500 fluorescence spectrophotometer (Hitachi)



Figure C – Internal block diagram of the F-4500

### APPENDIX D

# Instrument specifications for the Hitachi F-4500 and Perkin-Elmer LS 50B fluorescence spectrometers

PARAMETERS	F-4500	LS 50B
Light source	150 W Xenon lamp	Xenon flash lamp, pulsed at line frequency (50 or 60 Hz)
Monochromators	Excitation: 200 – 600 nm Emission: 200 – 900 nm	Excitation: 200 – 800 nm Emission: 200 – 900 nm
Excitation slit width	1.0, 2.5, 5.0, 10.0 nm	2.5 – 15 nm in 0.1 nm increments
Emission slit width	1.0, 2.5, 5.0, 10.0, 20.0 nm	2.5 – 20 nm in 0.1 nm increments
Scanning speed	15, 60, 240, 1200, 2400, 12000, 30000 nm / min	10 – 1500 nm / min in 1 nm increments
Wavelength accuracy	± 2.0 nm	± 1.0 nm
Response	0.004, 0.01, 0.05, 0.1, 0.5, 2.0, 8.0 seconds	0.02 – 9.98 seconds in 0.01 s increments
PMT Voltage	400, 700, 950 V	Auto

Table D – Instrument parameters of the F-4500 and LS 50B
### APPENDIX E

Appearance of the LS 50B fluorescence spectrophotometer (Perkin-Elmer) located in our laboratory

Figure E – Appearance of the LS 50B

#### APPENDIX F

Block diagram of the layout of the Perkin-Elmer LS 50B fluorescence spectrophotometer



Figure F - Optical layout of the LS 50B fluorescence spectrophotometer

### **APPENDIX G**

# Calculation of fluorescein absorbance at 280 nm as a percentage of that at 494 nm

Fluorescein  $A_{280 nm}$  = 1.023, and  $A_{494 nm}$  = 2.545. Therefore, absorbance at 280 nm as a percentage of the absorbance at 494 nm = (1.023 / 2.545) X 100 = 40 %.

### APPENDIX H

# Calculation of BODIPY absorbance at 280 nm as a percentage of that at 630 nm

BODIPY  $A_{280 \text{ nm}} = 0.196$ , and BODIPY  $A_{630 \text{ nm}} = 2.073$ . Therefore, absorbance at 280 nm as a percentage of the absorbance at 630 nm = (0.196 / 2.073) X 100 = 9.5 %.

---

## **APPENDIX 1**

### Calculation of $\alpha$ -casein molar extinction coefficient ( $\epsilon$ ) at 280 nm

According to the Beer Lambert Law absorbance is related to the molar extinction coefficient  $\varepsilon$  (in cm<sup>-1</sup> M<sup>-1</sup>), concentration c (in M), and path length I (in cm) by: A =  $\varepsilon$ cl. A 0.1% solution of  $\alpha$ -casein (0.01 g per 10 ml) was made up and the absorbance at 280 nm measured (A<sub>280 nm</sub> = 0 882).

Molar concentration = (weight (g) / RMM (g)) X (1000 (ml) / volume (ml)) =  $(0.01 / 24 \ 000) \times (1000 / 10) = 4.17 \times 10^{-5} M$ 

Substituting into the Beer Lambert Law:

 $\epsilon = A / c = 0.882 / 4 17 X 10^{-5} = 21, 151 cm^{-1} M^{-1}$ .

### APPENDIX J

# Calculation of $\alpha$ -casein absorbance at 350 nm as a percentage of that at 280 nm

 $\alpha$ -casein A<sub>280 nm</sub> = 0.417, and  $\alpha$ -casein A<sub>350 nm</sub> = 0.046. Therefore, absorbance at 350 nm as a percentage of the absorbance at 280 nm = (0.046 / 0.417) X 100 = 11 %.

### APPENDIX K

# Calculation of BODIPY absorbance at 280 nm as a percentage of that at 350 nm

BODIPY  $A_{280 \text{ nm}} = 0.117$ , and BODIPY  $A_{350 \text{ nm}} = 0.338$ . Therefore, absorbance at 280 nm as a percentage of that at 350 nm = (0.117 / 0.338) X 100 = 35 %.

### APPENDIX L

### Calculation of BODIPY molar extinction coefficient (ɛ) at 350 nm

According to the Beer Lambert Law absorbance is related to the molar extinction coefficient  $\epsilon$  (in cm<sup>-1</sup> M<sup>-1</sup>), concentration c (in M), and path length I (in cm) by: A =  $\epsilon$ cl.

A 1 66 X 10  $^{-5}$  g / ml solution of BODIPY was made up and the absorbance at 350 nm measured (A<sub>350 nm</sub> = 1.049).

Molar concentration = (weight (g) / RMM (g)) X (1000 (ml) / volume (ml)) =  $(1.66 \times 10^{-5} / 660.5) \times (1000 / 1) = 2.52 \times 10^{-5} M$ 

Substituting into the Beer Lambert Law:

$$\epsilon$$
 = A / c = 1.049 / 2.52 X 10 <sup>-5</sup> = 41, 627 cm <sup>-1</sup> M <sup>-1</sup>

### APPENDIX M

### Calculation of flow rate

Flowing buffer through the flow system and collecting the buffer for 5 minutes in a glass vial that had been used to zero the analytical balance was used to determine the flow rate for use in protease and protease inhibition assays. The weight of the vial was then taken and the weight recorded. This was carried out in triplicate. This calculation makes the assumption that 1 g = 1 ml buffer.

Triplicate weights for 5 minutes: 9.2536, 9 2417, 9.2561. Therefore, the average of weights = 9.250467 with a standard deviation (SD) = 0.007694, and percentage SD = 0.083.

9 250467 ml was collected over 5 minutes, therefore for 1 minute 9 250467 / 5 = 1.85 ml was collected. Therefore, flow rate = 1 85 ml / min.

### APPENDIX N

#### **Determination of incubation loop volume**

Buffer was flowed through the incubation loop and captured by closing the valve so that the incubation loop was segregated from the rest of the flow system. Air was then flowed through the rest of the system to remove any traces of buffer. A glass vial was used to zero the analytical balance after which the buffer captured in the incubation loop was flowed out into the glass vial. The glass vial was then re-weighed. This measurement was repeated 6 times. Determination of loop volume assumes that 1 g = 1 ml buffer.

Weight of six readings (mg): 660.6, 667.0, 667.6, 671.0, 675.5, 678.0 This gives an average = 669.95, with a standard deviation (SD) = 6 298 and a percentage SD = 0 94 %. Therefore, loop volume = 670  $\mu$ l.

-

, ` -.